

Impact of thermal and drying regimes, used to make powders derived from spinach leaves, on their material properties and digestibility

Chao Chi

Thesis submitted to the University of Nottingham

for the degree of Doctor of Philosophy

Division of Food, Nutrition and Dietetic

School of Biosciences

University of Nottingham

Sutton Bonington Campus

Leicestershire LE12 5RD

June 2023

Abstract

Chloroplasts are photosynthetic organelles in plants which are a rich source of lipophilic nutrients like galactolipids, carotenoids, vitamin E and α -linolenic acid (ALA, 18:3). Recent interest in the use of chloroplast fractions as functional food ingredients has led to the development of methods for their isolation from plant tissues. In the current research, the removal of intact chloroplasts from the confinement of the cell wall was achieved using a simple and clean method (slow-screw, twin-gear juicer), without the addition of solvents or chemicals, which is efficient for liberating chloroplasts. This approach was developed in an attempt to avoid excess dilution and contamination as well as maintaining food quality, enhancing shelf-life stability. The extracts were further dried with the use of modern industrial technologies such as spray drying and freeze drying. Thermal preservation technologies such as pasteurisation of spinach juice (70 °C 15s and 90 °C 5min) and blanching of spinach leaves (100°C 30s) were also investigated to inactivate enzymes and eliminate pathogenic microorganisms during storage.

Food processing techniques are designed to improve the quality of products, by promoting physical and chemical changes in tissues but can also lead to losses, especially of micronutrients. Thermal processes, such as pasteurisation and blanching, inactivated endogenous enzymes in spinach extracts, with 95%-99% of peroxidase (POD) activity being reduced. The study on drying techniques showed that both spray and freeze drying retained nutrient levels to a similar level in spinach juice powder; and freeze dried chloroplast rich fraction (CRF) had a higher nutritional value than freeze dried juice. In addition, significant reductions in moisture content (below 10%) and water activity (below 0.3) in spinach juice/CRF powders are achieved with this suggesting better microbiological safety and enhanced shelf-life stability. However, thermal processing caused a significant quality loss in terms of colour and nutritional value. Key morphology changes of different treated spinach juice/CRF powders were observed by scanning electron microscopy (SEM) and highlighting the impact of processing procedure on organelle structures and providing a means to differentiate between each method used.

In addition, in vitro digestion assays, used to mimic the physiological conditions of human digestion, were used to measure the bioaccessibility of nutrients when spinach juice/CRF were processed in different ways. The digestion model consists of three phases (oral phase, gastric phase and intestinal phase), and the results showed that pasteurisation of spinach juice promoted the bioaccessibility of carotenoids and vitamin E, as well as enhanced the digestibility of galactolipids but hindered proteins digestion in spinach juice/CRF compared with fresh samples. The lipolysis of galactolipids, mainly monogalactosyldiacylglycerol (MGDG) and digalactosyldiacylglycerol (DGDG) was monitored by high performance thin layer chromatography (HPTLC) converted during digestion. MGDG and DGDG were to monogalactosylmonoacylglycerol (MGMG) and digalactosylmonoacylglycerol (DGMG), respectively, and ALA was the main fatty acid released. While the major galactolipid lipolysis occurred during the intestinal phase with the presence of porcine pancreatic extracts (PPE), and the protein hydrolysis occurred mainly during the gastric

phase with the addition of pepsin.

List of publications

 WATTANAKUL, J., SYAMILA, M., DARWISH, R., GEDI, M. A., SUTCHARIT,
P., CHI, C., AKEPACH, P., SAHAKA, M., GONTERO, B., CARRIÈRE, F. &
GRAY, D. A. 2022. Bioaccessibility of essential lipophilic nutrients in a chloroplast-rich fraction (CRF) from agricultural green waste during simulated human gastrointestinal tract digestion. Food & Function, 13, 5365-5380.

Acknowledgement

I would like to express my sincere gratitude to my supervisor, Professor David A. Gray for his invaluable supervision, enthusiastic support and patient guidance throughout my PhD study. I am so grateful for him giving me the opportunity to study in the division of Food, Nutrition, and Dietetics, University of Nottingham. His immense knowledge and plentiful experience have encouraged me through my academic research. Without his tutelage, this work would not have been possible.

I would also like to thank Dr Peter Rose, my co-supervisor, and Professor Andy Taylor for providing valuable feedback and suggestions on my thesis. Their insights and guidance were instrumental in helping me to shape my research and complete this thesis.

I would like to extend my sincere thanks to my colleagues, Dr. Jutarat Wattanakul, Dr Randa Darwish, Dr Syamila Mansor, Dr Gedi Mohamed, Dr. Vincenzo Di Bari, Dr. Ruth Price, Dr. Poramat Sucharit, Dr. Filippo Bramante and Malgorzata Walczak. Special thanks to Dr. Jutarat Wattanakul who proofread this thesis. Sincere thanks to Lim Mui and Darrell Cobon for their technical support in lab instrument. Sincere appreciation also goes to Dr Jianhua Jia and all my friends, especially Xinxin Li who gives me support at my difficult times. All of your kind help and support that have made my study and life in the UK a wonderful time.

Finally, I would like to express my gratitude to my beloved parents, Lijun Chi and Shuyan Tang. Thanks for your tremendous understanding and belief in me.

CONTENT

ABSTRACT	<u> I</u>
LIST OF PUBLICATIONS	<u>IV</u>
ACKNOWLEDGEMENT	V
LIST OF ABBREVIATIONS	<u>XI</u>
1. GENERAL INTRODUCTION	1
1.1. GENERAL OVERVIEW	1
1.2. LITERATURE REVIEW	4
1.2.1. Chloroplast	4
1.2.2. NUTRIENT PROFILES OF PLANT CHLOROPLAST AND THEIR IMPORTANCE IN HUMAN	1
HEALTH 7	
1.2.3. FOOD PRESERVATION TECHNOLOGIES	23
1.2.4. Thermal pre-treatment	29
1.2.5. IN VITRO DIGESTION	32
1.3. RESEARCH WITHIN DAVID GRAY'S GROUP	39
1.4. AIMS, HYPOTHESES AND OBJECTIVES.	40
1.5. THESIS STRUCTURE & CLARIFICATION	41
2. MATERIALS AND METHODS	42

2.1.	MATERIALS AND CHEMICALS	42
2.2.	ENZYMES	43
2.3.	Methods	43
2.3.1.	Spinach chloroplasts processing	43
2.3.2.	Spinach juice characteristics during storage at 4 $^\circ C$ for 7 days	49
2.3.3.	PHYSICOCHEMICAL PROPERTIES OF DRIED SPINACH CHLOROPLASTS	51
2.3.4.	STATIC IN VITRO DIGESTION MODEL	61
2.4.	STATISTICAL ANALYSIS	70

3. EFFECTS OF DRYING METHODS ON PHYSICOCHEMICAL PROPERTIES OF

SPINACH JUICE/CRF	7:	1
		_

3.1.		71
3.2.	SAMPLE PREPARATION SUMMARY	73
3.3.	RESULTS AND DISCUSSION	73
3.3.1.	NUTRIENTS CONTENT	73
3.3.2.	Physical properties of dried spinach juice/CRF	82
3.3.3.	PARTICLE SIZE OF CHLOROPLAST RICH POWDER PREPARATIONS	85
3.3.4.	POWDER MORPHOLOGY	87
3.4.	CONCLUSION	91

4. EFFECT OF SPINACH JUICE PASTEURISATION ON THE PHYSICOCHEMICAL

4.1. I	NTRODUCTION
4.2. 5	SAMPLE PREPARATION SUMMARY96
4.3. F	RESULTS AND DISCUSSION
4.3.1.	QUALITY CHANGES OF FRESH AND PASTEURISED SPINACH JUICE DURING REFRIGERATED
STORAGE	97
4.3.2.	EFFECT OF SPINACH JUICE PASTEURISATION ON THE NUTRIENT CONTENT OF DERIVED
POWDERS (DI	ried juice and dried pellet)
4.3.3.	Physical properties of fresh/pasteurised spinach juice/CRF
4.3.4.	Powder Morphology
4.4. (CONCLUSION
<u>5.</u> <u>EF</u> JUICE/CRF	FECTS OF BLANCHING ON PROPERTIES OF FREEZE-DRIED SPINACH
5.1. I	NTRODUCTION
5.2. 5	SAMPLE PREPARATION SUMMARY122
5.3. F	RESULTS AND DISCUSSION
5.3.1.	QUALITY CHANGES OF FRESH AND PASTEURISED SPINACH JUICE DURING REFRIGERATED
STORAGE	122
5.3.2.	MORPHOLOGY OF SPINACH SAMPLES
5.3.3.	MACRONUTRIENTS OF FRESH/BLANCHED SPINACH JUICE/CRF POWDERS
5.3.4.	MICRONUTRIENTS OF FRESH/BLANCHED SPINACH JUICE/CRF POWDERS
5 2 5	

5.4. CONCLUSION
6. IN VITRO DIGESTION OF GALACTOLIPIDS AND PROTEINS FROM SPINACH
JUICE/CRF
6.1. INTRODUCTION
6.2. IN VITRO DIGESTION MODEL
6.3. SAMPLE PREPARATION SUMMARY156
6.4. LIPID EXTRACTION AND MEASUREMENT FROM SPINACH JUICE/CRF BEFORE AND
DURING IN VITRO DIGESTION
6.5. LIMITATIONS ABOUT HPTLC ANALYSIS160
6.6. RESULTS AND DISCUSSION164
6.6.1. GALACTOLIPID DIGESTION TRENDS IN DIFFERENT CHLOROPLAST-ENRICHED POWDERS
USING BAND INTENSITY FROM HPTLC SEPARATION OF LIPIDS
6.6.2. QUANTIFICATION OF GALACTOLIPID DIGESTION AND FFA RELEASE IN DIFFERENT
CHLOROPLAST-ENRICHED POWDERS USING BAND INTENSITY FROM HPTLC SEPARATION OF LIPIDS
169
6.6.3. PROTEIN CHANGES DURING HEAT TREATMENT AND <i>IN VITRO</i> DIGESTION DETERMINED
BY SDS-PAGE
6.6.4. Overall discussion
6.7. CONCLUSION

7. BIOACCESSIBILITY OF LIPOPHILIC NUTRIENTS FROM SPINACH JUICE/CRF

DURING IN VITRO DIGESTION	

7.1.	INTRODUCTION
7.2.	RESULTS AND DISCUSSION
7.2.1.	Lipophilic nutrients of spinach chloroplasts before <i>in vitro</i> digestion 186
7.2.2.	Behaviour of spinach juice/CRF during <i>in vitro</i> static digestion model 189
7.2.3.	Carotenoids release from chloroplast material during simulated human
DIGESTION	193
7.2.4.	A-TOCOPHEROL RELEASE FROM SPINACH JUICE/CRF DURING SIMULATED HUMAN
DIGESTION	204
7.3.	CONCLUSION
<u>8.</u> <u>N</u>	AAIN FINDINGS, CONCLUSION AND SUGGESTION FOR FUTURE WORK
8.1.	THOUGHTS ON MAIN FINDINGS
8.2.	CONCLUSIONS
8.3.	SUGGESTION FOR FUTURE WORK
<u>IMPA</u>	CT OF THE COVID-19 PANDEMIC221
APPE	NDICES
<u>REFE</u>	RENCES

List of abbreviations

Abbreviation and Meaning

Spinach chloroplast materials

CRF	Chloroplast-rich fraction
SJ	Spray dried juice
FJ	Freeze dried juice
FC	Freeze dried CRF
MJ	Mild pasteurised juice
MC	Mild pasteurised CRF
IJ	Intense pasteurised juice
IC	Intense pasteurised CRF

All abbreviations (samples, chemicals, nutrients, measurements, etc)

α-toc	α-tocopherol
μg	Microgram
μL	Microlitre
μm	Micrometre (micron)
А	Absorbance
ALA	α-linolenic acid
AMD	Age-related macular degeneration
ANOVA	Analysis of variance
ATP	Adenosine triphosphate
β-car	β-carotene

BCA	Bicinchoninic acid
BHT	Butylated hydroxytoluene
C/ conc.	Concentration
Cal	Calorie
CEH/BSSL	Carboxyl ester hydrolase/ bile salt
	simulated lipase
CHCl3	Chloroform
Chl	Chlorophyll
Chl.a	Chlorophyll a
Chl.b	Chlorophyll b
СНЗОН	Methanol
cm	Centimetre
CO2	Carbon dioxide
CRF	Chloroplast-rich fraction
DAG	Diacylglycerol
DGDG	Digalactosyl diacylglycerol
DGMG	Digalactosyl monoacylglycerol
DHA	Docosahexaenoic acid
DW	Dry weight
EPA	Eicosapentaenoic acid
FA	Fatty acid
FC	Freeze dried CRF

xii

FD	Freeze drying
FAMEs	Fatty acid methyl esters
FFA	Free fatty acid
FJ	Freeze dried juice
g	Gram
GC-MS	Gas chromatography-mass spectrometry
GI	Gastrointestinal tract
GPLRP2	Pancreatic lipase-related protein 2 from
	guinea pig
h	Hour
HCl	Hydrochloric acid
HGL	Human gastric lipase
HMF	5-hydroxymethylfurfural
H2O2	Hydrogen peroxide
НРЈ	Human pancreatic juice
HPLC	High performance liquid
	chromatography
HPTLC	High performance thin layer
	chromatography
HTST	High temperature short time
IC	Intense pasteurised CRF
ICP-MS	Inductively coupled plasma-mass

spectrometry

IJ	Intense pasteurised juice
KCl	Potassium chloride
kDa	Kilodalton
kg	Kilogram
KH2PO4	Potassium dihydrogen phosphate
LA	Linoleic acid
LAHase	Lipid-acyl hydrolase
Lut	Lutein
М	Molar (mol/L)
MAG	Monoacylglycerol
MC	Mild pasteurised CRF
mg	Milligram
MGDG	Monogalactosyl diacylglycerol
MGG	Monogalactosylglycerol
MGMG	Monogalactosyl monoacylglycerol
min	Minute
MJ	Mild pasteurised juice
mL	Millilitre
mM	Millimolar
mm	Millimetre
mol	Mole

xiv

MW	Molecular weight			
NaCl	Sodium chloride			
NAD	Nicotinamide adenine dinucleotide			
NADPH	Nicotinamide adenine dinucleotide			
	phosphate hydrogen			
NaOH	Sodium hydroxide			
NaTDC	Sodium taurodeoxycholate			
ND	Not determined			
(NH4)2CO3	Ammonium carbonate			
nm	Nanometre			
PA	Palmitic acid			
PC	Phosphatidylcholine			
PG	Phosphatidylglycerol			
рН	Potential of hydrogen ion			
PLRP2	Pancreatic lipase-related protein 2			
POD	Peroxidase			
PPE	Porcine pancreatic extract			
PTFE	Polytetrafluoroethylene			
PUFA	Polyunsaturated fatty acid			
R2	Coefficient correlation			
RCF	Relative centrifugal force			
RGE	Rabbit gastric extract			

rpm	Revolution per minute
Rubisco	Ribulose-1,5-bisphosphate carboxylase
s	Second
SD	Spray drying
SDS	Sodium dodecyl sulphate
SDS-PAGE	Sodium Dodecyl Sulfate-Polyacrylamide
	Gel Electrophoresis
SEM	Scanning electron microscope
SGF	Simulated gastric fluid
SIF	Simulated intestinal fluid
SJ	Spray dried juice
SQDG	Sulfoquinovosyl diacylglycerol
SSF	Simulated salivary fluid
Std	Standard
TAG	Triacylglycerol
TEA	Triethylamine
TEM	Transmission Electron Microscope
TGDG	Trigalactosyl diacylglycerol
TMSH	Trimethylsulfonium hydroxide
U	Unit
UV-Vis	Ultraviolet and visible
v	Volume

xvi

VE	Vitamin E
viz	videlicet
Vol.	Volume
vs	Versus
WLM	Whole leaf material
WSI	Water solubility index

CHAPTER I

1. General introduction

1.1. General overview

Spinach (*Spinacia oleracea* L.) is an economically important leafy vegetable crop in many countries, with around 30 million tonnes of spinach produced worldwide in 2020/2021 (Table 1.1). China is the largest global producer that accounts for more than 90% of the world total spinach production, followed by United States of America, with 367,433 tonnes of spinach produced annually.

Table 1.1: Countries with highest spinach production in 2020/2021 ("Spinachproduction in 2021; Crops/Regions/World/Production Quantity/Year from pick lists,"

2023) (Food and Agriculture Organisation of the United Nations, https://www.fao.org/faostat/en/#data/QCL/visualize)

Rank	Country	Production	Acreage	Yield
	Country	(Tonnes)	(Hectare)	(Kg/Hectare)
1	China	28,507,829	724,331	39,357
2	United States of America	367,433	22,743	16,156
3	Kenya	243,336	9,789	24,858
4	Türkiye	231,515	18,482	12,527

In other developed countries, approximately 80% of the available spinach is processed and is available as either canned food or as frozen food, while fresh spinach is usually sold loose, bunched, or packaged fresh in bags (Hodgson et al., 1984). However, in many developing countries like South Africa, farming of vegetables such as spinach is needed to provide foods that offer opportunities in tackling human nutrients deficiency (Mdoda et al., 2022). Currently, approximately 1 billion people are unable to fulfill their basic energy requirements, and at least 2 billion people are experiencing micronutrient deficiencies (Müller et al., 2005; Berti et al., 2014). According to the available information (Bai et al., 2022), a significant upsurge in nutritional deficiencies and associated health issues has been observed in many countries, and the development of spinach products could aid dealing with this problem. Indeed, researchers have proposed that spinach is a novel non-conventional functional ingredient, with potential to mitigate micronutrient deficiencies in developing populations (Khan et al., 2015).

Indeed, spinach is considered as a functional food due to its nutritional properties, since it contains various macronutrients, such as lipids, proteins, minerals and carbohydrates, as well as micronutrients including carotenoids, chlorophylls, vitamins and phenolic compounds (Bunea et al., 2008). A typical serving of spinach (100 g) contains high levels of iron, potassium and magnesium, that meet 15%, 16% and 20%, of the recommended dietary allowance (RDA). Therefore, the consumption of spinach in the human diet may help in preventing a variety of nutritional related disorders and therefore serves to lower the relative risk of certain types of chronic diseases, such as cancer, diabetes, and cardiovascular disease (Abu Al-Qumboz et al., 2019). The relationship between diet and these physiopathologies has arose interest in design of spinach functional foods as cost-effective means of health requirements.

It is known that spinach is a rich source of chloroplasts. These organelles are found

in photosynthetic organisms such as plants and algae and contain many nutritionally important molecules. Recent research has shown that the chloroplast synthesises molecules, such as omega-3 fatty acids, amino acids (mainly glutamate, aspartate, and threonine), tocopherols, carotenoids, and plant hormones (Gross, 1991). However, despite the well-known nutritional value of spinach, it is a highly perishable crop with a short shelf life, fresh spinach loses most of its folate and carotenoid content over one week refrigerated storage (Batziakas et al., 2020). Chen et al. (1983) reported that folate levels in fresh spinach decreased by 26 % and 27 % after 7 d of storage at 4 °C or 10 h at 20 °C. Pandrangi et al. (2004) reported that total carotenoids retained 54%, 61%, and 44% of initial levels after 8 d (4 °C), 6 d (10 °C), and 4 d (20 °C) of storage. For longer preservation, it is canned, dehydrated or cooked and frozen.

Because of the instability of fresh spinach, the focus of our laboratories (School of Biosciences, University of Nottingham) has been on the extraction of chloroplast rich fractions from green plants including spinach, pea vine haulm, kale, and nettle, to provide a source of essential micronutrients that are shelf stable. Figure 1.1 shows chloroplasts in situ on spinach leaves as well as in an extract.



Figure 1.1: Light microscope images of a cross section of spinach leaves (A) and fresh spinach chloroplast rich fraction (B). The scale bar corresponds to 50 µm.

The composition of freshly extracted juice/CRF from green biomass has been measured to determine the stability of micronutrients during processing and storage. In addition, methods to access the bioaccessibility of lipophilic nutrients (carotenoids, fatty acids and alpha-tocopherol) in chloroplast extracts have been developed using *in vitro* digestion models (Gedi et al., 2017; Gedi et al., 2019; Torcello-Gómez et al., 2019; Wattanakul et al., 2020; Wattanakul et al., 2022).

1.2. Literature review

1.2.1.Chloroplast

Chloroplasts are characteristic organelles in plant and green algae cells (Marchand et al., 2018). Chloroplasts are lens-shaped organelles with a thickness and diameter of roughly 1–4 μ m and 5–10 μ m, respectively (Barber, 1976; Hall et al., 1999). The chloroplast consists of a double membrane with outer and inner envelopes, separated by the intermembrane space (Figures 1.2 and 1.3) (Karim et al., 2014). The space within the inner membrane is known as the stroma, a matrix containing various enzymes including the key photosynthetic enzyme ribulose-1,5-bisphosphate carboxylase/oxygenase (Rubisco), plastoglobulus (pl. plastoglobuli), ribosomes, starch granules, and copies of the chloroplast genome (Staehelin, 2003; Morita et al., 2008). The thylakoid membrane is an additional internal membrane, located in the stroma, and

characterized by its disk like structure. The continuous aqueous phase is enclosed by the thylakoid membrane (Sakamoto et al., 2008). In most higher plants, the thylakoids are arranged in tight stacks called grana, that are connected by the stroma thylakoids/lamella (Pribil et al., 2014; Staehelin et al., 2020).



Figure 1.2: Structure of chloroplast.(Chen et al., 2014)

Chloroplasts are the main site of photosynthesis, a process that converts light energy into chemical energy, and then leads to the production of organic molecules such as glucose, sucrose and starch (Staehelin, 2003). Photosynthesis occurs at two sites for light-independent reactions (also called the Calvin-Benson cycle), (Figure 1.4), which takes place in the stroma and the thylakoid membrane. Rubisco is responsible for catalysing carbon dioxide fixation. The series of embedded pigments (chlorophylls and carotenoids) and proteins are essential to perform the light-dependent reactions of photosynthesis. The chlorophylls capture the energy from sunlight and convert it into chemical energy, such as adenosine triphosphate (ATP) and nicotinamide adenine dinucleotide phosphate (NADPH). The Calvin Cycle uses the ATP and NADPH from the light-dependent reactions plus carbon dioxide to produce organic molecules such as glucose and starch (Raines, 2003; Krupinska et al., 2004).



Figure 1.3: Thin section electron micrograph of a chemically fixed chloroplast. The chloroplast lies flat against the plasma membrane and the cell wall (CW) and shows a more or less elliptical structure. The stacked grana thylakoids (GT) are interconnected by non-stacked stroma thylakoids (ST). Stroma (S) surrounds the membranes, and the lightly stained region of the stroma indicates the presence of DNA. A few plastoglobuli lie adjacent to stroma thylakoids. Two envelope membranes (EM) form the boundary layer of the chloroplast. The scale bar corresponds to 0.5mm. (Staehelin & Paolillo, 2020).

In additional to photosynthetic processes, chloroplasts also have other specialized functions crucial for plant growth and development including nitrate and sulphate assimilation, and various biosynthetic pathways linked to plant secondary metabolite metabolism (Jensen et al., 2014). It is well know that chloroplast synthesises numerous essential compounds including hormones (auxin, gibberellin, cytokinin, ethylene, and abscisic acid), proteins (amino acids), fatty acids (C16 and C18 fatty acid), lipids (galactolipids and phospholipids) and secondary metabolites (alkaloids and terpenoids) (Joyard et al., 1998; Verpoorte et al., 1999). Thus, chloroplast can be considered as a significant source of macro- and micro- nutrients in plants.



Figure 1.4: Photosynthesis reaction takes place in thylakoid membranes. (Sekar et al., 2015)

1.2.2.Nutrient profiles of plant chloroplast and their importance in human health

1.2.2.1. Lipids

The major lipids contained in the chloroplastic photosynthetic membrane are galactolipids, which account for around 70-80 % of the total membrane lipids (Dörmann, 2013). The lipid bilayer of the thylakoid membrane in plant chloroplasts and cyanobacterial cells is predominantly composed of four unique lipid classes; Their chemical structures are characterized by the number of galactose moieties attached to

glycerol backbone at the SN-3 position. The main form of galactolipids present in the chloroplast are monogalactosyldiacylglycerol (MGDG) and digalactosyldiacylglycerol (DGDG), containing one or two saturated and/or unsaturated fatty acids linked to the glycerol moiety, and account for approximately 50 % and 20-30 % of total glycerolipids respectively (Christensen, 2009). In addition, levels of lower sulfoquinovosyldiacylglycerol (SQDG) and trigalactosyldiacylglycerol (TGDG), a sulfolipid with a sulfoquinovose residue instead of galactose, are found (Amara et al., 2010). Various phospholipids are also contained in the chloroplastic membrane system, in the form of phosphatidylglycerol (PG), phosphatidylcholine (PC) and phosphatidylinositol (PI). Furthermore, under stress conditions various neutral lipids (triacylglycerol and fatty acid phytyl esters) accumulate in chloroplasts and are mostly found in plastoglobuli; small lipid droplets in the stroma enclosed by a lipid monolayer membrane (Wijk et al., 2017). These lipids are crucial for maintaining the function of chloroplasts as they consist of a large fraction of the photosynthetic membranes and are believed to play important roles in photosynthesis and thylakoid membrane biogenesis.



Figure 1.5: Chemical structure of galactolipids in chloroplasts (Wattanakul et al., 2019)

A constant ratio of MGDG to DGDG in thylakoid membranes is essential for maintaining the structure and stability of the photosynthetic membranes (Makshakova et al., 2020). The MGDG/DGDG ratio response under different abiotic stresses contribute to maintain membrane structures and enzymatic activities (Kobayashi, 2016). In spinach, the lipid composition of the thylakoid membrane consists of 53 % MGDG and 27 % DGDG and low levels of anionic galactolipids SQDG (7%) and anionic phospholipid PG (7 %). In the inner envelope membrane, the galactolipids account for 84 % (MGDG 49 %, DGDG 30 % and SQDG 5 %) and 15 % phospholipid (PG 8%, PC 6% and PI 1%). Moreover, the outer envelope membrane has a lipid composition of

52% galactolipids (MGDG 17 %, DGDG 29% and SQDG 6%) and 47 % phospholipid (PG 10 %, PC 32% and PI 5% (Table 1.2), (Block et al., 1983; Dorne et al., 1990).

Table 1.2: Membrane lipid composition of spinach (% fresh weight) (Block et al., 1983;Dorne et al., 1990)

	MGDG	DGDG	SQDG	PG	PI	PC	Others
TM	53	27	7	7	2	0	4
IEM	49	30	5	8	1	6	1
OEM	17	29	6	10	5	32	1

Abbreviations: thylakoid membrane (TM), inner envelope membrane (IEM) and outer envelope membrane (OEM), phosphatidylglycerol (PG), phosphatidylcholine (PC) and phosphatidylinositol (PI).

Chloroplasts are the major site for de novo fatty acid synthesis (Hölzl et al., 2019). Palmitic acid (16:0) and unsaturated C18 fatty acids such as oleic acid (18:1, 9-cis), linoleic acid (18:2, 9-cis, 12-cis) and α -linolenic acid (18:3, 9-cis,12-cis,15-cis) are highly abundant in chloroplasts. Unsaturated C16 fatty acids are found in PG (3-transhexadecenoic acid, 16:1, 3-trans) and MGDG (hexadecatrienoic acid, 16:3, 7-cis,10cis,13-cis) such as *Arabidopsis* and spinach (*Spinacia oleracea*). A large amount of isoprenoid lipids can also be found in chloroplasts which include chlorophyll, carotenoids, and prenylquinols (plastoquinol-9, phylloquinol, tocopherol) (Schultz, 1990; Zhai et al., 2013).

1.2.2.2. Proteins

In addition to lipids, chloroplasts also contain large amounts of protein. Indeed,

chloroplast proteins are main compounds in plant cells, which represents up to 40 % of total plant cellular protein in leaves. The envelope contains only 1 % of the chloroplast proteins, and the majority of proteins are located in the stroma (40-50 %) and thylakoid membranes (40-50 %) (Bouchnak et al., 2019). The stroma and thylakoid membrane are the main sites for conducting photosynthesis. The thylakoid contains two photosystems which is Photosystem I (PSI) and Photosystem II (PSII). These photosystems are responsible for generating the energy needed to produce carbohydrates in the Calvin cycle. The protein light harvesting complexes LCI and LCII embedded in the photosynthetic membrane are responsible for capturing sunlight. These proteins bind to a network of chlorophyll and carotenoid molecules are the pigment-protein complexes acting as LHCs (light harvesting complexes) that absorb photons of light and effectively leading to the electronic excitation with the excitons eventually arriving at the reaction centers (Lokstein et al., 2021; Zayadan et al., 2021). In addition, the major protein in chloroplast and the envelope membranes is Ribulose-1,5-bisphosphate carboxylase-oxygenase (Rubisco) (Figure 1.6). This is a critical enzyme in the Calvin cycle in the stroma for the light-independent reactions (Ferro et al., 2000; Bouchnak et al., 2018).



Figure 1.6: Coomassie blue-stained SDS-PAGE gel of chloroplast protein subfractions. Rubisco large subunit (RBCL), Triose-phosphate translocator (TPT), Light harvesting complex protein (LHCP). (Bouchnak et al., 2018)

1.2.2.3. Minerals

Chloroplasts are rich sources of minerals such as magnesium (Mg), calcium (Ca), sodium (Na), phosphorus, potassium (K), sulphur (S), chloride (Cl), and zinc (Zn). These minerals are essential for numerous processes taking place in our bodies. They play essential roles in maintaining a healthy body, including blood, bone and teeth health, and only can absorbed through diet. Mg is bound or incorporated into cellular compartments (Waters 2011), with the highest concentrations in chloroplasts (Karley and White 2009). A significant amount of Mg in leaves is bound as the central atom in the tetrapyrrole ring of chlorophyll a and b molecules, which are the major pigments for photosynthetic light capturing. Both K^+ and Mg^{2+} can facilitate the organization of the grana and stroma lamellae, thus support chloroplast integrity, the efficiency of light

absorption, Rubisco diffusion and carbon assimilation (Tränkner et al., 2018). Calcium (Ca^{2+}) plays a significant role in maintaining both structural and regulatory processes in chloroplast stroma and thylakoids (Miqyass et al., 2007; Wang et al., 2016). Ca²⁺ is a structural component of photosystem II (PSII) and is essential for efficient oxygen evolution. At low concentrations Ca²⁺ can activate fructose 1,6-bisphosphatase, a key enzyme involved in the Calvin cycle responsible for sucrose synthesis pathway (Rakus et al., 2013). Interestingly, high concentration of Ca²⁺ in stroma tends to inhibit CO₂ fixation in photosynthetic reactions (Wang et al., 2019).

The major mineral composition of green leafy plants like spinach, kale, nettles and grass is shown in Table 1.3 (Gedi et al., 2017). It can be found that the CRF from green leafy plants are especially rich in minerals such as magnesium, phosphorus, potassium, and calcium. In the United Kingdom, the reference nutrient intake (RNI) of minerals is shown in Table 1.4.

Table 1.3: Major minerals (mg/100g DW) in freeze-dried chloroplast rich fraction(CRF) of green leafy plants (Gedi et al., 2017)

	Na	Mg	Р	K	Ca	Mn	Fe	Cu
Spinach	21±5	337±24	298±42	1196±68	677±15	7.7±0.6	31.4±3.1	2.0±0.1
Kale	39±11	322±17	628±111	518±95	920±30	6.5±0.7	25.7±9.1	0.5±0.1
Nettles	63±2	396±19	370±54	872±64	1736±11	7.9±0.6	42.1±12	1.0 ± 0.1
Grass	44±2	232±28	436±94	1195±234	354±4	12.0±3.4	37.0±0.6	1.1±0.4

Data were expressed as the mean \pm SD of 3 separate experiments.

Minerals	Males	Females
Na	1600	1600
Mg	270	300
Р	550	550
К	3500	3500
Ca	700	700
Cl	2500	2500
Fe	14.8	8.7
Cu	1.2	1.2
Zn	7.0	9.5
Se	0.06	0.075
Ι	0.14	0.14

Table 1.4: Reference Nutrient Intake (RNI) for minerals in the United Kingdom (mg/day) (Derbyshire, 2018)

1.2.2.4. Carotenoids

Carotenoids are lipophilic pigments characterized by their diverse colours in plant tissues spanning bright yellow, red and orange colourations. They are highly abundant in plants, algae, photosynthetic bacteria and some fungal species. Carotenoids synthesis in plants and play important function in photosynthesis and biomembrane photoprotection (Swapnil et al., 2021).

In addition, carotenoids are essential components in the diets of humans as the source of antioxidants and vitamins. Commercially, they are used as natural colourants, and nutritional ingredients in food supplements (Li et al., 2012). More than 600 carotenoids are found in natural structural variants which are classified into two main groups: carotenes and xanthophylls (Figure 1.7), (Ellison, 2016). Carotenes, such as α -

carotene, β -carotene and lycopene, are hydrocarbons. Xanthophylls, such as β cryptoxanthin, lutein, zeaxanthin, are carotenoids containing oxygen atoms (Maoka, 2020). Carotenoids consist of eight repetitive units of isoprene with C40 chain backbone. The cyclic or linear structures at both ends of the carbon chains, resulting in multiple *cis* and *trans* isomers, with the *all-trans* isomeric configuration exist more abundant and stable in nature (Milani et al., 2017). Carotenoids all have antioxidant properties, and some are precursors of vitamin A (e.g. β -carotene, α -carotene and β cryptoxanthin). Moreover, carotenoids are involved in some beneficial functions including immune system activation, intercellular communication, and disease prevention and therefore enhance human health (González-Peña et al., 2023). The consumption of carotenoid abundant foods is highly recommended since it is linked with the lower risk of chronic diseases. However, according to the existing evidence collected by the European Food Safety Authority (2006), it is insufficient to establish a Recommended Dietary Allowance (RDA) or Adequate Intake (AI) for β-carotene and other carotenoid.



Figure 1.7: Chemical structures of carotenoids (Ellison, 2016)

 β -carotene is the most common form of carotenoid found in the diets of humans and is a precursor of vitamin A. In biological systems, the predominant isomer is the all-trans β-carotene (E-isomer) and to a lesser extent, the cis-isomers including 9-cis, 13-cis and 15-cis-β-carotene (Z-isomers) (Figure 1.8) (Grune et al., 2010; Pénicaud et al., 2011). All trans-β-carotene is highly unstable and isomerized into cis-isomers. Trans-β-carotene immediately undergoes thermal and chemical oxidation, isomerization, and photosensitization when exposed to oxygen, light, and high temperature during processing and storage (Rodriguez-Amaya, 1999). The main sources of β -carotene are fruits and vegetables like tomato, pumpkin, asparagus, carrot, broccoli, cabbage, spinach and lettuce (Weber et al., 2012). β -carotene is believed to have many health benefits due to its anti-cancerous and antioxidant properties. The consumption of β -carotene in the diet of humans has the potential to lower the risk of heart diseases and certain types of cancers, enhancing the immune system, and protection from age-related macular degeneration (Gul et al., 2015).



Figure 1.8: Chemical structures of isomers of β -carotene (Pénicaud et al., 2011).

Lutein is a type of xanthophyll carotenoids found particularly in dark-green leafy vegetables (e.g., kale, spinach, broccoli, peas and lettuce), fruits and egg yolks (Perry et al., 2009). Lutein and zeaxanthin are the main dietary carotenoids found in human retina. Numerous studies indicating that xanthophyll intake or status has been linked with reduced risk of age-related macular degeneration (AMD), cataracts, certain types of cancer, particularly those of the breast and lung, and the prevention of heart disease and stroke (Ribaya-Mercado et al., 2004; Abdel-Aal el et al., 2013). Their beneficial health effects due to the ability to act as scavengers for reactive oxygen species and to bind with physiological proteins in humans (Mitra et al., 2021).

The nutritional credential of chloroplasts recovered from plants are found by (Gedi et al., 2017), that higher concentration of carotenoids is contained in CRF. Therefore, chloroplasts can be potentially used as a source of carotenoid in food formulations.

The composition of different forms of carotenoids in some important vegetables is shown in Table 1.5. These data will be helpful for the selection of high nutritional vegetables for food formula design and diet recommendation (Saini et al., 2015).

Vegetable Lutein	T / '			β -Carotene	Total
	Lutein	Zeaxanthin	<i>β</i> -Cryptoxantnin		carotenoids
Broccoli	28.05	-	0.15	11.38	46.46
Kale	65.22	-	0.21	44	132.9
Lettuce	13.5	-	-	14.9	-
Spinach*	775.8	15.1	-	365.3	2386.2

Table 1.5: Dietary sources of carotenoids from vegetables ($\mu g/g$) (Saini et al., 2015)

Values are expressed as $\mu g/g$ fresh weight (*values are expresses as $\mu g/g$ dry weight).
1.2.2.5. Chlorophylls

Chlorophylls are lipid-soluble green pigments widely distributed in plants; they function in photosynthesis. These molecules are located in Photosystems I and II within the thylakoid membrane in chloroplasts (Mauzerall, 1976; Staehelin, 2003). Structurally chlorophylls are composed of a central magnesium atom surrounded by a nitrogen-containing structure called a porphyrin ring; attached to the ring is a long carbon-hydrogen side chain, known as a phytol chain (Grimm, 2001). To date, several structural variants of chlorophylls have been characterized in higher plants, including chlorophyll a and b (Gaur et al., 2006); the chemical structures are shown in Figure 1.9 (Schwartz et al., 1990). Chlorophyll a appears blue-green in colour and chlorophyll b shows yellow-green (Rodriguez-Amaya, 2019). Chlorophyll a contains a methyl constituent at the C-3 carbon atom, and chlorophyll b this is replaced by a formyl group (Schwartz et al., 2017). Chlorophylls are unstable and readily degrade during thermal treatment (Tan et al., 1962). Indeed, during thermal processing, chlorophyll degradation studies revealed that chlorophyll loses its bound magnesium and transform to other compounds known as pheophytins, chlorophyllides and pheophorbides (Schwartz et al., 1983). The colour changes from bright green to olive brown during this process which ultimately cause the loss of quality in food products. The mechanism of chlorophyll degradation is shown in Figure 1.10.

Chlorophyll and its derivatives can be used as a source of antioxidant nutrients and alternative medicine (Sakata et al., 1990; Hoshina et al., 1998; Ferruzzi et al., 2002). After release from the plant food matrix, natural chlorophyll derivatives are converted to respective metal-free pheophytins under the acidotic conditions of the gastric phase during digestion. The chlorophyll derivatives are then absorbed by intestinal cells and eventually passes into blood circulatory system (Mishra et al., 2011). Chlorophyll and derivatives have potent antioxidant and radioprotective effects *in vitro* and *in vivo*. They have the ability to inhibit lipid peroxidation, protein oxidation, DNA and membrane damage (Kumar et al., 2001; Taïbi et al., 2016). Several *in vitro* and animal studies have shown that chlorophylls have anti-inflammation, anti-cancer, simulating immune system, normalize blood pressure and wound healing (Gaur et al., 2006; Mishra et al., 2011). With the well-known health benefits of chlorophylls and colour preference in determination of product acceptability, it is important for the food industry to minimise its loss during processing.



Figure 1.9: Chemical structures of chlorophylls (Schwartz & Lorenzo, 1990)



Figure 1.10: Mechanism of chlorophyll degradation (Schwartz & Von Elbe, 1983)

1.2.2.6. Vitamin E

There are eight naturally occurring forms of vitamin E which includes four tocopherols (α , β , γ and δ) and four tocotrienols (α , β , γ and δ), shown in Figure 1.11 (Reboul, 2019b). Alpha-tocopherol is the most abundant form of vitamin E and has the highest biological activity (Burton et al., 1990; Brigelius-Flohé et al., 1999; Reboul, 2017). The antioxidant activity of vitamin E has been linked to numerous health benefits which effectively reduce risk of some diseases, including cancer, ageing, arthritis, oxidative stress, cataracts, immune system disorders, and Vitamin E deficiency (Meydani et al., 1998; Gaziano, 2004; Lippman et al., 2009; Rizvi et al., 2014; Traber, 2014). The Recommended Dietary Allowances (RDAs) for vitamin E (alpha-tocopherol) is 15 mg per day for adults (Kowdley et al., 1992).



Figure 1.11: Chemical structures of tocopherols and tocotrienols (Reboul, 2019b)

Vitamin E is found in various foods and oils. Nuts, seeds and vegetable oils contain high amounts of alpha-tocopherol, which is around 500-2000 μ g/g of fresh weight (FW) (Hirschberg, 1999). And significant high amounts of α -tocopherol are also available in green leafy vegetables and fortified cereals (Miyazawa et al., 2011). It has been reported that more than 90% of total tocopherols are found in the form of α -tocopherol in green leaves such as spinach (16.2 μ g/g, FW), parsley (47.51 μ g/g, FW), and also found in the leaves of tree, including birch, maple, elder, pine, aspen, and juniper (118-519 μ g/g, FW) (Szymańska et al., 2008).

1.2.3.Food preservation technologies

Vegetable products consumption is highly depending on changes in pigments, nutritional value, and visual appearance during storage. The fresh vegetables are highly perishable and have a relatively short shelf life. The degradation of vegetable products can be summarised with the following types of reaction: microbiological degradation, hydrolytic reaction, oxidative reaction and enzymatic/non-enzymatic browning reactions (Mounir et al., 2017). As descried earlier, spinach is rich in micronutrients such as carotenoids, chlorophylls, vitamins as well as macronutrients such as proteins and lipids. These nutrients are sensitive to heat, light, oxygen, pH and enzymes (Mehta, 2015). During storage, chlorophylls are susceptible to chemical and enzymatic changes and eventually degrade into olive green catabolism products, including pheophytins, pheophorbides. Major chemical degradation routes are pheophytinization, epimerization, pyrolysis, hydroxylation, oxidation or photo-oxidation (Mangos et al., 1997). Carotenoids also show poor stability in vegetable products. The degradation of carotenoids is accelerated by intermediate radicals in plant as a result of autoxidation and lipid peroxidation (Liu et al., 2015). The isomerisation of carotenoids happens under high temperature and leads to the colourless products (Zeb, 2012). Lipid undergoes lipolysis and enzymatic oxidation reaction during storage and degrades to unsaturated fatty acids (Gyamfi et al., 2019).

Meanwhile, the enzymatic and non-enzymatic browning are also major degradation affecting vegetable product quality. The polyphenol oxidases (PPO) and peroxidase (POD) are main oxidative enzymes in enzymatic oxidation which are responsible for the changes in flavour, texture, colour and nutritive properties of vegetables (Hamdan et al., 2022). Enzymatic browning occurs in fresh-cut vegetables, where the damaged tissues allow the interaction of phenolic compounds with PPO and exposed to oxygen, consequently results in a rapid browning reaction. So far, various techniques have been developed to control the enzymatic browning of fresh-cut fruits and vegetables, including dips in anti-browning solutions, modified atmosphere packaging (MAP), heat treatment and refrigeration storage (He et al., 2007). Non-enzymatic browning (NEB) occurs in fruits and vegetables due to the interaction of aldehyde and ketone group of reducing sugars with amino acids and proteins (Mitra et al., 2015). The Maillard reaction have been identified as the main non-enzymatic reaction pathways. The NEB reaction is favoured by high heat treatment and water activity. Therefore, it can be inhibited by lowering temperature during the drying process to achieve desired high quality dehydrated food products (Manzocco et al., 2000).

Moisture content and water activity are also crucial factors in microbial growth of food products. Moisture content is simply considered as the total water content in food products. While, water activity is defined as the ratio of the water vapor pressure of a food system to the saturated water vapor pressure at a specific temperature. High value of moisture content and a_w of food products increase the risk of microbiologically growth (Alegbeleye et al., 2022). To avoid microbial growth, the moisture content and water activity must be kept below approximately 10% and 0.60, respectively (Wason et al., 2021). However, fresh foods are commonly high in moisture content and water

activity nearly 0.99 which are particularly prone to microbial growth (Vera Zambrano et al., 2019). Therefore, drying is an ideal food preservation technique which can efficiently remove the total moisture of food products and increase shelf stability (Calín-Sánchez et al., 2020).

Food preservation is defined as the processes or techniques undertaken in order to mitigate internal and external factors which may cause food spoilage (Amit et al., 2017). The principal objective of food preservation is to extend its shelf life retaining original nutritional values, colour, texture, and flavour, and improve consumer acceptability. Different food processing steps are involved to maintain food quality with maximum nutrition values and minimum side effects. Conventional food preservation technologies such as drying, freezing, pasteurisation and sterilization is widely used (Tavman et al., 2019). Recently, food preservation methods include thermal, electrical, chemical and radiation techniques have been emerged in food industry (Sridhar et al., 2021), heat transfer, moisture removal, and prevention of enzymatic and chemical reaction was included. Here, we review the advanced food preservation techniques, which focuses on green leafy plants (i.e. spinach) processed with thermal/nonthermal and spray/freeze drying methods. During which processing, enhances the shelf stability and improves nutritional and physicochemical properties.

1.2.3.1. Spray drying

Spray drying (SD) is a method of particle production which comprises the transformation of a fluid material into dried particles with hot drying medium (Rattes et al., 2007). During the 1880s-1920s, spray drying equipment and techniques was

developed and applied in industry for the need to reduce the transport weight of foods and other materials during World War II (Patel et al., 2009). Nowadays, spray drying is widely applied in food and pharma industries, for the production of food powders, nutraceuticals, aromas, probiotics, enzymes, antibiotics, blood plasma, organic and inorganic chemicals, ceramic powders, detergents and fertilizers (Samborska et al., 2022). Spray drying is a thermally efficient process maintaining quality of a product by rapid dehydration. It provides a large surface area in the form of fine liquid droplets through atomization in the drying chamber, which leads to the production of regularly and spherically shaped powder particles (Fazaeli et al., 2012). Spray drying does have several advantages over other production processes such as freeze drying, since it is faster and cheaper (Sagar et al., 2010; Eun et al., 2020).

The spray drying process consists of four basic steps: (i) atomization of the liquid feed, (ii) droplet-hot air contact, (iii) water evaporation, and (iv) drying gas-powder separation and collection (Sagar & Suresh Kumar, 2010). As shown in Figure 1.12 a schematic is presented for a conventional spray drying process (Patel et al., 2009). Recently, spray drying has been widely applied in the production of fruit and vegetable juice powders for use in food and the pharmaceutical industry (Tontul et al., 2017).

The spray-dried vegetable products are highly stable, because of its low moisture content (2–5%), and water activity (0.2–0.6) (Patil et al., 2014). However, the qualities of spray dried products such as hygroscopicity, morphology, particle size and yield, is highly dependent on the different processing factors during operating. These factors including inlet temperatures, feed flow rate, air dry flow rate, atomizer speed, carried

agent (Phisut, 2012). Moreover, the inlet air temperature appreciably affects the stability of carotenoids in spray dried food products. The reduction of carotenoids content is presumably due to the thermal degradation and oxidation, when expose to oxygen during drying (Goula et al., 2005). Therefore, it is important to optimize the various processing factors prior to spray drying.



Figure 1.12: Spray dryer configuration (Patel et al., 2009). The laboratory scale spray dryer used in these experiments was FT 80 Tall form with a two-fluid atomizing nozzle.

1.2.3.2. Freeze drying

Freeze drying (FD), also known as lyophilization is a well-known technique for the production of high-quality food powders and solids (Figure 1.13) (Nowak et al., 2020). It is the preferred method for drying foods containing thermally sensitive and oxidative compounds since it operates at low temperatures and under high vacuum (Bhatta et al., 2020). The freeze dried products have many benefits, such as nutrients stability during storage due to low moisture content, reduction of nutrients degradation in heat sensitive products, minimally changed structures by the process (Adams, 2007; Nireesha et al., 2013). FD has been applied to various plant-based foods, such as apple, blackberry, coffee, guava, strawberry, pumpkin, garlic, and ginger (Marques et al., 2006; Sablani, 2006; Ciurzyńska et al., 2014; Fissore et al., 2014; Franceschinis et al., 2014; Fante et al., 2015; An et al., 2016). The advantage of FD compared to SD is that it retains nutritional value better while preserves the actual colour and shape of the original raw material, that supporting consumers' demand for whole healthy food (Serna-Cock et al., 2015).



Figure 1.13: Construction of freeze dryer (Garcia-Amezquita et al., 2016)

FD is a process in which water in the form of ice under low pressure is removed from a material by sublimation. It has become an important technology for heatsensitive drugs and food preservation (eg. fruits, vegetables, meat) (Liu et al., 2022). The FD process can be divided into three steps, (a) freezing, where the sample should be completely frozen, (b) primary drying, when ice is sublimated, usually at subatmospheric pressure, and (c) secondary drying, when the remaining unfrozen/bound water is desorbed from the drier food matrix. Water exists in three different states: solid, liquid, or gas (vapor). As shown in Figure 1.14, a phase diagram of water (pressure versus temperature), where the curve lines show the passage from solid to vapor (sublimation), from liquid to vapor (evaporation), or from solid to liquid (fusion) during freeze drying (Gaidhani et al., 2015; Nowak & Jakubczyk, 2020).



Figure 1.14: Phase diagram of water (T: triple point of water, C: critical point of water). "A" represents the starting point prior to freeze-drying (atmospheric pressure and ambient temperature), while "B", the desired final conditions during sublimation (below the triple point T) (Bhatta et al., 2020).

1.2.4. Thermal pre-treatment

Thermal pasteurisation and blanching are predominantly used in the food industry for inactivation of endogenous enzymes of plants and extend its shelf life after harvesting. However, excessive heat treatment may cause undesirable quality loss of food products, such as protein denaturation, non-enzymatic browning and loss of vitamins and volatile flavour compounds (Torcello-Gómez et al., 2019).

1.2.4.1. Pasteurisation

Pasteurisation is a thermal treatment which widely accepted as an effective preservation method for killing pathogens in food products, with minimal loss of desired food quality (Ramesh, 2020). It usually refers to a mild heat treatment (temperature<100°C), designed mainly for the inactivation of spoilage enzymes and vegetative forms of micro-organisms (bacteria, moulds, and yeasts) (Lund et al., 1988).

However, even a mild thermal process tends to cause a significant loss of quality in terms of colour, texture, flavour, and nutritive value. Thus, it is necessary to optimize a time-temperature combination for thermal processing (Maesmans et al., 1990; Grant et al., 2005). There are two main types of traditional pasteurisation (excluding ultrahigh temperature or UHT method) i.e. low-temperature-long time (LTLT) and hightemperature-short-time (HTST) methods (Urquieta-Herrero et al., 2021). Selection of the optimal process is usually dependent upon which will provide the best sensory properties and overall quality for the pasteurised food. There are many factors that need to be considered, such as the nutrients retention, types of the food, type of microorganisms, and endogenous enzyme activities (Aamir et al., 2013).

Vegetables are generally classified as low acid foods (pH \ge 4.6), more severe heat treatment is needed to kill spores of pathogenic and spoilage microbes (McGlynn, 2003). However, one consequence of heating is that heat sensitive compounds contained in vegetables such as ascorbic acid, thiamine is reduced during thermal processing (Akinyele et al., 1990; Leistner et al., 1994). Moreover, the presence of residual endogenous enzymes in processed vegetable products may cause further

quality loss during storage and shorten its shelf life. Thus, the complete inactivation of endogenous enzymes is used as a target for an adequate thermal treatment. Major enzymes related to vegetable quality attributes (texture, colour and flavour) are polyphenol oxidase (PPO), peroxidase (POD) and lipoxygenase (LOX) (Silva et al., 2004). Typically, POD is used as an indicator for the effectiveness of heat treatments in fruits and vegetables because this enzyme is the most heat resistant enzyme presented in most plant tissues (Gonçalves et al., 2010; Ercan et al., 2011; Zheng et al., 2011). The assumption is that destruction of POD is accompanied by inactivation of all other enzymes and vegetative micro-organisms (Whitaker, 2018).

1.2.4.2. Blanching

Blanching is a thermal treatment conducted in hot water, steam or microwave, that is commonly used as a pre-treatment prior to further processing (canning, freezing, dehydrating) of many vegetables (Peng et al., 2017). The purpose of blanching is to inactivate endogenous enzymes, preserve colour, flavour, and nutritional value and release the entrapped air within vegetable cells and replaces them with water. Water blanching also involves two process namely, low-temperature long-time (LTLT) or high-temperature short-time (HTST) processing (Aguilar et al., 2004). Typical blanching temperatures ranging from 70 to 100°C depending upon the product and processing conditions (Stanley et al., 1995). POD is chosen as the target for inactivation during the blanching process which indicates if product quality is to be maintained (Ramesh et al., 2002).

Other food quality attributes such as flavour, colour, texture and nutrients are

greatly affected by the type and extent of blanching (De Corcuera et al., 2004). Blanching indirectly and directly affects food flavour by removing bitter flavour and increase flavour retention (Velasco et al., 1989); the colour changes of green leafy plants during blanching is caused by the destruction of chlorophyll and other colouring pigments and Maillard browning due the presence of reducing sugars (Tijskens et al., 2001; Richter Reis, 2017). Blanching can also cause undesirable softening of vegetable tissues and higher leaching of some heat liable vitamins and minerals (Stanley et al., 1995; Severini et al., 2016).

1.2.5.*In vitro* digestion

Food digestion is a complex process and is of direct interest to the food industry since this process dictates the availability of nutrients associated with food quality and health benefits like reduced risk of chronic diseases (Bornhorst, 2016). For the purpose of understanding these changes, *in vitro* digestion methods have been developed. *In vitro* digestion closely simulates the physiological conditions of *in vivo* digestion with either static or dynamic models being commonly used. Several factors are considered in *in vitro* digestion models such as the occurrence and concentration of digestive enzymes, the pH values in gastric and intestinal phases, digestion time and salt concentrations (Minekus et al., 2014). These techniques are widely used in research fields such as nutrition, pharmacology and food chemistry since it is flexible, rapid, inexpensive and reproducible (Calvo-Lerma et al., 2018). To date, *in vitro* digestion models provide a useful alternative to animal and human models without limitations of ethical issues (Egger et al., 2017).

1.2.5.1. Static in vitro digestion model: INFOGEST

The INFOGEST in vitro digestion model is a standardized model first introduced by Minekus et al. (2014). An updated INFOGEST 2.0 model has been published with some modifications that better reflect *in vivo* digestion processes (Brodkorb et al., 2019). This model consists of three phases that including an oral, gastric and intestinal phase as shown in Figure 1.15. In the oral phase, the food should be diluted at 1:1 (wt/wt) with simulated salivary fluid (SSF), with or without salivary amylase (α -amylase, 75U/ml), incubated for 2 min at pH 7, 37 °C. α-amylase is used for starch digestion, as it can hydrolyse starch molecules into maltose and some glucose molecules. Thus, the α -amylase in the oral phase is selective and sometimes omitted by some researchers which mainly focus on the digestion of other components such as lipids and proteins. Following the oral phase, the gastric phase, the oral bolus is then mixed with simulated gastric fluid (SGF) (1:1, v/v) and gastric enzymes (pepsin 2,000 U/ml and gastric lipase 60 U/ml) and incubated at pH 3.0 for 2 h (37 °C). The addition of gastric lipase (for example, rabbit gastric extracts, RGE) is the main modification based on the INFOGEST 1.0 (Minekus et al., 2014). As lipid starts to digest in the stomach, and gastric lipolysis contributes to the overall digestion of TAGs (10% with a solid-liquid test meal to 25% with an emulsified liquid test meal). Moreover, it also triggers the subsequent action of pancreatic lipase on lipid substrates that may be poorly digested by pancreatic lipase alone (Brodkorb et al., 2019).

Finally, in the intestinal phase, the gastric chyme is mixed with simulated intestinal fluid (SIF) (1:1, v/v), porcine bile (bile salts 10mM) and porcine pancreatic extracts

(PPE) is added and incubated at pH 7 for a further 2 h (37 °C). In each phase of this model, it is possible to collect the processing fluids to assess nutrient liberation and stability.



Figure 1.15: Flow diagram of INFOGEST 2.0 digestion method (Brodkorb et al., 2019).

1.2.5.2. Static *in vitro* digestion model from Carriere et al. (2000)

Other static models of digestion have also been developed in recent times. The *in vitro* digestion model simulated meal lipid digestion was based on *in vivo* studies of gastric conditions from a half gastric emptying. As shown in Figure 1.16 and 1.17, the pH in half gastric meal emptying was around 5-5.5 for liquid meal and pH 4.5 for solid-liquid meal (Carrière et al., 2001; Sams et al., 2016). Generally, the use of pH 2-3 for gastric phase is set for most of *in vitro* models and mirrors the recently developed

INFOGEST system (Minekus et al., 2014). While this is more acidic than these in *in vivo* experiments, the pH 2-3 is correspond to approximately 85-95% of meal gastric emptying (Sams et al., 2016). Likewise, the setting of gastric lipase activity to 60 U/ml is correspond to approximately 86% of gastric emptying. In order to accurately mimic the gastric condition, the 50% of gastric emptying is considered, and the pH and gastric lipase will need to be set to pH 5 and 20.4 U/ml respectively.



Figure 1.16: Variations in gastric pH during test meal digestion in healthy volunteers. Panel A: pH variation as a function of time during a liquid test meal (256 values from 30 individual experiments); panel B: the same pH values plotted as a function of meal gastric emptying (%); panel C: pH variation as a function of time during a solid–liquid

test meal (545 values from 53 individual experiments); panel D: the same pH values plotted as a function of meal gastric emptying (%) (Sams et al., 2016).



Figure 1.17: Variation of the gastric lipase concentration in gastric contents (Human gastric lipase, [HGL]g) during test meal digestion in healthy volunteers. Panel A: [HGL]g variation as a function of time during a liquid test meal (256 values from 30 individual experiments); panel B: the same [HGL]g values plotted as a function of meal gastric emptying (%); panel C: [HGL]g variation as a function of time during a solid–liquid test meal (545 values from 53 individual experiments); panel D: the same [HGL]g values plotted as a function of meal gastric emptying (%) (Sams et al., 2016).

1.2.5.3. Bioaccessibility of nutrients

Key to our understanding of how to better improve foods and their nutritional

quality is the process of bioaccessibility. Bioaccessibility is a key concept to ascertain nutritional efficiency of food and food formula developed with the aim of improving human health (Fernández-García et al., 2009). It is defined as the proportion of a compound that is released from a food matrix in the gastrointestinal tract, that then becomes available for intestinal absorption and distributed systemically. The aforementioned *in vitro* digestion models are better facilitating our understanding of this process (Hur et al., 2011; Verhoeckx et al., 2015).

Plant based foods including fruits, vegetables are rich source of numerous types of phytonutrients such as vitamins, carotenoids, minerals, polyphenols, polyunsaturated fatty acids, proteins and dietary fibers that have a range of health benefits in humans (Espín et al., 2007). However, phytonutrient compounds must be first released from foods, and effectively absorbed from the gut into the circulatory system and delivered to the appropriate target location to achieve key health beneficial effects (Actis-Goretta et al., 2013). Several factors influence the rates of bioaccessibility of phytonutrients including the food matrix, processing and preservation methods (Thakur et al., 2020). For example, the bioaccessibility of carotenoids is relatively low varying from 5 to 30%, in comparison to other food phytochemicals (Kopec et al., 2018). Moreover, the low bioaccessibility of β-carotene from natural food sources has mainly been attributed to the limited release from food matrix, being bound to proteins, and its restricted release by the cell wall and low efficacy of micellization (Van Loo-Bouwman et al., 2014; Shilpa et al., 2020). Thermal treatment such as cooking, frying and pasteurisation can increase the bioaccessibility of carotenoids like beta-carotene in plant tissues

(Priyadarshani, 2017). Heat treatment can soften and disrupt the cell membranes and plant cell walls, and is known to break protein-carotenoid complexes (Erdman Jr et al., 1988). As a result, this stimulates the release of carotenoids and facilitates efficient digested by digestive enzymes during the digestive tract. Following the release from the food matrix, another major limiting factor is the poor solubilization of carotenoids in the digest. The addition of oil can promote the dispersion of carotenoids by dissolving them and by forming an emulsion in the digestive tract (Nagao et al., 2013). This process aids the incorporation of carotenoids in to mixed micelles and allows for better transport to intestinal epithelia. The bioaccessibility of carotenoids can therefore be improved from plant sources using different cooking methods (Hornero-Méndez et al., 2007; Qian et al., 2012). Reduced particle size has also been evident in several studies that can improve bioaccessibility of carotenoids by increasing the surface area available for enzymes to act (Hedren et al., 2002; Salvia-Trujillo et al., 2013; Lyu et al., 2021). The bioaccessibility of β -carotene from liquefied spinach was higher than that of the whole leaf or minced spinach (Castenmiller et al., 1999). Powder processing is also an effective way to produce small particles and easily cause the cell wall rupture and improve the β -carotene bioaccessibility (Lyu et al., 2021).

1.3. Research within David Gray's group

Our research group has been involved in studies focused on the utilisation of chloroplasts from green plants over a decade. Recent studies have shown interest in the use of chloroplast materials as food supplements, either for animal feed (Gedi et al., 2019) or as functional food ingredient (Torcello-Gómez et al., 2019). Spinach CRFs and whole leaf materials were fed to zebrafish as partial fish diet replacement compared with a standard fish diet, and the fish growth was monitored (Gedi et al., 2019). (!!! INVALID CITATION !!! Syamila et al. (2019)) has investigated a simple way to concentrate the liberated chloroplast materials, and the optimum way of dehydration (spray drying, freeze drying, conventional oven and vauum oven drying) of spinach juice/CRF with enhanced nutrients stability; In addition, Wattanakul et al. assessed the impacts of thermal processing (steam sterilisation, hot water blanching and pasteurisation) on the stability of nutrients (!!! INVALID CITATION !!! (Wattanakul et al., 2020; Wattanakul et al., 2022)). The research present in the current thesis extends our knowledge built on the previous findings. This study mainly tested the optimum processing technologies such as thermal (blanching and pasteurisation) and drying (spray and freeze drying), used to produce nutritionally rich powders derived from spinach leaves, on the material properties and digestibility of powders derived from spinach leaves. The bioaccessibility of essential lipophilic nutrients derived from chloroplast materials was investigated within a modified in vitro gastrointestinal human digestion model.

1.4. Aims, hypotheses and objectives.

Aims:

To establish the impact of thermal treatments, used in the production of a chloroplast-rich fraction (CRF) from spinach leaves, on the material properties, nutritional composition, and digestibility of the CRF material.

Hypothesis:

The hypotheses being tested were that a certain degree of thermal treatment and drying process can help to retain nutrients composition, restrict quality loss in the juice/CRF material and can increase their bioaccessibility by altering the material properties of the juice/CRF material at the macro and molecular level.

Objectives:

- Compare the physical properties of dried spinach juice through spray drying and freeze drying.
- 2. Compare the nutrients composition of freeze-dried spinach juice and CRF powders.
- Investigate the effect of pasteurisation and blanching of spinach biomass prior to freeze drying on the physical properties of the derived CRF material.
- 4. Determine the impact of heat pre-treatment of spinach juice on the bioaccessibility of lipophilic nutrients from the derived CRF material.
- 5. Monitor the digestion of galactolipids from spinach CRF material using an in-vitro system to simulate human digestion.

1.5. Thesis Structure & Clarification

This thesis consists of eight chapters starting with a general introduction, literature review and materials & methods chapters. It will then go on to five major results chapters, as outlined earlier. Chapter I introduces the background of the current research, raises the aims, hypotheses and objectives. Chapter II provides the explanations and justifications for the research methodology used, explaining the preparation steps and data analysis methods. Chapter III, IV and V compares the effects of different processing procedures such as drying (spray drying and freeze drying) and heat treatment (pasteurisation and blanching) on the physicochemical properties of spinach juice/CRF. Chapter VI presents the digestion of galactolipids and proteins from spinach juice/CRF by using a static *in vitro* digestion model. Chapter VII shows the bioaccessibility of lipophilic nutrients from spinach juice/CRF during *in vitro* digestion. The last chapter summarises the overall findings, conclusions and suggests the prospect of future work.

CHAPTER II

2. Materials and methods

2.1. Materials and chemicals

Fresh baby spinach was purchased from a local supermarket (Tesco, UK). The spinach was stored at 4 °C before using. Different batches of spinach were purchased during 2019-2022, and each chapter uses the same batch of spinach. All solvents (chloroform, methanol, acetonitrile, trimethylethylamine (TEA), ethyl acetate) were high-performance liquid chromatography (HPLC) grade and purchased from Fisher Scientific, UK. All chemicals: butylated hydroxyl toluene (BHT), sodium chloride, trimethyl sulfonium hydroxide (TMSH), triethyl amine (TEA) were purchased from Sigma-Aldrich (Merck, Darmstadt, Germany). All standards: β -carotene, α -tocopherol, and methylpentadacanoate were also purchased from Sigma Aldrich (Merck, Darmstadt, Germany), except for lutein which was purchased from Pan Reac AppliChem (Darmstadt, Germany). 10x Tris/Glycine/SDS Buffer, precision plus protein standards, 12% mini-PROTEAN TGX gels were purchased from Bio-Rad, California, USA. InstantBlue coomassie protein stain was purchased from ABCAM, Cambridge, UK. The galactcolipids standards including monogalactosyldiacylglycerol (MGDG) and digalactosyldiacylglycerol (DGDG) were purchased from Avanti, Polar lipids, Alabaster, USA. Linolenic acid and lipid standards (mono, di-, & triglyceride mix) were purchased from Sigma-Aldrich, Missouri, USA. High-performance thin layer chromatography (HPTLC) glass plates (10×20 cm) were purchased from Merck Life Science, Gillingham, UK. Other chemicals used were of analytical grade material.

2.2. Enzymes

The following enzymes were used in the current research. Pepsin was obtained from porcine gastric mucosa lyophilized powder, 2500 U/mg protein (P7012), porcine pancreatin extract (PPE) (P7545) and porcine bile extract (B8631), were all purchased from Sigma-Aldrich (Merck, Darmstadt, Germany). Rabbit gastric extract, 15 U/mg, was obtained from Lipolytech (Marseille, France). All enzymes were stored at -20°C before use.

2.3. Methods

2.3.1. Spinach chloroplasts processing

2.3.1.1. Blanching of spinach leaves

Fresh spinach leaves (200 g) were blanched in water at 100 °C for 30 s and then immediately immersed in ice cold water to cool samples to room temperature. Blanched leaves were the spun for 1 min using salad spinner to remove the excess water and then further being juiced. The blanching steps are illustrated in Figure 2.1.



Squeeze to get fresh/blanched spinach juice

Spin blanched leaves

Figure 2.1: Preparation steps of fresh/blanched spinach juice

2.3.1.2. Pasteurisation of spinach juice

Fresh spinach juice (500 mL) was pasteurised at mild (70 °C,15 s) and intense conditions (90 °C, 5 min) in a beaker covered by foil, the temperature was controlled using a water bath with a magnetic stirrer (500 rpm/min), (IKA RCT basic, England). The temperature of the juice was increased gradually and monitored using a thermometer, the average heating time was 4-6 minutes. The conditions were chosen to represent mild and intense pasteurisation respectively. Pasteurised spinach juice was then cooled in cold water (0-4 °C) immediately. All treatments were performed in the dark to avoid any possible interference from light. The chloroplast rich fraction (CRF) was collected by further centrifugation.

Fresh spinach juice/CRF was chosen as control, all the samples were stored at 4 °C

in the dark until required. The pasteurisation steps are shown in Figure 2.2.



Mild pasteurization 70°C15s; Intense pasteurization 90 °C5min (stir at 500rpm)



Pasteurized juice

Figure 2.2: Preparation steps of mild/intense pasteurised spinach juice

2.3.1.3. Preparation of spinach juice and chloroplast rich fraction (CRF)

Fresh and blanched spinach leaves were juiced using a twin gear juicer (Angel 7500, Busan, Korea), which separated the fibrous pulp from the nutrient-rich juice. The juice was then filtered through a 75 µm stainless steel mesh sieve.

The resultant juice was then centrifuged at 17,700 RCF (10,000 rpm), using a Beckmann Coulter JS-21M centrifuge with a JA-10 rotor (London, UK) for 10 min at 4 °C. The pellet containing the CRF was retained whilst the supernatant was centrifuged again under the same conditions to obtain another pellet and this was combined with the original pellet (Torcello-Gómez et al., 2019). This method utilises a juicer to mechanically press out fluid from spinach leaves, omitting the need to use harsh physical cell wall disruption steps which could potentially destroy intact chloroplast structures. The whole processing steps are shown in Figure 2.3.







Fresh juice

Heated juice

75 μm seive



Filtered juice



Freeze drying 7 days



Dried fresh/heated juice and CRF



Centrifugation

Vacuum packed (-20°C) 2

 $250 \mu m \ seive$

Grinding the sample



2.3.1.4. Spray drying

Spray drying (SD) was conducted using a spray dryer (Armfield FT80 Tall Form Spray Dryer, Ringwood, UK) using a modified method developed by Syamila et al. (2019). The drying condition was set at 1mm two-fluid atomizer nozzle, inlet air temperature 132 ± 2 °C, outlet temperature of 65 ± 5 °C, feed flow rate (rate of feeding) 420 mL/h, relative humidity range 5-11 %, compressed air pressure of 1-1.5 bar, respectively. The spray-dried samples were then stored in a vacuum-sealed aluminum bag at -20°C for further analysis (Figure 2.4). The principle of spray dryer involves converting a liquid or slurry into dried particles by atomizing the feed solution into droplets and then contacting with hot air.



Spray drying

spinach juice

Figure 2.4: Preparation steps of spray drying spinach juice

2.3.1.5. Freeze drying

Freeze drying (FD) was performed by freeze dryer (Super Modulyo, York, UK) with a modified method of Torcello-Gómez et al. (2019). The filtrated juice and pooled CRF (section 2.3.1.3) were weighed and frozen at -80 °C prior to freeze drying using the freeze dryer (Super Modulyo, York, UK) for 7 days. Freeze dried juice and CRF were then ground using a pestle and mortar, filtered with 250 µm sieve then stored in a vacuum-sealed foil bag at -20 °C for further analysis. The preparation steps are shown in Figure 2.3. The principle behind freeze dryer is to remove moisture from a substance (solution or a slurry) by freezing it and then sublimating the frozen water under vacuum. Main steps of freeze drying including freezing phase, sublimation phase and desorption phase. When the samples frozen below its freezing point, the water molecules become ice crystals form. The frozen ice is then sublimated to the vaper state in vacuum chamber and accelerated by heating. The residual moisture is further removed during desorption phase, which is a critical for preserving the stability of the product (Ward et al., 2021).

2.3.1.6. Representative freeze dried samples

The morphology and appearance differences of freeze dried spinach chloroplast samples can be observed in Figure 2.5. All the samples were after grinded and filtered with 250 μ m sieve, thus the particle size of these samples is lower than 250 μ m. This data has been confirmed by the particle size analyser as mentioned in section 2.3.3.3 (f).



Figure 2.5: Representative freeze dried spinach samples

2.3.2.Spinach juice characteristics during storage at 4 °C for 7 days

2.3.2.1. Total soluble solid content (Brix^o) and pH

The total soluble solid content of the spinach juice was determined at 20 °C using a digital handheld refractometer (Pocket PAL-α, Atago, Tokyo, Japan) and expressed as °Brix (which is based on the relationship between refractive index and the total soluble solid content of a pure aqueous sucrose solution). The refractometer measures the refractive index of a liquid sample to determine its sugar content, expressed as Brix°, 1 Brix° equivalent to 1 gram of sucrose per 100 grams of solution. By passing light through the sample, the angle at which the light is bent and determined as the refractive index index. The refraction index is proportional to the brix value, higher refractive index indicates higher sugar content of the sample.

The pH of the juice samples was determined by a pH meter (InoLab® pH 7110, German) at 20 °C. The pH meter works by measuring the concentration of hydrogen

ions (H⁺) in a solution to determine its acidity or alkalinity, expressed as the pH value. Two or more buffer with different pH values are used to calibrate the meter across the range of interest. The electrodes are immersed in the sample solution, and the voltage difference between the glass electrode and the reference electrode is measured. This voltage difference is converted into a pH value using the calibration curve obtained from the buffer solutions.

2.3.2.2. Determination of POD (peroxidase) reduction rate

POD activity was determined by UV-Vis spectrophotometry (Thermo Scientific, Genesys10S, US) at 20 °C using a modified method of Bahçeci et al. (2005). The reaction assay mixture was prepared with 500 μ L guaiacol, 500 μ L H₂O₂ diluted to 100 mL with distilled water. Spinach juice (0.1 mL) was then added to 2.9 mL of the assay mixture. The absorbance was measured at 420 nm for 1 min with 2 s interval. The enzyme activity was estimated from the linear portion of the curve of absorbance v/s time. One unit of POD activity was defined as $0.001A_{420}/min$. The activity of the samples was expressed as % POD reduction rate as given in the following equation (Eq. 2.1):

%POD reduction rate =
$$\left(1 - \frac{enzyme \ activity \ of \ pasteurized \ juice}{enzyme \ activity \ of \ fresh \ juice}\right) \times 100$$
 (Eq. 2.1)

The guaiacol-based assay for peroxidase activity is based on the following reaction:

 $2H_2O_2$ + guaiacol \rightarrow $2H_2O$ + guaiacol oxidation product

The oxidation of guaiacol results in the formation of a colored product, typically a brownish-red or yellow-brown compound. The intensity of the color formed is proportional to the peroxidase activity in the sample. The absorbance of the colored oxidation product is measured spectrophotometrically at 420 nm. The increase in absorbance over time reflects the rate of guaiacol oxidation, which is indicative of the peroxidase activity in the sample (Zhang et al., 2005).

2.3.3.Physicochemical properties of dried spinach chloroplasts

2.3.3.1. Macronutrients analysis: total lipids, proteins, ash and carbohydrates

Lipids were extracted from dried spinach chloroplasts, using a modified technique of Folch et al. (1957). Samples were dissolved with 1.2 mL (chloroform: methanol solvent at a ratio of 2:1 (v/v), then vortexed for 1 minute. To this 0.5 mL of a 0.9% NaCl solution was added to the mixture. The sample was then vortexed again and further centrifuged using a Thermo Jouan CR3i multifunction centrifuge, 3000 rpm for 10 min at 4 °C. The lowest phase containing the lipids was transferred into another tube and 1.2 mL of chloroform: methanol (2:1) added. The remaining two phases were then vortexed and centrifuged again. The lipid phase was removed, pooled with the original lipid layer, and centrifuged to separate the lipid from residual compounds. The lipid extract was filtrated through 0.45 μ m PTFE filter membrane, dried under a flow of nitrogen, and weighed to calculate the total lipid content. Dried lipid extracts were used to measure β -carotene, lutein, α -tocopherol, chlorophyll, and for fatty acid analyses. Total lipid content was calculated using the equation below (Eq. 2.2).

Total lipid (%) =
$$\frac{\text{dried lipid/g}}{\text{initial sample/g}} \times 100$$
 (Eq. 2.2)

Proteins were measured using the total nitrogen method (Lynch et al., 2020). In brief, samples (3 mg) were weighed in a tin capsule and two sulphanilamide standards were also weighed (3 mg). All samples along with the standards (Sulphanilamide) were run on an Organic Elemental Analysis Eager Experience (Flash 200, Thermo Fisher Scientific Inc, 2010). The Nitrogen values were automatically calculated in percentage. The element nitrogen value was converted to protein using 6.25 as a conversion factor.

In preparation for mineral analysis, an acid digestion was conducted by adding 1mL concentrated HNO₃ to 0.2 g of the powdered samples. This was heated at 115 °C for 4 h with occasionally stirring. Then, 1 mL H₂O₂ was added and digested for another 90 min at 115 °C. The flask was then removed and left to cool (in the fume cupboard) for several minutes. Small quantities of deionized water were then added to each flask and the solution was filtered through a Whatman No 42 (equivalent) filter paper and diluted up to 50 ml using deionized water. Multi-element analysis of the diluted solutions was undertaken by inductively coupled plasma mass spectrometry (ICP-MS) (Thermo-Fisher Scientific iCAP-Q; Thermo Fisher Scientific, UK). The instrument was run employing collision-cell technology with kinetic energy discrimination (CCT-KED) to remove polyatomic interferences; the collision cell gas used was helium (He). Samples were introduced using an auto sampler (Cetac ASX-520) incorporating an ASXpressTM rapid uptake module through a PEEK (polyether ether ketone) nebulizer (Burgener Mira Mist). Internal standards were introduced to the sample stream using a separate line via the ASX press unit and comprised of Sc (20 µg/L), Rh (10 µg/L), Ge (10 μ g/L) and Ir (5 μ g/L) in 2% trace analysis grade (Fisher Scientific, UK) HNO₃. External multi-element calibration standards (Claritas-PPT grade CLMS-2 from SPEX Certiprep Inc., Metuchen, NJ, USA) included As, Cd, Cu, Fe, Mn, Pb, Se and Zn, in the

range 0–100 μ g/L (0, 20, 40, 100 μ g/L). A bespoke external multi-element calibration solution (PlasmaCAL, SCP Science) was used to create Ca, Mg, Na and K standards in the range of 0 – 30 μ g/L. Phosphorus calibration utilized an in-house KH2PO4 solution standard (10 μ g/L). Sample processing was undertaken using QtegraTM software (Thermo-Fisher Scientific), utilizing external cross-calibration between pulse-counting and analogue detector modes as required. The ash contents of spinach chloroplasts were determined from the total amount of minerals presented in samples.

The percentage of total carbohydrates of spinach chloroplasts were estimated using the following equation (Eq. 2.3).

Carbohydrate (%) =
$$100 - (lipid + protein + ash)$$
 (Eq. 2.3)

2.3.3.2. Micronutrients analysis: chlorophylls, β-carotene, lutein, vitamin E (α-tocopherol) and fatty acid composition

The fatty acids of a lipid extract were esterified to fatty acid methyl esters (FAMEs) and analysed using gas chromatography-mass spectrometry (GC-MS) (Thermo Scientific, DSQ), (Gedi et al., 2017). A dried lipid extract was dissolved in 2 ml of chloroform containing 0.1% butylated hydroxyl toluene (BHT). To 1 ml of dissolved lipid, 100 μ l of internal standard, methyl pentadecanoate (10 mg/ml), 200 μ l of trimethylsulfonium hydroxide (TMSH) were added and filtered (0.45 μ m, 13 mm) into amber GC-MS vials. Esterification was achieved through the addition TMSH and the solution was then left, to ensure complete conversion, for a minimum of 10 minutes before GC-MS analysis. The sample was injected at a volume of 10 μ l onto a Phenomenex Zebron ZB-FFAP (30 m × 0.22 mm) column using a vaporising injector

with a split flow of 50 ml/min. The oven temperature was maintained at 120 °C for 1 min and then increased to 250 °C at a ramp of 5°/min. The final temperature of 250 °C was held for 2 minutes. Detection was conducted using a mass spectrometer and identification of individual fatty acids was achieved through mass spectrum library and comparison of retention times to FAME standards. The percentage content of each fatty acid was calculated and the concentrations (mg/mL) are determined using the response factor of the internal standard.

GC-MS combines two components, Gas Chromatography (GC) and Mass Spectrometry (MS) to identify and quantify of compounds from complex mixtures. In principle, the targets of interest are volatilized and carried with mobile phase (inert gas normally helium or nitrogen gas) then interact with the stationary phase (often a coated capillary column). As the sample travels through the column, different compounds interact differently with the stationary phase based upon their chemical properties such as boiling point, polarity, and affinity. And thus, being separated by different time that eluted from the column and transferred to MS. The individual compounds are ionised in the MS and characterised by their mass-to-charge ratio (m/z). The identification can be achieved by comparing their retention times and mass spectra with known standards or databases. Quantification can be obtained by correlating the peak areas in the chromatogram with known concentrations or using internal standards (Hussain et al., 2014).

The chlorophyll content was determined using a UV-Vis spectrophotometry (Thermo Scientific, Genesys10S, US). The lipids extract was diluted with acetone to be
a factor of 1:1000. The sample solution was measured at two different wavelengths: 662 nm (chlorophyll a) and 645 nm (chlorophyll b). Pigments concentrations (μ g/mL) were calculated as previously described using equation 2.4 and 2.5 (Lichtenthaler et al., 2001). The principle behind UV-Vis spectroscopy involves the interaction of light with matter, particularly molecules, which results in the absorption of specific wavelengths of light.

Chlorophyll a =
$$(11.24 \times A661.6) - (2.04 \times A644.8)$$
 (Eq. 2.4)
Chlorophyll b = $(20.13 \times A644.8) - (4.19 \times A661.6)$ (Eq. 2.5)

The
$$\beta$$
-carotene and lutein content of dried chloroplast material was analysed by

high performance liquid chromatography (HPLC) with photodiode array (PDA) detection (Agilent 1100, Germany) (Wattanakul et al., 2022). B-carotene was detected at 454 nm. The flow rate of the mobile phase (Acetonitrile: Methanol: Ethyl acetate with 0.05% triethyl amine) was set at 0.5 ml/min. Two gradient mobile phases were applied from 95:5:0 to 60:20:20 in 20 min, maintaining this proportion until the end of the run. Re-equilibration took 15 min. Lipid extract of dried spinach juice and CRF was injected at a volume of 10 μ l through a Security Guard guard-column and separated on a Waters Nova-Pak C18 (4 μ m, 3.9 × 150 mm) column with the column temperature set at 22°C.The concentration of β-carotene and lutein were determined using a linear equation created using a calibration curve produced from a range of external (β-carotene, lutein) standards dissolved in acetone containing 0.1% butylated hydroxytoluene (BHT).

The Vitamin E (a-tocopherol) content of dried chloroplast material was

determined using HPLC detection using an Agilent 1100 series unit with a fluorescence detector (Jasco, FP-920, Japan) as described (Wattanakul et al., 2022). The flow rate of the mobile phase was set at 0.8 ml/min starting with 100% solvent A (Acetonitrile: Methanol: Isopropanol: 1%Acetic acid solution with a ratio of 45:45:5:5, v/v). Lipid extracts of dried spinach juice and CRF were separated following the injection of 10 µl of sample extract a using a Zorbax RX-C18 5 µm (250 × 4.6 mm) column set at 20°C. After 6 min of solvent A running isocratically the mobile phase changed linearly to 100% solvent B (Acetonitrile: Methanol: Isopropanol with a ratio of 25:70:5, v/v) over the following 10 min. This was held for 12 min before being returned to the initial conditions with a total run time of 36 min. Detection of α -tocopherol, at approximately 23.2 min, can be achieved at excitation and emission wavelengths of 298 and 328 nm respectively and quantitation achieved using the linear formula produced from a calibration curve of α -tocopherol standards (4-100 µg/mL) in methanol containing 0.1% BHT.

The injected samples are carried by mobile phase (mixture of solvents) through the column and interacts with the stationary phase to separate the components of the mixture. Separation in HPLC is based on the differential partitioning of analytes between the mobile phase and the stationary phase. Compounds with different affinities for the stationary phase will elute from the column at different rates, leading to separation. HPLC is a technique used for separation, identification and quantification of individual components from mixture, and widely applied in various industries for its high sensitivity and resolution (Ali, 2022).

2.3.3.3. Physical characterization

a) Moisture content and water activity

The moisture content of fresh CRF and dried chloroplasts powders produced by spray drying and freeze drying was determined according to the method of Syamila et al. (2019). Dried powder samples (0.5 g) were weighed in a tared foil tray and then dried in oven at 105 °C The drying, cooling, weighing processes were repeated until constant weight was obtained. Water activity was measured at 25 °C using Aqualab Dew Point Water Activity Meter 4TE (Decagon Devices Inc., Pullman, WA, USA). Samples were triplicate and the mean values \pm standard deviation were recorded. The moisture content can be calculate based on the Eq. 2.6.

Moisture content (%) =
$$\frac{wet weight - dry weight}{wet weight} \times 100$$
 (Eq. 2.6)

b) Solubility at body temperature

Solubility was estimated according to the method proposed by Anderson et al. (Anderson et al., 1970), with some modifications. 0.5 g of samples was added to 14 mL of ultrapure water and vortexed for 1 min, then heated at 37 °C incubator which was designed to simulate the human body temperature and shake at 250 rpm for 30 min. The suspension was then transferred to a tube and centrifuged at 3500 rpm at 4 °C for 20 min. The supernatant was transferred completely to a pre-weighed aluminium can and dried at 105 °C for 4 h. The dried soluble solid after drying was then weighed and used to calculate the solubility as a percentage, according to the following equation (Eq. 2.7).

$$Solubility(\%) = \frac{\text{dried supernatant weight(g)}}{\text{initial sample weight (g)}} \times 100 \quad (\text{Eq. 2.7})$$

c) Dispersibility

Dispersibility was determined following the method of Jinapong et al. (2008) with slight modification. In brief, 0.5 g dried chloroplast powder/fresh CRF was added to 10 mL distilled water and stirred vigorously for 25 s. Then sieved by passing through 150 μ m sieve. A portion of the filtrate (1 mL) was transferred to a pre-weighed aluminum can and then dried at 105 °C for 4 h. The dried soluble solid was then weighted and the dispersibility calculated as a percentage.

d) Light Microscope and scanning electron microscopy (SEM)

The morphology of spinach juice/CRF powders was observed under a light microscope (Nikon Microscope Eclipse E400, Nikon Corporation, UK). The dried powdered samples were hydrated in water before loading on the plates for observation. All light micrographs were taken with polarised light technique equipped with 20X or 40X objective lens.

In addition, scanning electron microscopy (JEOL JSM-6060LV Variable Pressure Scanning Electron Microscope (Jeol. (UK)Ltd, Hertfordshire, UK) was used to observe physical changes of the chloroplast structure and the impacts of differing drying methods of organelle integrity. Platinum coating was carried out and analyses were performed with 10 kV accelerating voltage and under a pressure of 40 Pa. SEM was performed/operated with 1000x and 2000× magnification. Meanwhile, the Cryo-SEM was operated with a magnification of 1000× and 2000× for obtaining the micrographs of fresh CRF. Samples were prepared by high pressure freezing with a Leica ICE and transferred to the SEM with a Quorum Technologies PolarPrep 2000 cryo transfer system. After sublimation at -90 °C for approx. 6 minutes the sample was gold coated before transferring to the SEM chamber for imaging. The sample was maintained at, or below - 160 °C during imaging. Samples were imaged on a JEOL 6060LV SEM operating at 10 kV.

SEM images represent the morphology of a sample and can also reconstruct quasithree-dimensional views of the sample surface. In SEM, the focused electron beam scans the surface of the sample, and interact with atoms in the sample, producing various types of signals including secondary electrons and backscattered electrons that contain information about the surface topography and composition of the sample (Kannan, 2018).

e) Transmission electron microscope (TEM)

The morphology of fresh and blanched spinach leaves was analysed using a Tecnai Bio-TWIN T12 Biotwin transmission electron microscope (TEM) (FEI Company, Eindhoven, The Netherlands) and run at an accelerated voltage of 120 kV. Images were captured using a MegaView SIS camera.

The fresh and blanched spinach leaves were cut into small pieces around 1-3 mm² then immerged into 600 μ L 25% EM Glutaraldehyde and 2.5 mL 0.2M Cacodylate buffer and 1.9 mL distilled water overnight at 4 °C. The supernatant was discarded after fixation and then the fixed samples were washed twice and filled up with the 0.1 M Cacodylate buffer. Post fixed samples in 1% osmium tetroxide for 2 h, then washed

twice in distilled water 5 min each time. Then, dehydrated in graded ethanol series in the following sequence: $2 \times 10 \text{ min } 50\%$ ethanol, $2 \times 10 \text{ min } 70\%$ ethanol, $2 \times 10 \text{ min } 90\%$ ethanol, $3 \times 20 \text{ min } 100\%$ ethanol, $2 \times 10 \text{ min } 100\%$ propylene oxide (propox). The samples were then infiltrated with resin and immersed in 3:1 propox:resin for 2 h, and then mixed with 1:1 propox:resin overnight at room temperature. Then, the samples were put into the 100% resin 2.5 h for 3 separate times and embedded in an appropriate plastic mould and polymerised. The polymerised spinach leaves were cut into pieces before loading on TEM, and imaging at different magnifications.

TEM creates an image using a broad beam of electrons that transmitted through the ultrathin sample. As a result, TEM offers valuable information on the inner structure of the sample, such as morphology, crystal structure and stress state information (Rao et al., 2010).

f) Particle size measurement

The particle size distribution and mean diameter of fresh CRF and dried chloroplasts powders were determined using LS-13-230 Laser Diffraction Particle Size Analyser (Beckman Coulter, United States). The mean diameter of samples was based on the mean diameter of a sphere of the same volume, the Brouckere diameter (D_{4,3}). Fresh CRF was analysed by wet method, with dispersion in RO water. Obscuration values ranged from 1% to 8%, and the refractive index used for dispersed phase was 1.33. Dried powders were analysed by solid mode, with a dry dispersion unit. Obscuration values around 1-6%. The volume-weighted mean diameter D_{4,3} was calculated as follows:

$$D_{4,3} = \frac{\sum n_i D_i^4}{\sum n_i D_i^3}$$
 (Eq. 2.8)

Where D_i is the mean particle diameter and n_i is the number of particles.

The span was calculated according to the equation as follows:

$$span = \frac{d_{90} - d_{10}}{d_{50}}$$
 (Eq. 2.9)

Where d90, d50, and d10 are the average volume diameters equal to 90%, 50%, 10% of the cumulative volume, respectively.

The laser diffraction particle size analyser works on the principle that when the laser beam passes through the sample, it interacts with the particles suspended in the medium. The laser light is scattered in different directions by the particles, and the angle of light scattering is inversely proportional to particle size. The angular scattering intensity is subsequently assessed to determine the size of the particles accountable for generating the scattering pattern, with the Mie theory of light scattering (Ryżak et al., 2011).

2.3.4. Static *in vitro* digestion model

An *in vitro* static digestion model was chosen for this study because it is time saving and cost-effective without ethical issues. The static *in vitro* digestion model was based on the method proposed by Minekus and Brodkorb et al. (Minekus et al., 2014; Brodkorb et al., 2019) with some modifications, to encompass oral, gastric, and intestine phases of digestion. The spinach juice/CRF powders (1.5 g) were dissolved in 3.5 mL of ultra-pure water as the starting mixture solution. To the mixture, 5 mL of simulated salivary fluid (SSF) without α-amylase was added and incubated for 2 min at pH 7 in oral phase. For the gastric phase, 10 mL simulated gastric fluid (SGF) was added, along with the selective enzymes, rabbit gastric extracts (RGE) (20.4 U/ml) and pepsin (2000 U/ml). The pH was adjusted to 5 and the solution and then incubated for 2 h. For the intestine phase, the solution was mixed with simulated intestine fluid (SIF) (1:1), porcine pancreatic extracts (PPE) with final concentration of pancreatic lipase at 2000 U/ml and trypsin around 100 U/ml, in addition, 10 mM porcine bile extracts were also added, then pH of solution was adjusted to 6.5 and incubated for another 2 h. The whole digestion was performed inside an incubator at 37 °C while shaking with the rotator at 20 rpm. The pH of gastric and intestine was set at 5 and 6.5 for mimicking the GI conditions at half gastric emptying time during a meal (Borgstrom et al., 1957; Sams, 2016). The rotating during digestion was in order to gently mix the solution and enzymes. Samples (1 mL) from the digestion mixture were collected at the end of oral, gastric and intestine phase. The samples (1 mL) after oral and gastric phase were mixed with 1 mL NaHCO₃ to stop the enzymatic reaction, while for the samples at the end of intestine phase, 1M Orlistat was added to the digesta to stop the enzyme activity. The samples were all kept at -80 °C until further lipid extraction as Section 2.3.4.2.

The preparation of stock solutions for simulated digestion fluids is shown in Table 2.1.

Table 2.1: Volume of electrolyte stock solutions of digestion fluids for a volume of400 mL diluted with water ($1.25 \times$ concentrations). The stock solution will be furtherdiluted to the final volume of 500 mL ($1 \times$ concentrations) for each simulated digestivefluid.

			SSF		SGF		SIF	
	Stock		(pH 7)		(pH 3)		(pH 7)	
Salt solution			Vol.	T ' 1	Vol.	T ' 1	Vol.	T' 1
			of	of	of C	Final	of	Final
			stock	Conc.	stock	Conc.	stock	Conc.
			for	1n	for	1n	for	1N
			0.4L	SSF	0.4L	SGF	0.4L	SIF
	(g/L)	(M)	mL	mM	mL	mM	mL	mM
KCl	37.3	0.5	15.1	15.1	6.9	6.9	6.8	6.8
KH ₂ PO ₄	68	0.5	3.7	3.7	0.9	0.9	0.8	0.8
NaCl ^a	58.4	1	6.8	13.6	12.5	25	42.5	85
NaCl	117	2	-	-	11.8	47.2	9.6	38.4
MgCl ₂ (H ₂ O) ₆	30.5	0.15	0.5	0.15	0.4	0.12	1.1	0.33
(NH ₄) ₂ CO ₃	48	0.5	0.06	0.06	0.5	0.5	-	-
HCl		6	0.09	1.1	1.3	15.6	0.7	8.4
CaCl ₂ (H ₂ O) ₂	44.1	0.3	0.025	1.5	0.005	0.15	0.04	0.6

a, NaHCO₃ was replaced by NaCl in the electrolyte solutions to avoid unwanted pH drift. Simulated salivary fluids (SSF). Simulated gastric fluids (SGF). Simulated intestinal fluids (SIF). All simulated fluids stock solutions were prepared at x1.25 concentration, and stored at -20 °C.



Figure 2.1: Flow chart to measure nutrient bioaccessibility in *in vitro* digestion model.

2.3.4.1. Study of bioaccessibility

The bioaccessibility of carotenoids (β -carotene and lutein) and vitamin E (α -tocopherol) was measured in the current work. The digesta (5 mL), a mixture solution at the end of intestine digestion was centrifuged at 5000 rpm for 1 h at 4 °C, and the supernatant was filtered with a 0.2 µm PTFE filter membrane to obtain the micelle phase. Both the digesta and micelle was further transferred to a lipid extraction vessel (Section 2.3.9.2) and used for nutrients analysis.

The micellarisation rate and bioaccessibility (%) was calculated based on the research conducted by Chen et al. (2019) with the following equations (Eq. 2.10, Eq. 2.11):

Micellarisation rate $\% = (C \text{ micelle/C digesta}) \times 100 (Eq. 2.10)$

Bioaccessibility % = (C micelle/C initial) × 100 (Eq. 2.11)

Where C_{micelle} and C_{digesta} is the concentration of the nutrient in micelle and digesta fraction, C_{initial} is the concentration of the nutrient in CRF before digestion.

2.3.4.2. Lipid extraction for digested sample

Lipid extraction was performed using the method of Folch et al. (1957) (Section 2.3.8.1) with slightly modification. Samples (1 mL for HPTLC analysis or 5 mL digesta and micelle for HPLC analysis) were mixed with 1mL or 2 mL of 150 mM NaCl solution, and 1.2 mL or 5 mL of a mixture of chloroform and methanol (2:1,v/v), respectively. The mixture was then vortexed for 1 min and centrifuged (Thermo Electron Corporation, Jounan CR3i multifunction, US) at 3000 rpm for 10 min at 4 °C.

This processing step allowed for the phase separation of the samples. The lowest phase which contained the lipids were transferred and collected and a further 1.2 mL or 5 mL of a mixture of chloroform and methanol (2:1) added. The remaining aqueous phase and the mixture was then vortexed and centrifuged again twice more. The lipid extracts were pooled and dried under gas nitrogen until constant weight and stored at -20 °C until further analysis.

2.3.4.3. High performance thin layer chromatography (HPTLC) analysis

HPTLC technique was used in this study for the analysis of both polar and nonpolar lipids of spinach juice/CRF throughout digestion. This analysis can be divided into two separate tests: Polar lipids (Galactolipids: monogalactosyldiacylglycerol (MGDG) and digalactosyldiacylglycerol (DGDG)) and non-polar lipids (monoacylglycerol (MAG), diacylglycerol (DAG), triacylglycerol (TAG) and free fatty acid (FFA)). Separation of individual class of lipids from each other is based on the competition of the solute and the mobile phase for binding places on the stationary phase (normally silica gel) due to polarity differences of the analytes (Hussain et al., 2019). The significant advantage of HPTLC for the analysis of lipids is that lipids can be easily visualised by staining.

The analysed lipids were first extracted from a 1ml sample obtained at the end of the gastric phase (2 h) and intestine phase (4 h) of the digestion process (shown in 2.3.9.2). Then, both the extracted lipids and standards (MGDG, DGDG, mix lipids and α -linolenic acid) were spotted onto a thin-layer silica plate using a Linomat 5 (CAMAG, Swiss) equipped with a 100 µL Hamilton syringe. The plates were then developed in a

TLC developing chamber contained separation mobile phase and dried at room temperature in a fume hood (10 min for polar lipids, 5 min for non-polar lipids).

For polar lipids (galactolipids), the separation mobile phase consists of chloroform: methanol: water (47.5:10:1.25, v/v/v); Then, the polar lipids plate dipped in a thymol solution for derivatisation/staining while avoiding the interference of pigments, like chlorophylls during the densitometric analysis of the plates. The thymol solution was prepared by dissolving 1g of thymol in 190 mL ethanol and then put in the ice bath and gradually added 10 mL 96% sulphuric acid since the mixing reaction is highly exothermic. After staining, the plates were dried and transferred into the oven at 110 °C for 4 min until the bands turn brown-pink and the background turns pink.

For non-polar lipids (MAG, DAG, TAG and FFA), the mobile phase consisting of heptane: diethyl ether: acetic acid (55:45:1, v/v/v). And the non-polar lipids plates dipped in phosphoric acid-copper acetate solution for acyl chain staining. The dipping solution was prepared by mixing a saturated copper acetate solution with 85% phosphoric acid (1:1, v/v). The plates were left in fume hood until dry and transferred in the oven at 180 °C for 6 min.

Densitometry analysis of stained lipids on the TLC plates was carried out by the TLC visualizer 2 (CAMAG) and 'visionCATS' software. Lipid bands were scanned at 366 nm for thymol staining and at 500 nm for copper acetate-phosphoric acid staining, with a 0.5 x 7 mm slit and a speed of 2.5 cm/min. Slit conditions were selected accordingly to band size. The slit should always cover the whole band size. Quantities of the lipids on the TLC plates were estimated from the linear standard curves

established with the pure lipid standards (MGDG, DGDG, mix lipids and α -linolenic acid (C18:3)).

In the field of lipids, HPTLC is typically used for separations, identification of the individual lipids and quantifications. Three main steps are included: sample application, chromatogram development and chromatogram evaluation (Rani et al., 2015). The separation principle relies on adsorption. One or more compounds are applied as spots onto a thin layer of adsorbent material coated on a chromatographic plate. Capillary action, rather than gravitational force, propels the mobile phase (developing solvent) through the system. The compounds migrate based on their affinities towards the adsorbent. Components with greater affinity towards the stationary phase travel slower, while those with lesser affinity travel faster. There are some advantages of HPTLC that makes it widely accepted for lipid analysis in various fields such as lipidomics, food science, pharmaceuticals, and clinical research. Firstly, High quality separations are achievable. HPTLC offers high-resolution separation, allowing for the effective separation of lipid components. Lipid samples often contain multiple lipid classes and subclasses, each with various molecular structures. HPTLC's ability to separate these components efficiently is crucial for accurate analysis. Also, HPTLC can analyse a wide range of lipid classes, including fatty acids, phospholipids, glycolipids, sterols, and triglycerides. This versatility makes it suitable for comprehensive lipid profiling studies (Fuchs et al., 2011).

2.3.4.4. Sodium dodecyl sulfate–polyacrylamide gel electrophoresis (SDS-PAGE)

Protein digestion was characterized using SDS-PAGE on a Mini-PROTEAN Tetra System (Bio-Rad, USA) with 12% Mini-PROTEAN TGX gels. The spinach chloroplasts powder was dissolved in ultra-pure water with the final protein concentration 2 mg/mL, and then mixed with sample buffer (1:1, v/v). The sample buffer was prepared by mixing 2×laemmli with β-Mercaptoethanol (19:1, v/v) the mixture is then heated at 95 °C for 5 min to denature the protein, followed by centrifugation at 3,000 rpm for 5 min. 20 µL of the supernatant were loaded into separated wells on the gel, under this condition, the different preparations of spinach chloroplasts were loaded to constant protein around 20 µg/well, and the precision plus protein standards was loaded 5µg in the first well in order to identify the proteins. SDS gel was ran for 15 min under 80 V and then for 60 min under 160 V in Tris/Glycine/SDS Buffer. Gels were subsequently stained with InstantBlue Coomassie. ChemiDocTM MP imaging system (Bio-Rad, USA) was used to scan and detect the bands densities and the final analysis was performed by using the Image Lab Software.

SDS-PAGE works on the principle that the denatured proteins can be separated by their molecular weight. Sodium Dodecyl Sulfate (SDS) as a strong ionic detergent along with reducing agent can denaturate proteins by heating, and leading to the unfolding protein structures. The denatured SDS-bound protein with the negative charge migrates towards the electrode with the positive charge when placed in an electric field. The separation of the charged molecules depends on their molecular weight. The smaller molecules migrate faster due to less resistance during electrophoresis. The proteins can be visualised by specific staining technique, the size of protein can be estimated by comparing with the migration of known molecular weight protein standards (Roy et al., 2014).

2.4. Statistical analysis

All experiments were carried out in triplicate, and the results were expressed as mean \pm standard deviations. The statistical analysis was performed by IBM SPSS 25. The comparison of data more than two groups was analysed by one-way ANOVA followed by a post-hoc Tukey test, the statistical significance was set to the p<0.05 level, which means a difference of means at p<0.05 were considered significant.

Chapter III

3. Effects of drying methods on physicochemical properties of spinach juice/CRF

3.1. Introduction

Spinach (*Spinacia oleracea* L.) is a widely cultivated leafy green vegetable grown all over the world, with around 30 million tonnes of spinach produced worldwide in 2020/2021. Spinach leaves are readily consumed for fresh usage or after processing such as canning, freezing or dehydration. However, current production methods results in approximately 25% waste production even though the waste tissues are abundant in high value chemicals like lutein and chlorophyll (Derrien et al., 2017). Despite the high water content, spinach has a high content of macronutrients, such as lipids, proteins, carbohydrates, as well as micronutrients such as carotenoids (beta-carotene and lutein), chlorophylls (chlorophyll a and b), vitamins (vitamin A, C and E), minerals, phenolic compounds (including flavonoids, phenolic acids and others) (Bunea et al., 2008).

The chloroplast is an organelle abundant in plant leaves; Recent research has shown that chloroplasts contain the majority of plant nutrients with 75–80% of the total leaf nitrogen being found within chloroplastic proteins in C3 plants (most vegetables and trees, and several major cereals) (Ishida et al., 2014). Moreover, the chloroplastrich fraction (CRF) from green materials, such as spinach, kales, nettles and grass show a larger amount of macro and micronutrients than whole leaf materials (WLM) (Gedi et al., 2017). Spinach is known to have a number of health benefits including anticancer (Abu Al-Qumboz & Abu-Naser, 2019), anti-obesity (Lasya, 2022), hypoglycemic (Park et al., 2012), and hypolipidemic functionalities (Roberts et al., 2016). Therefore, many researchers are interested in developing approaches needed to maintain nutrients levels in spinach products to aid the development of more effective functional food ingredients. Sadly, fresh spinach can only store 2-4 days at room temperature, and nutrient profiles are diminished particularly notable are losses in folate and carotenoid content (Rambuda et al., 2015). It is well known that nutrient contents start to decrease after harvesting due to the processing that expose foods to high levels of heat, light or oxygen and continues during storage. Thus, the development of novel processing methods that extend shelf life is important in the fresh fruit and vegetable industry. One such method is to produce spinach powder as an alternative way to preserve the main bioactive compounds of spinach, and to ensure product stability, while also reducing transportation cost (Daza et al., 2016). Spray drying and freeze drying are used commercially to dry materials, for this reason this we tested the impact of these drying methods on the physico-chemical properties, as well the nutritional content, of the resulting materials.

The current chapter focuses on the characterization of spray dried juice (SJ), freeze dried juice (FJ) and freeze dried CRF (FC) derived from spinach. In all samples, nutrient compositions and physical properties were studied to determine the optimum processing procedure used for spinach chloroplast materials production.

3.2. Sample preparation summary



Figure 3.1: Spinach juice/CRF powders preparation steps.

3.3. Results and discussion

3.3.1.Nutrients content

3.3.1.1. Macronutrients content of dried spinach juice/CRF

The macronutrients content of various chloroplast fractions is shown in Table 3.1. The content of macronutrients in dried spinach juice were comparable following both drying methods; slightly higher protein and lipid content were seen in spray dried juice. The major contribution of dried spinach juice was protein (35.5-38.3%), and carbohydrates (33.2-37.1%). While the main contribution of freeze-dried spinach CRF (FC) were proteins (44.6%) and lipids (26.9%), and these nutrients were significantly higher than in both spray and freeze dried juice (SJ and FJ) (Table 3.1).

Our results correlate well with those reported by (Gedi et al., 2017). This research reported that the protein content in CRF and whole leaf material of spinach were 42.6%, 35.3% respectively, and showed close agreement with our data. Similarly, lipid contents were 36.9% and 19.3% respectively, higher than the results observed in this study (Table 3.1). These differences could be mainly due to the batch of spinach used during harvesting rather than preparation methods. Indeed, Gedi et al. (2017) prepared spinach CRF by blending spinach leaves with 0.3 M sucrose solution (ratio of 1:6 (w/v)), however, in our work spinach CRF were prepared with a slow-screw twin-gear juicer. While, no significant difference of nutrients content was observed in same batch of spinach CRF between these preparations as studied by Mansor (2019).

Overall, both drying methods can retain the macronutrients in spinach juice at similar level. Spinach CRF contained higher concentrations of proteins and lipids than juice, due to the fact that CRF samples were more concentrated with thylakoid membranes which are mainly composed of proteins and lipids.

 Table 3.1: Macronutrients content of spinach juice/CRF collected by freeze drying and

 spray drying method

%DW	SJ	FJ	FC
Protein	38.3±0.6b	35.5±0.2c	44.6±0.7a
Lipid	13.0±1.9b	9.7±1.2c	26.9±2.0a
Ash	15.5±0.3b	17.7±0.7a	7.6±0.3c
Carbohydrates	33.2	37.1	20.9

Data was analysed by one-way ANOVA according to Tukey's test (p < 0.05) to access the differences of the samples. Means (n=3) designated with different letters in the same row are significantly different (p<0.05). Carbohydrates were estimated by difference, 100-(proteins+lipids+ash)/%. *Abbreviations: SJ* Spray dried juice, *FJ* Freeze dried juice, *FC* Freeze dried CRF

3.3.1.2. Fatty acid composition of dried spinach juice/CRF

The effect of different drying methods on fatty acid composition of chloroplasts from spinach is presented in Figure 3.2 (a). Seven fatty acids were detected including palmitic acid (C16:0), palmitoleic acid (C16:1), hexadecatrienoic acid (C16:3), stearic acid (C18:0), oleic acid (C18:1), linoleic acid (C18:2), and alpha-linolenic acid (C18:3). And among which polyunsaturated alpha-linolenic acid (C18:3) (ALA) exhibited the highest amounts in both spinach CRF and juice, followed by palmitic (C16:0) (PA) and linoleic acid (C18:2) (LA). The predominant ALA in chloroplasts is because it is a key component of the thylakoid membranes (Wang et al., 2012a). These fatty acids were also reported as the most predominant in spinach, kale, nettle and grass (Gedi et al., 2017).



(a)

75



(b)

Figure 3.2: Fatty acid composition of spray-dried juice, freeze-dried juice and freezedried CRF Results are expressed on a dry weight (a) and total lipid (b) basis and as means \pm SD (n=3). Results are analysed using post-hoc analysis of variance (ANOVA) and according to a Tukey test with statistical significance at p<0.05, a>b. *Abbreviations: SJ* Spray dried juice, *FJ* Freeze dried juice, *FC* Freeze dried CRF

Comparing the SJ and FJ fractions, there were no significant differences between the different drying methods on fatty acids composition except for hexadecatrienoic acid (C16:3) and alpha-linolenic acid (C18:3). This finding indicated that each drying method retained the natural profile of fatty acids in chloroplast preparations. In addition, fatty acids were more concentrated in the CRF as compared to spinach juice, and critically the total fatty acid content in freeze dried CRF (FC) was 61.8 mg/g DW, being 3 times that of freeze dried juice (FJ) which was 20.2 mg/g DW. These differences could be due to the fact that FJ contains higher ash (15.5%) and carbohydrates (39.3%) levels than that of lipids. Similar results were found by (Gedi et al., 2017), who also quantified the same fatty acids in spinach preparations and who noted ALA as the most abundant fatty acid. Higher amounts were present in freeze dried CRF compared with freeze dried powdered whole leaf materials (WLM) on a dry mass basis.

As shown in Figure 3.2 (b), the fatty acids contents (mg/g lipid) were not significantly influenced by drying methods. The levels of alpha-linolenic acid were higher in the CRF (131.6 mg/g lipid) than in juice ([SJ]:110.6 and [FJ]:107.7 mg/g lipid), dry weight material. The chloroplasts contain high levels of lipid so expressing fatty acid per unit mass of lipid tends to result in a convergence of values for all samples.

As reported in previous research, vegetable-based fatty acids are a mixture of unsaturated fatty acids, such as oleic, linoleic and linolenic acid, which is comparable to previous studies (Bajpai, 2014). Alpha-linolenic acid (ALA) is an essential omega-3 fatty acid, which is unable to be synthesised in the human body and must be ingested through diet and is precursor of eicosapentaenoic acid (EPA) and docosahexaenoic acid (DHA). It has been reported to have the function of reducing risk of cancer, cardiovascular disease, inflammation, and various cognitive disorders (MacLean et al., 2006; Rizos et al., 2012). Therefore, the high content of ALA observed in chloroplasts of spinach can be seen as a source of potential nutritional vegetarian ingredient which provides EPA and DHA instead of fish oil.

3.3.1.3. Pro-vitamin A, lutein and vitamin E of dried spinach juice/CRF

The content of pro-vitamin A (β -carotene), lutein and vitamin E (α -tocopherol) from dried spinach juice and CRF are presented in Figure 3.3 (a) and (b). In this study,

 β -carotene and lutein were the most abundant forms of two categorized carotenoids (xanthophylls and carotenes). α -Tocopherol as the major form of vitamin E existing in nature as well as the most biologically active were considered (Van Bennekum et al., 2005).





(b)

Figure 3.3: β -carotene, lutein and alpha-tocopherol content of spray-dried juice, freezedried juice and freeze-dried CRF.

Results are expressed as means \pm SD (n = 3). Results are analysed using post-hoc analysis of variance (ANOVA) and according to a Tukey test with statistical significance at p<0.05, a>b. *Abbreviations: SJ* Spray dried juice, *FJ* Freeze dried juice, *FC* Freeze dried CRF. β -car β -carotene, *Lut* lutein, α -Toc alpha-tocopherol.

It is known that the majority of leaf pigments (carotenoids) and vitamin E compounds (tocopherols) are found in chloroplast preparations. It is therefore not surprising that the relative concentration of these lipophilic molecules increased in CRF preparations as this action would concentrate the chloroplasts (CRF compared with dried juice). When the results are expressed relative to lipid content it appears that the freeze-dried CRF retains a higher level of carotenoids and tocopherol than when the juice is dried directly. If one assumes that there is no significant loss of the major lipids during drying (Figure 3.2 b), then it appears that the juicing step caused the loss of a measurable proportion of non-chloroplast membrane lipids in dried juice samples compared with CRF. The extent of lutein and tocopherol losses appears to be less if freeze-drying is used by comparing with spray drying. According to the study of Bergquist et al. (2006), baby spinach leaves contained four carotenoids, lutein, violaxanthin, β -carotene and neoxanthin, among which lutein took the major quantity at about 39%, which is in agreement with the results found in this study, the content of lutein was higher than β -carotene and α -tocopherol (Bergquist et al., 2006).

 β -carotene is expressed as pro-vitamin A which is rich in green leafy plants like spinach. The carotenoid composition found in this study agrees with that previously reported for spinach, with lutein (50.3-76 mg/kg wet weight) and β -carotene (60.7-83.1 mg/kg wet weight) being the major carotenoids in six spinach genotypes (Kidmose et al., 2001). These molecules are reported to be important for human health. Indeed, lutein is present in high concentration in the retina close to the macula dense, where it protects the eye tissues against damage related to light and the age-related macular degeneration (Carpentier et al., 2009). Another important molecule measured in the current work is the antioxidant, alpha-tocopherol (Vitamin E). Vitamin E is a lipid soluble vitamin with antioxidant and anti-inflammatory effects. In view of the high levels of these compounds in chloroplastic fractions of spinach viz. α -tocopherol and ALA, is seems reasonable to speculate that choroplastic fractions of plants could be used to deliver nutrients that are beneficial to human health.

3.3.1.4. Chlorophyll levels of dried spinach juice/CRF

Chlorophyll a and b levels in dried spinach juice/CRF powders were measured and data are shown in Figure 3.4 (a) and (b).



(a)



(b)

Figure 3.4: Chlorophyll content of spray-dried juice, freeze-dried juice and freeze-dried CRF.

Results are expressed as means \pm SD (n = 3). Results are analysed using post-hoc analysis of variance (ANOVA) and according to a Tukey test with statistical significance at p<0.05, a>b. *Abbreviations: SJ* Spray dried juice, *FJ* Freeze dried juice, *FC* Freeze dried CRF. *Chl.A* chlorophyll a, *Chl.B* chlorophyll b, *Total Chl.* chlorophyll a+b.

If one compares these values with those of carotenoids and tocopherol there is a very similar trend. Syamila et al. (2019) recently compared the chlorophyll content of the freeze dried spinach juice and CRF, with the content of chlorophyll a and b in freeze dried juice being 12.73 and 5.17 mg/g DW. Freeze dried CRF was 45.49 and 20.51 mg/g DW. The results were similar to the current study, the content of chlorophyll a was higher than chlorophyll b in both freeze dried spinach juice and CRF, and the total

chlorophylls were around 4 times greater in CRF. This finding is supported by the study of Fritschi *et al.* (2007) who found that the level of chlorophyll a was higher than chlorophyll b, with the mean chlorophyll a/b ratio of 3.79. The quantity of chlorophyll a is always higher than chlorophyll b in higher plants. In addition, the chlorophyll b is more soluble than chlorophyll a in polar solvents because of its carbonyl group. The highest content of total chlorophylls was found in CRF, then followed by spinach juice (SJ and FJ). This trend is explained by effects of the centrifugation force during sample preparation, which can accumulate as much chloroplasts as possible, due to the fact that chlorophylls concentration is an indirect indicator of chloroplasts (Liang et al., 2017).

3.3.2. Physical properties of dried spinach juice/CRF

Property	SJ	FJ	FC	Fresh CRF
Moisture content/%	$8.78{\pm}0.67^{b}$	$8.45{\pm}0.27^{b}$	3.98±0.77 ^c	83.85±0.21 ^a
Water activity	0.26 ± 0.01^{b}	0.18 ± 0.02^{c}	0.24 ± 0.02^{b}	1.00±0.00 ^a
Solubility/%	63.15 ± 0.43^{b}	73.87±1.16 ^a	42.55±2.44°	40.93±4.64 ^c
Dispersibility/%	96.64±1.56 ^a	98.70±1.70 ^a	93.06±1.02 ^b	94.30±3.80 ^{ab}

Table 3.2: Characterization of chloroplasts from spinach juice and CRF

Data was analysed by one-way ANOVA according to Tukey's test (p<0.05) to access the differences of the samples. Means (n=3) designated with different letters in the same row are significantly different (p<0.05). *Abbreviations: SJ* Spray dried juice, *FJ* Freeze dried juice, *FC* Freeze dried CRF.

The physical properties of dried spinach chloroplast materials are displayed in Table 3.2. Moisture content in food products can significantly affect its appearance, texture and taste, in addition to other impactors that affect shelf life, quality and resistance to bacterial contamination (Isengard, 2001). Fresh CRF had a high moisture content (83.85 \pm 0.21 g/100g), which potentially could promote food degradation and growth of microorganisms, and therefore reduce shelf life. This study shows that drying processes can significantly reduce the presence of water in chloroplast fractions. For example, the moisture content of spray dried and freeze-dried juice were around 8%. Moreover, the moisture content of freeze dried CRF was 3.98%, less than 5%. The recommended value for fruit and vegetable juice powder products needed to guarantee reduced microbiological growth and to enhance stability of products for long-term storage is 5% (Tontul & Topuz, 2017). The relatively lower moisture value in FC compared with FJ could be possibly caused by the preparation steps difference. The centrifugation processing that used to collect CRF materials can remove water and accumulate chloroplast membranes which resulted in higher proportion of total lipids and insoluble proteins.

Similarly, water activity is a crucial indicator which generally affects shelf life of dried products. The water activity found in fresh CRF was 1, which agrees with the water activity of distilled water, this high value of water activity supports the growth of bacteria, yeasts, and moulds (Ölmez et al., 2009). The water activity of CRF after freeze drying significantly decreased to 0.24 ± 0.02 . There was significant difference between SJ and FJ, with SJ showed higher water activity of 0.26 ± 0.01 , and FJ showed the

lowest water activity of 0.18 ± 0.02 , this difference could be affected by drying process. This result was in accordance with the water activity of 0.23 for spray dried and 0.18 for freeze dried spinach juice obtained by Syamila et al. (2019). The water activity of the dried chloroplast powders was in the range of 0.2-0.6 as found in dried fruit and vegetables in other studies (Shishir et al., 2017), with values lower than 0.3 commonly regarded as microbiologically and chemically safe, which can be seen as resistant to the microbiological and enzymatic reaction such as browning and oxidative degradation (Tontul & Topuz, 2017).

It is obvious that FC presented higher water activity but lower moisture content than FJ, one possible explanation could be due to the higher proportion of soluble carbohydrates in FJ can reduce the water activity (Sevenich et al., 2015).

Solubility is defined as the ability of powders to form solution or suspension in water, which is a crucial criterion to estimate the behaviour of powder in an aqueous solution (Tontul & Topuz, 2017). Solubility can influence some properties of powder products, such as the reconstitution of the dry extract or the bioaccessibility of nutrients during digestion. In addition, the solubility of powder directly related to the technological application in food industry, i.e. spray dried products used as food ingredients were more widely accepted owing to better stability and enhanced solubility which enable to satisfactory capability of the powders in the desired applications (Lee et al., 2018c). Importantly, solubility can be affected by many factors, such as temperature, pressure, moisture content, particle size, polarity and the physical state of particles (Caparino et al., 2012). Solubility values in dried juice were 63.15 - 73.87 %,

significantly higher than CRF of 40.93 – 42.55 %. The lower solubility of CRF could possibly be due to the state of the powdered CRF is more compact and less easily penetrated by water by gentle mixing to promote material disintegration to form a suspension. From the other way, the relatively higher lipid content contained in CRF could also leads to a lower solubility due to their hydrophobic nature. This also explains the lower solubility of SJ compared with FJ, as higher lipids contained in SJ (data shown in Table 3.1). Solubility in this study was lower than that of Brazilian ginseng roots powders obtained by SD and FD around 90% (Vardanega et al., 2019).

Dispersibility is a key indicator of a powder property in an industrial setting, it represents the ability of powders to separate into individual particles when dispersed in water with gentle mixing (Sharma et al., 2012). High dispersibility indicates less lump formation and agglomerates of powder that fall apart in the water (Laokuldilok et al., 2015). As shown in Table 3.2, all the samples showed great dispersibility with the value higher than 90%, and dried spinach juice exhibit slightly higher dispersibility than CRF. Similar to solubility, dispersibility can be influenced by the properties of the samples, such as particle size and morphology (Ding et al., 2020). Thus, the morphology, particle size, as well as other properties of the spinach samples were investigated in the following study.

3.3.3.Particle size of chloroplast rich powder preparations

Table 3.3: Particle diameter of powders prepared from spinach juice or a chloroplastrich pellet (μ m)

Parameter	SJ	FJ	FC	Fresh CRF
D4,3	9.81±0.22 ^c	66.44 ± 0.38^{b}	113.95±4.44 ^a	10.01±0.60°
d ₁₀	2.26±0.18 ^c	9.26 ± 0.47^{b}	19.23±1.29 ^a	2.39±0.02°
d 50	7.63±0.06 ^c	48.75 ± 0.48^{b}	96.46±2.62 ^a	6.90±0.23°
d 90	20.23±1.01°	151.62±1.81 ^b	215.93±2.88 ^a	19.10±1.54 ^c
Span	2.36 ± 0.12^{b}	2.92±0.07 ^a	2.04±0.10 ^c	2.42 ± 0.14^{b}

Data was analysed by one-way ANOVA according to Tukey's test (p<0.05) to access the differences of the samples. Means (n=3) designated with different letters in the same row are significantly different (p<0.05). D_{4,3}: volume mean diameter; d₁₀, d₅₀, d₉₀: median volume diameters of 10%, 50% and 90% of the cumulative volume (μ m), respectively; span: polydispersity of the particles. Span: an indication of the width of the distribution. Span is the distance between two points equally spaced from the median. Span= (d₉₀-d₁₀)/d₅₀. *Abbreviations: SJ* Spray dried juice, *FJ* Freeze dried juice, *FC* Freeze dried CRF.

Particle diameter of spinach juice/CRF powders is influenced by the original characteristics and drying conditions. The volume mean diameter of spinach juice (D_{4,3}) was significantly affected by drying techniques, with the value of 9.81 μ m for SJ, and 66.44 μ m for FJ (Table 3.3). These results are in agreement with the reported for Brazilian ginseng roots powders obtained by spray and freeze drying, with D_{4,3} values of 9 and 207 μ m respectively (Vardanega et al., 2019), which suggests that spray dried powders normally produce products with smaller particle size than freeze dried powders. While the D_{4,3} of spinach CRF powders (FC) was nearly twice as FJ. This significant difference is caused by the original characteristics difference between juice

and CRF samples during grinding, that CRF was composed of more compact structures with highly concentrated and thick properties.

The spinach juice and CRF powders used as food ingredient need to be prepared as liquid food before ingestion to perform health benefits. Therefore, the rehydration process in water is worth to be analysed. Particle size is a key parameter influencing the solubility and dispersibility of spinach juice/CRF powders in water. Previous study has shown that superfine powders with particle size in the range of 1-100 μ m can increase the dispersibility because of the increased particle surface area and the breakdown of the cell wall of the vegetable materials (Park et al., 2001; Hu et al., 2012).

From another perspective, the reduced particle size can also improve the nutrients bioaccessibility by increasing the surface area accessible for enzymes to act (Priyadarshani, 2017).

3.3.4.Powder morphology

3.3.4.1. Microscope images of spinach juice/CRF

The morphology of the spinach juice/CRF samples were studied under the light microscope at 400× of magnification. As shown in Figure 3.5, individual green organelles or small clusters can be observed in all cases. Some individual chloroplasts were visualised in the picture with the diameter ranged 4-10 μ m, which is in agreement with the reported diameters for chloroplasts (Nicholls et al., 2013). Others have reported on similar morphological structures for isolated chloroplasts with observed starch granules in pea vine haulm juice (Torcello-Gómez et al., 2019). Syamila et al.

(2019) found that the spray dried juice contains intact chloroplasts. These findings confirm that intact chloroplasts can be released without the restriction of cell wall by juicing and preserved after both spray drying and freeze drying. In order to better understand how preparative methods, impact on chloroplastic structures various chloroplastic preparations were studied using scanning electron microscopy.



Figure 3.5: The light microscope images of SJ, FJ, FC and fresh CRF at magnification of 400×. *Abbreviations: SJ* Spray dried juice, *FJ* Freeze dried juice, *FC* Freeze dried CRF.



3.3.4.2. Scanning electron microscope (SEM) images of spinach juice/CRF

powders

Figure 3.6: Scanning electron microscopy micrographs of spinach chloroplasts *Abbreviations: SJ* Spray dried juice, *FJ* Freeze dried juice, *FC* Freeze dried CRF.

The effect of different drying methods on morphology/surface of spinach juice/CRF powders was observed using scanning electron microscopy (SEM) micrographs (Figure. 3.6).

The SEM images of SJ powders indicate irregularly spherical shaped particles with many shrinkages and dents on their surface. Surface dents result from a complex interaction between various factors, such as capillary forces, inlet temperature, drying rate, rapid wall solidification, and uneven shrinkage at early stages of drying. Shrinkage occurs due to the presence of capillary forces. The surface of the droplets dries upon contacting the hot air atmosphere. Then, water diffuses from the inner core towards the surface, which creates sub-atmospheric internal pressures and eventually result in the collapse of the particles (Abdallah et al., 2018). The dented structures can also be affected by the low inlet temperature and insufficient solid content in the feed solution, which cause the deficient drying of the feed droplets therefore leading to the breakage of film or crust (Maas et al., 2011). Spray drying with higher inlet air temperatures caused faster water evaporation and lead to the formation of smooth particle surfaces with a hard crust, which the hollow particles could not deflate when they were cooled (Nijdam et al., 2006). Higher drying rate can also lead to the rapid wall solidification prior to expansion of the particles and forms a strong crust on the particle surface and restricts the complete bubble inflation, thus resulted in caved-in or dented surfaces (Chinnaswamy et al., 2007).

Chloroplast powders obtained by freeze drying (FJ and FC) had porous and irregular plate shaped particles which are typical characteristics of freeze-dried
products. This phenomenon resulted from ice crystals that formed inside the material during processing, and this helps prevent the shrinkage and collapse of the organelle structure (Chranioti et al., 2016). Similar structures have been reported for freeze dried mango and açaí pulp (Caparino et al., 2012; Lucas et al., 2018). The particle size of FJ and FC as shown in the micrographs was higher than SJ and fresh CRF, which in consistent with the particle size distribution in Table 3.3.

3.4. Conclusion

Overall, the spinach chloroplasts can be released without restriction of cell wall by juicing and sequential centrifugation concentrates the intact chloroplast structures as chloroplast rich fractions (CRF). The quality of spinach chloroplasts can be maintained after drying process, in this study viz. spray and freeze drying. They can potentially be developed as functional food ingredient for the source of macronutrients such as proteins and lipids, as well as micronutrients such as carotenoids, chlorophylls, vitamin E and α -linolenic acid. Both spray drying and freeze drying applied in this study was proved to retain the nutrients in spinach juice at similar level; spinach CRFs contained higher concentration of nutrients (per unit mass) than dried juice. The physical parameters (particle size, solubility, dispersibility etc.) of spinach chloroplast materials after both drying methods will possibly have impact on the nutrients release and delivery in humans. Significantly reduced moisture content (below 10%) and water activity (below 0.3) in spinach chloroplast powders suggest microbiological safe and enhanced shelf stability. Freeze dried CRF (FC) was less soluble and dispersible compared with dried juice (SJ and FJ). The morphology changes of spinach chloroplasts after spray drying and freeze drying was observed by scanning electron microscopy. Spray dried samples exhibited irregular spherical dented structures, while freeze dried samples showed porous and plates shaped particles. These observations highlighting the impact of each drying method on organelle structures and providing a means to differentiate between each method used.

The bioaccessibility of nutrients and *in vitro* digestion of lipophilic nutrients in these materials during in vitro digestion will further be investigated and discussed in following chapters.

Chapter IV

4. Effect of spinach juice pasteurisation on the physicochemical properties of derived powders

4.1. Introduction

In Chapter 3, it was demonstrated that spinach chloroplasts are rich source of macronutrients, such as proteins, lipids and fiber; as well as micronutrients, such as carotenoids, chlorophylls, vitamins (Pérez-Marín et al., 2019). Sufficient consumption and increased intakes of these compounds is associated with a reduced risk of cardiovascular diseases, cancer, obesity and other chronic diseases (Joseph et al., 1999; Wang et al., 2005; Vázquez et al., 2013; Roberts & Moreau, 2016; Torcello-Gómez et al., 2019). Despite these valuable attributes, the consumption of processed spinach remains low in the general population (Klockow et al., 2009). Therefore, there may be advantages in developing new spinach products or functional ingredients to encourage greater consumption, however, a key challenge here is spinach suffers from high perishability, postharvest (Bassetto et al., 2005). In this context, there is a need for developing novel preservation techniques to ensure the qualities of spinach products are retained during storage and to enhance shelf life.

To date, pasteurisation is considered as the most commonly used pre-treatment for food preservation, since this method inactivates some endogenous enzymes involved in spoilage, and eliminates spoilage microorganisms and pathogens (Peng et al., 2017). Despite its utility, pasteurisation causes some undesirable quality losses in terms of colour, flavour, texture, nutrients degradation and functionalities (sensorial, physicochemical, nutritional properties and levels of antioxidants). For this reason, it is important to adjust the temperature and time to keep the physicochemical changes to a minimum but sufficient to inactivate the deleterious enzymes (Anthon et al., 2002).

It is known that the presence of endogenous enzymes in raw or processed fruit and vegetable leads to quality loss during storage. Peroxidases (POD) and polyphenol oxidases (PPO) are the main enzymes that contribute to enzymatic browning which can cause both off-flavours and off-colour formation (Lopez et al., 1994; Negri Rodríguez et al., 2021). Therefore, enzymatic browning is regarded as the secondary loss for fruits and vegetables during processing and storage (Thirumdas et al., 2015). Peroxidase (POD) is usually chosen as an indicator enzyme of thermal processing adequacy due to its high concentration in most plant tissues, and the relatively high heat-resistant ability (Barrett et al., 1995). Additionally, the non-enzymatic browning results from pigment degradation (especially chlorophylls and carotenoids), and Maillard browning reactions during thermal processing (Zhu et al., 2009; Jaeger et al., 2010). Several studies have investigated the effect of different processing methods on the level of pigments retention. High temperature short time processing can increase chlorophyll retention while the carotenoids retention varies at different circumstances (Gupte et al., 1964; Finten et al., 2016).

Pasteurisation is a primary step in processing of spinach. In addition, the production of spinach juice powders is also an alternative approach with the advantages of increasing the consumption period and economic potential, reducing cost for transportation (Çalışkan Koç et al., 2017). The dried spinach powders can be used as a

food colourant and nutritious additive in different formulations (Karaaslan et al., 2008; Koç et al., 2018). Spinach powders can be obtained by freeze drying the pasteurised spinach juice with increased stability during storage. Freeze drying is the most efficient drying technique for nutrients preservation in food industry due to the low temperature which stop the microbiological and deterioration activity (Sagar & Suresh Kumar, 2010). Moreover, the powders produced from fruit and vegetable juice extend the shelf life from weeks to months or years depending on packing and storage conditions (Walkling-Ribeiro et al., 2009; Chauhan et al., 2013).

In recent years, there has been an increasing consumer demand for nutritious products with high sensorial quality and increased shelf life. The trend for healthy, minimally processed foods has driven research and development in novel food processing technologies to produce foods with fresh-like quality that are safe for consumers. However, few studies have assessed the combined impacts of thermal pretreatment with non-thermal drying process on food quality attributes.

The aim of this research is to develop an efficient processing technique to produce dried spinach chloroplast-rich powders that are highly nutritious and retain quality attributes throughout long-term storage. The influence of pasteurisation and freezedrying processes on the physicochemical properties of dried spinach chloroplasts were investigated and key quality attributes determined including enzymatic inactivation, chromatic parameters and physicochemical characteristics. The importance of this work is to establish the 'novel processing' technique, produce better quality of spinach final products which can be developed as raw material or additive in functional food formulations for further processing and stored for subsequent use.

4.2. Sample preparation summary



Figure 4.1: Fresh/pasteurised spinach juice/CRF powders preparation steps

Figure 4.1 illustrates the preparation steps of spinach juice/CRF powders. The freeze drying was used in the following studies. As discussed in Chapter 3, both spray drying and freeze drying can retain similar nutrients level in spinach juice powders. Whereas, from technical perspectives, spray drying is limited since only liquid samples with particle diameter lower than 100 µm can be passed through the nozzle of spray dryer (Armfield FT80 Tall Form Spray Dryer, Ringwood, UK) in this study.

4.3. Results and discussion

4.3.1.Quality changes of fresh and pasteurised spinach juice during refrigerated storage

The main goal of pasteurisation of spinach juice is to eliminate microbial contamination, inactivate endogenous enzymes, and to extend shelf life. Some properties may change under thermal treatment which may affect the food quality and sensory characteristics. Food quality and shelf-life safety are crucial criteria when customer select food products. Therefore, the physical-chemical properties of both non-pasteurised and pasteurised spinach juice were monitored under 4°C for seven days (data shown in Table 4.1).

pH plays a key role in food preservation as it determines the extent of microbiological activity. Generally a low pH (pH \leq 4.6) tends to prevent microbial growth in fresh juices (Tola et al., 2018). In general vegetables are 'low acid' foods (pH \geq 4.6), and so require stronger thermal treatments to achieve the desirable shelf life. Heat treatments in the range of 82-93°C are commonly used to kill bacteria in low-acid foods (pH \geq 4.6) (Barrett et al., 2012). In this study, the initial pH value of non-pasteurised juice (NPJ) was pH 7. After pasteurisation this value slightly increased to around pH 7.35-7.59, and was attributed to the release of NH₃ from the breakdown of cell membranes (Kung et al., 1991). The pH of all juice samples significantly decreased to pH 5.75-5.93 over seven days of refrigerated storage. This is probably due to the growth of microorganisms that cause juice spoilage via the fermentation of sugars. In turn, this action produces organic acid hence reduce the pH and changes in the Brix

value. Similar results were also found in carrot and spinach juice which initially showed a neutral pH of 7-8 that significantly decreased to pH 4-6 during refrigerated storage for fourteen days (Khandpur et al., 2016). The decreased pH was also in agreement with the reports for refrigerated passion fruit pineapple and mango juice over the twelve day period (Kaddumukasa et al., 2017). There was no significant difference of pH value in the different treated spinach juices during refrigerated storage. Decreased pH was observed in all cases over the seven days period, while stable pH was only found in pasteurised juice in initial three days of storage (4 °C) (data not shown). Low pH encourages the growth of acid-tolerant bacteria, causing sample deterioration that reduces product storage stability. Both fresh and pasteurised juice after seven-days of refrigerated storage showed unpleasant colour and smell which confirms the microorganisms induce spoilage.

°Brix values showed a significant decrease in spinach juice samples after pasteurisation, decreased from 5.9 to 3.7. This reduction could be due to the reaction between reducing sugar and the alpha amino acids after heating (data shown in Table 4.1). In addition, a further reduction of Brix value was found in fresh juice over seven days storage at 4°C. This decline may probably have been due to microbial metabolic activities which causes sugars in the samples to be converted to organic acids, hence decreased the pH and shortened shelf life. Similar behaviour was also supported by Khandpur and Gogate (2016) who reported a significantly decrease of Brix value ranged from pH 6-8 to 3-5 in both fresh and thermally treated spinach and carrot juice over 10 weeks storage at 4°C.

Thermal treatments like pasteurisation were shown to inactivate enzymes and to eliminate pathogens, which is important for food preservation. Among all endogenous enzymes found in fruits and vegetables, peroxidases (POD) are a particularly heat resistant group of enzymes, which play important roles in quality deterioration (Umair et al., 2022). In this study, POD activity was used as an indicator for the inactivation of endogenous enzymes during spinach juice pasteurisation. It can be observed that 96-99% of POD activity was reduced in pasteurised spinach juice when compared with non-pasteurised juice. In addition, the inactivation of POD activity remained after 1 week of refrigerated storage. Therefore, both mild and intense pasteurisation conditions (70 °C 15s and 90 °C 5min) were sufficient to inactivate spinach POD producing a stable product with improved shelf life. Similar results were reported by Wattanakul et al. (2020), who found that the reduction of POD activity was around 92.3 - 95% in pea vine haulm juice after hot water blanching (100°C 2 min) and steam blanching (100°C 4 min). It was also reported that spinach leaf blanched at 95°C for 1 min or steamed for 3-6 min was sufficient for reduction of peroxidase activity (Onayemi et al., 1987; Okoli et al., 1988).

The particle size distribution of fresh and pasteurised spinach juice during refrigerated storage was also investigated in this study. The mean volume weighted $D_{4,3}$ value of non-pasteurised juice was around 4.4 µm, and this value significantly increased to 40.55 and 102.97 µm after mild and intense pasteurisation. Moreover, the average particle size ($D_{4,3}$ value) of spinach juice was significantly increased as the pasteurisation level (time and temperature) increased. The larger particles in pasteurised

spinach juice could be caused by the protein denaturation and aggregation events during thermal processing (Bernat et al., 2015; Atalar et al., 2019). Additionally, the high $D_{4,3}$ values suggest that the particles are heavier and tend to settle down rapidly during storage. The sedimentation during the storage further leads to the interactions between particles and aggregation, which explains the increased particle size at the end of the storage (7 days, 4°C).

Table 4.1: Physicochemical changes of fresh and pasteurised spinach juice during

 refrigerated storage

	Days	pН	°Brix	% reduction of POD activity	Particle size D _{4,3} /µm
Fresh juice	0	7.00 ^a	5.9 ^a	0	4.40 ^e
	7	5.78 ^b	4.8 ^b	0	11.82 ^e
Mild pasteurised juice	0	7.35 ^a	3.9 ^c	96 ^b	40.55 ^d
	7	5.93 ^b	3.7 ^{cd}	97 ^a	61.92 ^c
Intense pasteurised juice	0	7.59 ^a	3.9 ^c	98 ^a	102.97 ^b
	7	5.75 ^b	3.7 ^d	99 ^a	170.78 ^a

Data was analysed by one-way ANOVA according to Tukey's test (p<0.05) to access the differences of the samples. Means designated with different letters in the same row are significantly different (p<0.05). *Mild pasteurised spinach juice (70°C 15s); Intense pasteurised spinach juice (90°C 5min)*.

4.3.2.Effect of spinach juice pasteurisation on the nutrient content of derived powders (dried juice and dried pellet)

4.3.2.1. Macronutrients content of fresh/pasteurised spinach juice/CRF

The results of macronutrients content of fresh/pasteurised spinach chloroplasts were presented in Table 4.2. The proteins showed the largest proportion in all samples, with a ratio of 38.5~58.6% approximately. In CRF samples, a significant increase of proteins and decrease of lipids was observed after pasteurisation. While, for juice samples, the proportion of proteins and lipids maintained relatively stable after pasteurisation. This is likely due to the high heat cause the break of lipid-proteins complex in pasteurised spinach juice, while after centrifugation, the protein-rich pasteurised CRFs were collected, the lipids released in the supernatant were discarded.

Significant higher proteins and lipids levels were present in CRFs (FC, MC, IC) compared with juice (FJ, MJ, IJ), which is attributed to the synthesis and concentration of thylakoid membranes, that are rich in proteins and galactolipids, and allied lipid soluble nutrients (carotenoids, chlorophylls and fatty acids) in chloroplasts (Kirchhoff et al., 2002). In contrast, lower ash (total minerals) and carbohydrates were obtained in CRFs, due to the fact that minerals and some low molecular weight carbohydrates are more soluble and therefore abundant in supernatants. The ash (total minerals) content maintained relatively stable after heat processing in both dried juice and CRF, since minerals are resistant to the light, heat and oxygen. As reported by Azima et al. (2020), who indicated that cooking methods such as boiling, steaming and stir frying have little effect on Fe, Zn and Mg profile of green leafy vegetables (e.g. spinach, pumpkin leaves,

squash leaves and green mustard).

In general, pasteurisation can retain macronutrients in dried spinach juice. The significant changes in terms of the proteins and lipids in CRFs is mainly due to the denaturation by heat and physical separation.

/%DW	FJ	FC	MJ	MC	IJ	IC
Proteins	39.6±0.4 ^{de}	50.6±0.5°	41.2 ± 0.1^{d}	58.6 ± 0.2^{a}	38.5±0.9 ^e	$54.9{\pm}1.0^{\text{b}}$
Lipids	11.8 ± 0.4^{d}	28.3±0.4 ^a	11.5 ± 0.5^{d}	20.3±0.7 ^b	10.7 ± 0.5^{d}	18.1±0.3 ^c
Ash	20.8±0.2 ^a	7.2 ± 0.1^{d}	18.2±0.5 ^b	7.1 ± 0.1^{d}	18.9±0.3 ^b	7.7 ± 0.1^{d}
Carbohydrates	27.8	13.9	29.1	14.0	31.9	19.3

Table 4.2: Nutrients composition of spinach chloroplasts

Data was analysed by one-way ANOVA according to Tukey's test (p<0.05) to access the differences of the samples. Means (n =3) designated with different letters in the same row are significantly different (p<0.05). *Abbreviations: FJ Fresh freeze dried dried freeze dried CRF; MJ Mild pasteurised freeze dried juice; MC Mild pasteurised freeze dried cRF; IJ Intense pasteurised freeze dried juice; IC Intense pasteurised freeze dried CRF.*

4.3.2.2. Fatty acids composition of fresh/pasteurised spinach juice/CRF

The fatty acids (FAs) content and composition of unpasteurised and pasteurised spinach chloroplasts were displayed in Figure 4.2 and expressed as per gram of dry weight and lipid. As can be seen, these samples contain higher polyunsaturated fatty acids (PUFAs), with the most abundant form of fatty acid is the α -linolenic acid

(C18:3n-3, ALA) (n-3), followed by linoleic acid (C18:2n-6, LA) (n-6). These are essential fatty acids (EFAs) as they cannot be synthesised by the human body. This found is in accordance with the fatty acids in pea vine haulm and pea leaves chloroplasts (Dorne et al., 1989; Torcello-Gómez et al., 2019).

On a dry mass basis, the most obvious observation was the significant loss of ALA in pasteurised CRFs, which linked to the significant decrease of lipids in previous section. In contrast, oleic acid (C18:1) and linoleic acid (C18:2, LA) proved to be more heat stable than ALA, with no significant changes after pasteurisation. Moreover, the fatty acids profile as an essential indicator of the nutritional value of the oil (Harhar et al., 2011), did not change after the pasteurisation step (when heating the spinach juice at 70 °C and 90°C). Based on the previous study, fatty acid profile is more affected by heating temperature than time (Liu et al., 2018), olive oil which contains lower PUFAs, visible changes can be demonstrated in ALA when heated at 180 °C or higher at 220°C, since SFAs are more resistant to thermal treatment (Dordevic et al., 2020). Unlike the olive oil, which contains high MUFA, green leafy vegetables such as spinach usually contains higher proportional of PUFAs in the form of ALA (60-70% of total FAs) (Saini et al., 2014; Kim et al., 2018), which oxidize rapidly during storage and heating, therefore, actions should be made in order to maintain the quality of products.

On a lipid mass basis, both dried pasteurised juice and CRFs showed an increasing trend in the levels of α -linolenic acid and palmitic acid content. This could be due to the change of fatty acid profile after heating, that the galactolipids hydrolysis may cause release of FFA (mainly ALA).



Figure 4.2: Fatty acid composition of fresh/pasteurised dried juice and CRF

Results are expressed as means \pm SD (n = 3) and analysed using post-hoc analysis of variance (ANOVA) and according to Tukey test with statistically significant at p<0.05, a>b. *Abbreviations: FJ Fresh freeze dried juice; FC Fresh freeze dried CRF; MJ Mild pasteurised freeze dried juice; MC Mild pasteurised freeze dried CRF; IJ Intense*

pasteurised freeze dried juice; IC Intense pasteurised freeze dried CRF.

4.3.2.3. Chlorophyll content of fresh/pasteurised spinach juice/CRF

The effect of pasteurisation on colour degradation was evaluated following the analysis of chlorophyll (Figure 4.3) and carotenoids contents (Figure 4.4). Chlorophyll a and b are the two major chlorophylls present in plant chloroplasts (thylakoid membranes). Chlorophyll a is the most abundant form and is typically found at twice that of chlorophyll b in plant tissues. A significant loss of chlorophylls was observed in CRFs after pasteurisation, which is linked to the decrease in total lipid content as their lipophilic nature. Previously studies indicate that chlorophyll retention during processing of vegetables depends upon temperature and length of heat treatment, and that thermal degradation rates of chlorophyll a are two to six times higher than chlorophyll b in spinach puree (Canjura et al., 1991; Schwartz et al., 1991). Thermal degradation of chlorophylls a and b in green peas were studied across the temperature range of 70-100 °C; faster degradation of chlorophylls were detected at temperatures higher than 80 °C (Erge et al., 2009). In this study, intense pasteurisation caused significant (p<0.05) degradation of chlorophylls in dried juice. The decrease of chlorophylls is attribute to the high heat treatment induce the loss of central magnesium atom of the chlorophyll porphyrin ring, and converts chlorophylls to pheophytins and other derivatives, such as chlorophyllide and pheophorbide, which also lead to the colour change from green to olive brown (Cubas et al., 2008). It was reported that 1g dried spinach thylakoid powders can provide 0.3 mg chlorophylls (Li et al., 2019a), in

contrast, in this study chlorophylls were found higher with the value of 16.8 and 66.3 mg/g DW in fresh powdered spinach juice and CRF.



Figure 4.3: Chlorophylls content of fresh/pasteurised dried juice and CRF

Results are expressed as means \pm SD (n = 3) and analysed using post-hoc analysis of variance (ANOVA) and according to Tukey test with statistically significant at p<0.05, a>b. *Abbreviations: FJ Fresh freeze dried juice; FC Fresh freeze dried CRF; MJ Mild*

pasteurised freeze dried juice; MC Mild pasteurised freeze dried CRF; IJ Intense pasteurised freeze dried juice; IC Intense pasteurised freeze dried CRF.

4.3.2.4. Carotenoids and vitamin E content of fresh/pasteurised spinach juice/CRF

Carotenoids are important pigments with antioxidant properties that are widely found in vegetables. β -carotene and lutein are the predominant carotenoids in plants. Changes in carotenoid levels between unpasteurised and pasteurised dried chloroplasts are presented in Figure 4.4. On a dry mass basis, carotenoids remain stable after both mild and intense pasteurisation in dried juice but remarkably these molecules decreased in CRFs. These changes correlated with a general trend in the loss of total lipids in CRFs after thermal processing (Table 4.2). In addition, the degradation of carotenoids can be also caused by isomerisation of all-trans- β -carotene to cis forms followed by oxidation (Aman et al., 2005; Lemmens et al., 2010). The extend of such degradation is associated with the duration and severity of heat processing (Seybold et al., 2004). Indeed, similar results are supported by Wattanakul et al. (2022) who found 34% and 40% of the β -carotene and lutein losses and increase of cis- β -carotene in heat treated (steam or pasteurisation) pea vine haulm CRF. Reductions in carotenoids levels in heated vegetables have also been reported for boiled cabbage (34.45%) and spinach (24.78%) (Dumbrava et al., 2016). In this instance, the high temperatures induce the breaking of bonds present in the carotenoids and the reformation of single or double bond following isomerisation (Jonsson, 1991).

Similarly, vitamin E is the major lipophilic antioxidant that is responsible for

preventing membrane lipid peroxidation in plants, animals and fungi. This class of molecule consists of four tocopherols (α , β , δ and γ) and the corresponding tocotrienols $(\alpha, \beta, \delta \text{ and } \gamma)$ that contain unsaturated side chains. Of this group of compounds α tocopherol is considered to be the most effective biological antioxidant (Chun et al., 2006; Cruz et al., 2013). The content of α -tocopherol in dried fresh and pasteurised spinach chloroplasts are shown in Figure 4.4. The amount of vitamin E (VE) in FJ and FC was 0.26 and 0.70 mg/g of dry weight, higher VE in FC linked to the higher chloroplasts contained in CRF, since a-tocopherol is mainly situated inside chloroplasts (Barbara et al., 2000). A significant loss of VE was determined in pasteurised juice/CRF compared with fresh samples, the degradation of α -tocopherol could be due to the low heat stability of α -tocopherol from spinach juice/CRF materials (Knecht et al., 2015). However, between mild and intense pasteurisation conditions, it seems like the latter can retain higher VE levels, with 0.13 and 0.53 mg/g DW contained in the IJ and IC while 0.04 and 0.20 mg/g DW found in MJ and MC. This could be due to the increased sufficiency of α -tocopherol extractability following denaturation of proteins and a complete cell disruption as a result of higher temperature and time, which consequently cause more release of a-tocopherol from plastoglobule within the chloroplasts (Wójtowicz et al., 2021). While, as reported by Lee et al. (2018b) α-tocopherol retention varies across heating methods and food matrices. Cooking green leafy vegetables such as broccoli, spinach and perilla leaf leads to a significant higher vitamin E levels, while cooked root vegetables such as potato, sweet potato and carrot lead to lower retention of vitamin E.



Figure 4.4: Carotenoids and vitamin E content of fresh/pasteurised dried juice and CRF

Results are expressed as means \pm SD (n = 3) and analysed using post-hoc analysis of variance (ANOVA) and according to Tukey test with statistically significant at p<0.05, a>b. *Abbreviations: FJ Fresh dried freeze juice; FC Fresh dried freeze CRF; MJ Mild pasteurised freeze dried juice; MC Mild pasteurised freeze dried CRF; IJ Intense*

4.3.3.Physical properties of fresh/pasteurised spinach juice/CRF

Properties	FJ	FC	MJ	MC	IJ	IC
Moisture content/%	7.7±0.2 ^c	4.7 ± 0.2^{d}	9.3±0.1 ^a	3.8±0.1 ^e	8.3±0.2 ^b	3.8±0.1 ^e
Water activity	0.23±0.01 ^c	$0.25{\pm}0.00^{b}$	0.34±0.00 ^a	0.22±0.01 ^c	0.17±0.01 ^e	$0.19{\pm}0.01^d$
Solubility/%	68.6±0.6 ^a	25.3 ± 0.3^d	41.3±0.2 ^c	10.1±0.1 ^e	47.0 ± 0.0^{b}	10.2±0.1 ^e
Dispersibility/%	91.0±2.2 ^a	65.0±5.1 ^b	61.0±2.2 ^b	22.4±0.8 ^c	60.9±1.9 ^b	22.9±1.1°

Table 4.3: Characterization of freeze-dried spinach juice and CRF powders

Data was analysed by one-way ANOVA according to Tukey's test (p<0.05) to access the differences of the samples. Means (n =3) designated with different letters in the same row are significantly different (p<0.05). *Abbreviations: FJ Fresh dried freeze juice; FC Fresh dried freeze CRF; MJ Mild pasteurised freeze dried juice; MC Mild pasteurised freeze dried CRF; IJ Intense pasteurised freeze dried juice; IC Intense pasteurised freeze dried CRF.*

The moisture content of dehydrated materials plays an important role in preservation safety and defines the amount of water in any given food system. High moisture content is indicative of fast food degradation. Thus, producing powders can significantly reduce the moisture content available for chemical reactions and the growth of microorganisms (Mathlouthi, 2001). Higher moisture content was observed in dried spinach juice (7.7-9.3 %) compared to CRF materials (3.8-4.7 %). Similar results were reported by (Koç & Dirim, 2018), with the moisture content of spray dried spinach powders is in the range of 5.56-9.29 %, wet basis.

Water activity is a critical indicator for powders microbiological safety during storage. Different from moisture content, water activity measures the free water available in food which is responsible for the biochemical reactions to occur (Quek et al., 2007). The higher water activity indicates shorter shelf life, since more free water available for deteriorative reactions (Tontul & Topuz, 2017). The moisture content observed in this study was lower than 10% and the water activity was around 0.3. This guaranteed the powders' microbiological safety and stability during long-term storage (Tze et al., 2012a).

Solubility is a fundamental physical property/characterization of powder products which is the evaluation of the behaviour of the powders in the aqueous phase. This indicator is directly linked to the technological application in food products, since powders used as food ingredients must exhibit good solubility which enable the desired functionalities for food design (Bicudo et al., 2015; Vardanega et al., 2019). The solubility of fresh dried juice and CRF (FJ and FC) was 68.6 and 25.3%, and significantly decreased to 41.3-47.0%, and 10.1-10.2% after mild and intense pasteurisation. The solubility which due to the greater surface area available for hydration (Kuck et al., 2016). The higher solubility of dried juice could be attributed to its lower lipid content but higher ash and carbohydrates which are more solubilised. In addition, thermal treatment and protein denaturation caused a marked and irreversible reduction of solubility. As reported by (Perez-gago et al., 2001)) the solubility of the protein decreased as a function of heating time and temperature (70-100 °C, 5-20 min).

Dispersibility is the ability of powders to separate into individual particles when dispersed in water with gentle mixing (Laokuldilok & Kanha, 2015). Less lumps and agglomeration fall apart in water when the powder dispersibility is high (Suhag et al., 2021). Table 4.3 shows a significant effect of pasteurisation conditions on the dispersibility of dried spinach juice and CRF, as fresh juice and CRF showed higher dispersibility value of 91% and 65% respectively. Significantly decreased dispersibility of dried juice and CRF was observed after both mild and intense pasteurisation, which was in the range of 60.9-61.0% and 22.4-22.9%. This observation indicates that dispersibility was greatly affected by pasteurisation but no significant difference between mild and intense pasteurisation were seen. This index is important in industrial settings (Sharma et al., 2012).

After pasteurisation and freeze drying, the dried spinach juice/CRF were ground to a fine powder with mortar and pester, and then filtered with 250 μ m sieve. After these processing steps, the changes in particle size were noted (data shown in Table 4.4). The mean particle size (D_{4,3}) of dried spinach juice was increased from 43.48 μ m to 69.53 μ m and 101.38 μ m after mild and intense pasteurisation, and the dried CRF was increased from 81.91 μ m to 182.11 μ m and 167.52 μ m respectively. Moreover, the span of dried juice and CRF was decreased after both pasteurisation conditions. Spinach powder obtained by freeze drying the spinach leaves in previous studies showed that the D_{4,3} value of 315.3±70.22 μ m, D₅₀ value of 182.3 μ m ± 10.69 μ m which is significantly higher than our results. Moreover, micronisation was used to reduce the particle size of freeze-dried spinach leaves. After wet and dry micronisation the D_{4,3} value was decreased to 121.4 μ m ± 13.35 μ m and 28.1 μ m ±1.18 μ m respectively (Różyło et al., 2022). Micronisation as a technology that normally used in pharmaceutical industry, which is a mechanical and high shearing operation to downsize the particles of food material to the micron range (Dhiman et al., 2021). It has been recently proved its potential application in food industry.

/µm	FJ	FC	MJ	MC	IJ	IC
D _{4,3}	43.48 ± 2.00^{d}	81.91±1.91 ^c	69.53±3.41°	182.11±9.94 ^a	101.38±4.97 ^b	167.52 ± 7.50^{a}
d ₁₀	4.18 ± 0.27^{d}	9.87±1.03 ^c	13.30±1.79 ^c	54.07±4.00 ^a	15.23±0.64 ^c	38.13±1.90 ^b
d ₅₀	$29.36{\pm}1.77^{d}$	65.85±2.43°	68.09±3.45°	175.29±6.58 ^a	91.83±6.29 ^b	165.76±2.10 ^a
d ₉₀	109.80±4.77°	178.05 ± 3.22^{b}	126.42±6.48°	305.08±14.57 ^a	$202.07{\pm}7.82^{b}$	291.09±14.99ª
Span	3.60±0.07 ^a	2.56 ± 0.08^{b}	1.66 ± 0.08^{d}	1.43±0.02 ^e	2.04±0.07 ^c	1.53±0.08 ^{de}

Table 4.4: Particle size of powdered spinach juice/CRF

D_{4,3}: volume mean diameter; d₁₀, d₅₀, d₉₀: median volume diameters of 10%, 50% and 90% of the cumulative volume (µm), respectively; span: polydispersity of the particles. Span: an indication of the width of the distribution. Span is the distance between two points equally spaced from the median. Span= (d₉₀-d₁₀)/d₅₀. *Abbreviations: FJ Fresh dried freeze juice; FC Fresh dried freeze CRF; MJ Mild pasteurised freeze dried juice; MC Mild pasteurised freeze dried CRF; IJ Intense pasteurised freeze dried juice; IC Intense pasteurised freeze dried CRF.*

4.3.4.Powder morphology

4.3.4.1. Light microscope images of fresh/pasteurised spinach juice/CRF powders

Light microscope was used to observe the microstructure of the fresh and

pasteurised dried juice/CRF samples (Figure 4.5). All the dried powdered juice/CRF samples were hydrated in RO water as suspension before loading for observation, and then analysed using a light microscope with the polarised lens. In all cases, green organelles which corresponds to the intact liberated chloroplasts were observed. Individual chloroplasts with diameter in the range of 5-10 µm can be found in fresh samples, which is in agreement with the chloroplasts in higher plants (Block et al., 2007). This indicates that the chloroplasts were released from plant cells during the juicing process by disrupting its plant tissue. While, larger and darker green coloured clusters were visualized in pasteurised samples, which could be explained by the aggregation of proteins, and the loss of pigments caused by high temperature during pasteurisation. Chlorophylls are responsible for the green colour in spinach chloroplasts. The colour change from bright green to olive brown during heating is caused by conversion of chlorophylls to pheophytins due to the loss of central magnesium ion (Gaur et al., 2007). The extent of chlorophyll degradation was highly depending on the thermal severity of the process, with intense pasteurised samples showed more loss of chlorophylls and darker olive brown colour than mild pasteurised samples. In order to balance the benefits of thermal processing and the quality of food products (microorganisms, appearance, flavour and nutrients), high temperature short time (HTST) technique can shorten the exposure to heat and preserve green colour in vegetables (Aamir et al., 2013). While the colour differences seems to be more obviously visualised between powdered samples as shown in Figure 2.5. The CRF powders showed darker greenness than juice powders, and pasteurised powders showed

loss of greenness and brightness. This is due to the fact that other than the impact of pasteurisation on chlorophyll degradation, particle size additionally influenced the colour of the grinded powders (Table 4.4). As reported by Haas et al. (2019), increased particle size cause decreased brightness which can be attributed to decreased light scattering of samples. Therefore, the brighter appearance of powders was observed in powders with lower mean particle size.



Figure 4.5: Light microscope images of fresh and pasteurised spinach chloroplasts at 400x magnification. *Abbreviations: FJ Fresh dried freeze juice; FC Fresh dried freeze CRF; MJ Mild pasteurised freeze dried juice; MC Mild pasteurised freeze dried CRF;*

4.3.4.2. Scanning electron microscope (SEM) images of fresh/pasteurised spinach juice/CRF

The morphology of freeze-dried spinach juice/CRF samples was examined using a scanning electron microscope (SEM) at 250x and 1000x magnification (Figure 4.6). As can be observed from Figure 4.6, the pasteurisation conditions resulted in powders with different particle morphologies. First of all, the spinach powder samples were all obtained by freeze drying and thus exhibited an irregular and broken glass structure. This has been reported to be typical structure characteristics of freeze-dried products (Amin et al., 2021). Moreover, the intense pasteurisation resulted in largest particles followed by mild pasteurised spinach juice/CRF and fresh juice/CRF (data shown in Table 4.4). In addition, both the fresh/pasteurised dried spinach CRFs exhibited larger particles than juice. The increased size of particles observed after pasteurisation can be explained by the aggregation of proteins in spinach juice during heating. And the extent of increased particle size was depending on the severity and duration of thermal processing. Spinach CRFs were obtained by centrifuging the juice and only collected the pellet which thus were more concentrated than juice. In this consequence, CRFs contained more chloroplasts and showed larger particle size than juice. The SEM morphology of spinach juice/CRF powders were similar to that freeze-dried functional tea powders and blackberry powders (Franceschinis et al., 2014; Vardanega et al., 2019).



Figure 4.6: Scanning electron microscope images of fresh and pasteurised spinach chloroplasts at x250 and x1000 magnification. Abbreviations: *FJ Fresh dried freeze juice; FC Fresh dried freeze CRF; MJ Mild pasteurised freeze dried juice; MC Mild pasteurised freeze dried cRF; IJ Intense pasteurised freeze dried juice; IC Intense pasteurised freeze dried CRF.*

4.4. Conclusion

The conclusion in this chapter can be summarized as follows:

- Pasteurisation of spinach juice is sufficient to inactivate endogenous enzymes linked to reduced shelf-life. Peroxidase activity after mild (70 °C, 15 s) and intense (90 °C, 5 min) pasteurisation was reduced to 96-99% of its activity. Therefore, freeze drying combined with thermal pre-treatment produces spinach chloroplast powders with guaranteed nutrients stability and shelf life.
- Pasteurisation of spinach juice prior to making CRF powders can cause some losses in nutrient concentration, including total lipids (28-36 %) and lipophilic nutrients such as carotenoids (51-58 %), α-tocopherol (29-71 %), chlorophylls (55-70 %), fatty acids especially α-linolenic acid (24-29 %); pasteurised spinach juice powders seems to retain the nutrients level better than CRFs, and only chlorophylls were significantly reduced after intense pasteurisation.
- The moisture content (below 10%) and water activity (0.17~0.34) in all these dried spinach juice and CRF powders were lower than the parent juice. The lowest moisture content was 3.8 % which found in both MC and IC; and the lowest water activity was 0.17 in IJ. The results suggest all these samples would be

microbiologically safe and would have enhanced shelf stability.

- Fresh spinach juice (FJ) showed higher solubility and dispersibility than CRF (FC).
 Moreover, the solubility and dispersibility was reduced after both mild and intense pasteurisation in spinach juice and CRF samples.
- Both mild and intense pasteurisation of spinach juice affected the morphology of the subsequent freeze-dried samples. The colour loss and aggregation were observed in all pasteurised spinach samples. This quality degradation was highly depending on the thermal severity of the process, with intense pasteurised juice/CRF powders showed olive brown colour and largest particle size.

In this chapter, thermal pre-treatment together with freeze drying was proved to be a method to produce spinach chloroplast powders. In order to balance the benefits of thermal processing and the quality of food products, mild pasteurisation (70 °C 15s) of spinach juice was recommended. While, in the following chapters, it is still worth to investigate different thermal pre-treatment such as blanching on the quality of spinach juice/CRF powders (Chapter 5). Moreover, the pasteurised spinach juice/CRF was selected as samples for determine the impact of heat processing on digestibility of their nutrients using *in vitro* static digestion model in Chapter 6 and 7.

Chapter V

5. Effects of blanching on properties of freeze-dried spinach juice/CRF powders

5.1. Introduction

Pasteurisation of spinach juice before freeze drying is an efficient method to inactivate endogenous enzymes associated with spoilage, preserves nutrient composition, and prolong shelf life. In addition to pasteurisation, blanching is also used as a practical thermal pretreatment that can be applied on green leafy vegetables before consumption or preparation (freezing, drying or canning). Blanching can be defined as a process of heating vegetables to a desired temperature and maintain for a specific time for the purpose of breaking up enzymes contained in the tissue. The typical temperature and duration time of water blanching is around 70-100 °C less than 10 min (Lin et al., 1995; Njoroge et al., 2015). The benefits of blanching can be the elimination of pathogens, the inactivation of enzymes, preservation of colour, flavour, and nutritional quality. Moreover, the entrapped air and metabolic gases within vegetable cells are removed and replaced by water during blanching, allowing a semicontinuous water phase to form, which promotes more uniform crystal growth during freezing (Aamir et al., 2013). This structure change also increases cell membranes permeability, result in faster drying rate and higher nutrients retention (Severini et al., 2005; Zhu et al., 2021). Blanching can be conducted in several ways, among which hot water and steam are the most traditionally used methods in the food industry. However, food quality is greatly affected by the type and extent of blanching. Water blanching can lead to leaching of some nutrients (minerals and vitamins), softening vegetable tissues and causing colour loss (chlorophylls and other pigments degradation, Maillard browning) (Severini et al., 2016). Therefore, blanching should be conducted carefully to maximum the benefits of its use to preserve nutrients levels. As mentioned, peroxidases are one of the most heat resistant endogenous enzymes present in vegetables which are associated with food spoilage. Importantly, this class of enzyme can be used as a marker to assess the impacts of blanching on edible plant tissues (Gökmen et al., 2005).

In addition to blanching, drying is a common food processing technique, which is usually applied after blanching to further stabilize vegetables tissues (Mujumdar et al., 2020; Zhu et al., 2021). Freeze drying, as described in Chapter 3 was shown to retain nutrients in both spinach juice and CRF preparations. Indeed, in recent studies, it has been reported that freeze drying of blanched spinach leaves (85 °C 3min) maintained the nutritional content and colour retention of tissues (Joshi et al., 2021).

Therefore, the aims of the current chapter are to determine the impacts of blanching of spinach. To this end, fresh spinach leaves were blanched at 100 °C for 0-90s before juicing, and the physical properties of blanched spinach juice were monitored during refrigerated storage for 7 days. This information was used to determine the optimum blanching condition to preserve spinach juice. The blanched spinach juice/CRF were then freeze dried to stabilize the nutrients composition, to improve quality and extend shelf life.





Figure 5.1: Fresh/blanched spinach juice/CRF samples preparation steps

5.3. Results and discussion

5.3.1.Quality changes of fresh and pasteurised spinach juice during refrigerated storage

The blanching process was adopted for use in the inactivation of peroxidase (POD) activity, and for the assessment of the effects of blanching on other quality attributes of spinach samples. In order to balance nutritional quality and acceptable appearance quality of spinach products, spinach leaves were blanched at 100 °C for 0-90s before juicing, the physical properties of fresh/blanched spinach juice were monitored at 4°C for 7 days. The results of this assessment are summarised in Table 5.1.

The pH of fresh/blanched spinach juice was around 6.49-6.96, which indicates a neutral pH. Blanching cause a slightly decrease in pH within 60s heat processing, these changes being time dependent. The changes in pH are associated with the fact that heating causes the release of organic acids that decrease the pH level of the spinach solutions as compared to the fresh samples. These organic acids are further leached out into blanching water after 90s of heating leading to a further decrease in acidity (Arroqui et al., 2001). The increase in pH values were also reported in 'New Zealand spinach' (*Tetragonia expansa* Murr.) blanched at 95-97 °C for 2-3 min and turnip green blanched at 90 °C for 1-2 min (Grzeszczuk et al., 2007; Martínez et al., 2013). According to Severini and colleagues, 47% of the ascorbic acid content in broccoli is reduced after 90s hot water blanching, and the extent of loss in ascorbic acid was associated with blanching time (Severini et al., 2016)

The blanched spinach juice produced a significantly decrease of pH during the 7 days refrigerated storage, reaching values between 5.34-5.81. No significant pH changes were observed in fresh samples during storage under the same conditions. The lower pH values in blanched juice indicated a higher microbially stability than fresh juice during storage (4 °C 7 d). Moreover, the changes in pH also lead to chlorophyll degradation and thus affect the colour of the vegetable samples, higher stability of chlorophylls at alkaline pH levels.

The juice appearance can be seen in Figure 5.2. Browning and turbidity are the major factors which affect quality of fresh and pasteurised spinach juice during storage. Thermal processing is generally applied to improve quality and extend shelf life of juice

products. However, it will also cause the negative effects on the colour and translucency. Indeed, fresh spinach juice had a vibrant green colour and differing changes in the green pigmentation following blanching (Figure 5.2(a)). The loss of green colour (from bright green to olive brown) in the blanched samples could be mainly attributed to pheophytisation, that chlorophyll a and b converted to their corresponding pheophytins under heat treatment (Martins et al., 2002).

It also can be observed that spinach juice became more translucent with increased blanching time. The higher translucency of heated juice maybe due to the inactivation of browning-associated enzymes, and/or reduced phenolic compounds and proteins in the juice (LI et al., 2009).

Meanwhile, the chlorophylls degradation and decrease in translucency also occurs in spinach juice during refrigerated storage. As shown in Figure 5.2(b), the fresh spinach juice still retains a green colour after 7 days of refrigerated storage, due to the low acidic environment (pH 6.07). In contrast, after 7 days storage at 4 °C, the blanched juice tended to show an olive brown colouration under more acidic conditions (pH 5.34-5.81), with the intermediate blanched samples (10-60 s) showing a more brown colour, 90s blanched juice still maintained its green colour. It was reported that high temperature short time (HTST, 150 °C 80s) processing can increase the chlorophyll retention in spinach initially (Gupte et al., 1964). This could explain why spinach leaves blanched at 100°C for 90s in this study is sufficient enough to retain the chlorophylls in juice samples during refrigerated storage within one week. Overall, HTST combined with pH adjustment was proved to have positive impact on chlorophyll retention. The chlorophyll degradation extent also largely depends on the thermal severity during the processing (Gaur et al., 2007). It is therefore important to investigate the effect of less drastic thermal treatment while maximizing the retention of chlorophyll in thermally processed food.

^oBrix values are commonly used to evaluated fruits and vegetables products as it will affect the sensory quality. The brix value of fresh spinach juice was 6.80 with this decreasing to 3.37-4 after blanching. In addition, the brix value was further reduced in all cases over 7 days storage following storage at 4°C. The losses of the soluble solids during blanching could be attributed to the leaching of some sugar into the water. Furthermore, the heat treatment may lead to changes in the physical properties of tissues and therefore result in losses of total soluble solids (TSS) (Quarcoo et al., 2016). Previous research has indicated similar results in turnip greens with 38-50% of TSS reduced after blanching in hot water for 1-2 min (Martínez et al., 2013). In the current work, °Brix values decreased during storage and this could be explained by the microbial metabolic activities that promote sugars in samples to be converted into organic acids. This conversion would decrease the pH of fresh/blanched spinach juice (Khandpur & Gogate, 2016).

The main purpose of blanching is to destroy endogenous enzymes, which is responsible for food deterioration reactions. As given in Table 5.1, approximately 88.6-98.5% of the POD activity was reduced after thermal blanching for 10-90s. Moreover, there was no significant difference of POD reduction rate after 30s water blanching. Similarly, 90% reduction of POD activity was observed after blanching at 90 °C for 3 min in green beans; 90% of POD inactivation rate ended up in carrots slice when blanching at 70 °C 25 min and 90 °C 1.4 min (Bahçeci et al., 2005; Gonçalves et al., 2010). These observations indicated that inactivation of POD is subjected to blanching temperature and duration time. Therefore, 100 °C 30s as high temperature short time (HTST) blanching condition was identified as the optimum condition in this study to reach a 95.3% reduction of POD activity in spinach leaves. In addition, blanched spinach juice/CRF obtained at this condition (100 °C 30s) was further freeze dried and the physical-chemical properties of these samples were evaluated and discussed in following studies.

 Table 5.1: Physicochemical changes of fresh and blanched spinach juice during

 refrigerated storage

					reduction
Blanching	p	Н	°B	of POD	
time (s)				activity/%	
	Day 0	Day 7	Day 0	Day 7	Day 0
0	$6.84{\pm}0.05^{ab}$	6.51 ± 0.07^{a}	6.80 ± 0.00^{a}	6.07 ± 0.06^{a}	$0.00 \pm 0.00^{\circ}$
10	6.49 ± 0.03^{d}	5.34±0.07 ^c	4.00 ± 0.06^{b}	3.63±0.12 ^b	88.6±3.9 ^b
30	$6.57{\pm}0.03^{cd}$	5.52 ± 0.16^{bc}	3.70 ± 0.30^{bc}	3.43±0.06 ^c	95.3±1.5 ^a
60	6.71 ± 0.06^{bc}	5.81 ± 0.16^{b}	3.53±0.06 ^c	3.13 ± 0.06^d	95.3±0. 8 ^a
90	6.96 ± 0.07^{a}	5.76±0.03 ^b	3.37±0.06 ^c	2.73 ± 0.06^{e}	98.5±0.0 ^a

Figure 5.2: The effect of blanching time on visualized colour difference of spinach juice during storage (4°C 7 days). Spinach juice was obtained after blanching leaves at 100°C for 0-90s.


5.3.2. Morphology of spinach samples

5.3.2.1. Transmission electron microscope (TEM) images of fresh/blanched spinach leaves

TEM was used to determine the ultrastructure changes of spinach leaves before and after the blanching process (100°C 30s) (Figure 5.3).

In blanched spinach leaves there was an obvious collapse of cells as shown in Figure 5.3 d, f and h as compared to fresh spinach leaves (Figure 5.3 c, e and g). In fresh leaves, the intact structures of chloroplasts could be clearly seen, with smooth and defined cell walls, thylakoids, with assembled stacks of grana and plastoglobuli. Chloroplasts in fresh leaves maintained their integrity with some lens shaped chloroplasts (3-10µm in diameter and 1-3µm thick), around the edges of cell, or slightly twisted bands at the cell edges (Burrows et al., 2020). Following blanching, it can be clearly seen that the integrity of the chloroplast fine ultra-structures was largely destroyed, the thylakoids were collapsed and the plastids were dispersed into the intercellular space leaving cell contents indistinguishable and disorganised. Loss of

structural integrity accounts for the low ability of the water blanched samples to retain water during rehydration (Okpala et al., 2014).



Figure 5.3: TEM images of fresh/blanched spinach leaves at different magnifications a) Fresh spinach leaves, b) blanched spinach leaves (100 °C 30s), c, e, g) were fresh

leaves at magnification of $800\times$, $8200\times 27000\times$; d, f, h) were blanched leaves at magnification of $800\times$, $8200\times 27000\times$.

5.3.2.2. Microscopic images of fresh/blanched spinach juice/CRF powders

Figure 5.4 shows the comparison of freeze-dried fresh and blanched (100 °C 30s) spinach samples after grinding and sieving. It can be seen from the image that the samples showed different colour intensities. The green colour comes from the chlorophylls contained in chloroplasts, with the dark green colour of FC samples being associated with a higher concentration of chloroplasts as compared with FJ. The bright green colour of fresh spinach samples turned to olive-green or olive yellow in blanched samples which was ultimately considered as a loss of quality. This loss of greenness was mainly due to chlorophyll degradation or diffusion during blanching treatment. In general, we expect that the chlorophylls were converted into their corresponding pheophytins by losing the magnesium ion in the porphyrin ring when heated (Minguez-Mosquera et al., 1989).

The light microscopic images are present in Figure 5.5 and reveal the microstructural changes in spinach chloroplasts following blanching. All the dried powdered juice/CRF samples were hydrated in RO water before loading for observation, and then analysed using a light microscope with the polarised lens. Therefore, the colour differences between fresh and blanched samples under light microscope was not obvious as shown in Figure 5.4. Comparing with FJ and FC, blanched chloroplasts (BJ and BC) tend to form clumps due to protein denaturation and/or aggregation that occurs

during blanching. Both fresh and blanched CRF showed a higher intensity of chloroplasts as compared with dried juice. The integrity of chloroplasts may vary in all samples as showed in Figure 5.5, and blanched samples appear to contain the most damaged structures.



Figure 5.4: Freeze dried fresh/blanched spinach chloroplasts. All samples were frozen at -80 °C overnight and freeze-dried for 7 days. Samples were grinded and passed through 250 μm sieve. FJ Fresh spinach juice; FC Fresh spinach CRF; BJ Blanched spinach juice; BC Blanched spinach CRF.



Figure 5.5: Light microscope images at X400 total magnification of freeze-dried fresh/blanched spinach juice and CRFs. The dried powders were diluted (X100) with ultra-pure water before imaging. FJ Fresh spinach juice; FC Fresh spinach CRF; BJ Blanched spinach juice; BC Blanched spinach CRF.

5.3.2.3. Scanning electron microscope (SEM) images of fresh/blanched spinach juice/CRF powders

The impact of blanching on the morphology of freeze-dried spinach tissues is given in Figure 5.6. All the samples showed typical morphology of freeze-dried powders with irregular flake shaped particles with porous structure (Chranioti et al., 2016). Compared with unblanched samples, BJ exhibited a smoother surface than FJ, while BC showed increased porosity than FC. Blanching destroyed the cell wall integrity (Figure 5.3), and the structural changes may result in increased porosity.



Figure 5.6: SEM images of fresh/blanched spinach juice and CRF samples at X500 and X2000 magnification. FJ Fresh spinach juice; FC Fresh spinach CRF; BJ Blanched spinach juice; BC Blanched spinach CRF.

5.3.3.Macronutrients of fresh/blanched spinach juice/CRF powders

The macronutrient composition of fresh/blanched (100 °C for 30s) spinach chloroplasts powders are documented in Table 5.2. Proteins made up the majority of the spinach CRF samples (FC and BC), whereas estimated carbohydrates made up the majority of the spinach juice samples (FJ and BJ). Significantly decreased proteins (11.9%) and lipids (8.6%) were observed in blanched spinach juice compared with fresh juice; while only significantly reduced proportion of lipids (9%) were found in CRF isolated from blanched spinach leaves (BC). Generally, blanching reduces the nutritional quality of vegetables by causing the leaching or diffusion of important nutrients like minerals (Mukherjee et al., 2007). During blanching the protein-lipids complex are destroyed and protein denaturation occurs. Therefore, a higher proportion of soluble proteins and lipids leached out from plant tissue into the blanching water. Moreover, the loss of lipids could be also due to the loss of membrane lipids such as galactolipids and phospholipids, and some lipophilic nutrients caused by high temperature treatments (Murcia et al., 1999). In addition, the relatively stable protein concentration remained in BC was due to the physical separation that the denatured proteins were all collected in pellet after centrifugation of juice, and lipids phase no longer bond to proteins were discarded in supernatant. The ash (minerals) is not destroyed by light, heat or oxygen and thus being relatively stable maintained during blanching (Bernhardt et al., 2006). In general, blanching may decrease the nutritional

quality of spinach juice/CRF, which is linked to the cellular structure modification and destruction of nutrients at the higher temperatures used during blanching (Philippon, 1984)

%DW	FJ	FC	BJ	BC
Proteins	34.2 ± 0.9^{b}	48.3±0.7 ^a	22.3±0.2 ^c	49.1±1.1 ^a
Lipids	11.3±0.8 ^c	27.9±1.3 ^a	$2.7{\pm}0.4^{d}$	18.9±1.3 ^b
Ash	16.8 ± 0.9^{b}	7.4±0.4 ^c	22.8±0.4 ^a	7.5±0.2 ^c
Carbohydrates	37.7	16.4	52.2	24.5

 Table 5.2: Nutrients composition of fresh/blanched spinach chloroplasts

Data was analysed by one-way ANOVA according to Tukey's test (p<0.05) to access the differences of the samples. Means (n=3) designated with different letters in the same row are significantly different (p<0.05). *Abbreviations: FJ Fresh dried juice; FC Fresh dried CRF; BJ Blanched dried juice; BC Blanched dried CRF.*

5.3.4.Micronutrients of fresh/blanched spinach juice/CRF powders

5.3.4.1. Fatty acids composition of fresh/blanched spinach juice/CRF powders

Lipids are most elemental nutrients for humans and fatty acids are building blocks of lipids. The composition of fatty acids (FAs) of fresh/blanched spinach juice/CRF is shown in Figure 5.7 and expressed as mg/g DW and mg/g total lipids. As can be seen, the primary fatty acids identified in all samples were palmitic acid (C16:0), palmitoleic acid (C16:1), hexadecatrienoic acid (C16:3), stearic acid (C18:0), oleic acid (C18:1), linoleic acid (C18:2) and α -linolenic acid (C18:3). Among which α -linolenic acid (ALA) is the most predominant fatty acid, followed by palmitic acid and linoleic acid. It was established that on a dry mass basis blanching significantly decreased fatty acid levels in both spinach juice and CRFs; the loss of fatty acids was more intense in dried spinach juice compared with CRFs. The levels of the detected FAs in FJ were in the range of 0.19-10.52 mg/g DW and 1mg/g DW in BJ. As for CRFs, the main loss of fatty acid is ALA which showed a 55% decrease than in fresh CRFs on a dry mass and lipid mass basis. The rapid loss of FAs could be attributed to the loss of total lipids, especially membrane lipids, lost during the blanching process. Spinach juice suffered major losses of total lipids as compared to CRFs. On a lipid mass basis, we detected an increase in C16:0, C18:0, C18:1 and C18:2 FA types after blanching, which is probably due to the increased extraction efficiency of FAs in blanched CRFs.

The ratio of unsaturated to saturated fatty acids (UFA/SFA) is considered as an oil degradation indicator (Romano et al., 2012). The ratio of UFA/SFA in fresh juice and CRF was 2.3 and 3.0, which was decreased to 1.3 and 2.1 after blanching. This reduction suggests a decline in the nutritive value of acyl fatty acids of the chloroplast envelope membrane in blanched spinach leaves, which is mainly ALA (C18:3) and palmitic acid (C16:0). Similar results were also found in canned broccoli were the UFA/SFA ratio was reduced to 3.5 from 4.2 (Murcia et al., 1999). Furthermore, linoleic acid is a n-6 polyunsaturated fatty acid (PUFA) and α -linolenic acid is a n-3 PUFA; they are considered as two plant derived essential fatty acids in diet as they cannot be synthesised by human body (Blondeau et al., 2015). The intake of increased level of

ratio of n-3/n-6 fatty acid is associated with lower risk of cancer and chronic diseases (Goodstine et al., 2003; Wijendran et al., 2004). The ratio of n-3/n-6 in fresh spinach juice/CRFs were 5.0 and 6.3, and this was reduced to 2.3 and 3.0 after blanching treatment. Consequently, blanching spinach leaves may cause the negative effect on fatty acid composition and their functional properties.





Figure 5.7: Fatty acid composition of fresh/blanched spinach juice and CRF. **a**) mg/g DW **b**) mg/g lipid. Results are expressed as means \pm SD (n=3). Data was analysed by one-way ANOVA according to Tukey's test (p<0.05) to access the differences of the samples. *Abbreviations: FJ Fresh dried juice; FC Fresh dried CRF; BJ Blanched dried dried juice; BC Blanched dried CRF.*

5.3.4.2. Chlorophyll content of fresh/blanched spinach juice/CRF powders

Colour is considered an important property in the food industry. Chlorophylls are the most abundant pigments in plant tissues, and is heat sensitive (Ghosh et al., 2022). Chlorophylls are an indicator for chloroplasts content and food quality, especially in food items that have been thermally processed. The content of chlorophyll a, b and total chlorophylls in fresh/blanched dried spinach juice and CRF is shown in Figure 5.8 and expressed in mg per g of spinach chloroplasts (DW) and mg per g of total lipids extracted from spinach chloroplasts. The higher chlorophylls content was observed in fresh juice and CRF samples (FJ and FC) compared with blanched samples, whereas chlorophyll a was higher than chlorophyll b in all samples. The total chlorophylls content decreased after blanching (100 °C at 30s) both in juice and CRF samples. Approximately 93% and 59% of the total chlorophylls were lost in BJ and BC samples on a dry weight basis respectively. In contrast 68.5% and 44% loss on a lipid biomass basis. Researchers have reported that the quality changes of green plants occurring during thermal processing could be attributed to the conversion of chlorophyll a and b to their respective pheophytins, and this was associated with the transition in colour loss from bright green to olive brown (Steet et al., 1996; Teng et al., 1999). The loss of total chlorophylls could be also linked to the loss of total lipids since they are chlorophylls are lipophilic components. The higher loss of chlorophylls and total lipids detected in blanched juice than CRF supports this observation. Previous research has reported that chlorophyll degradation depends on blanching conditions (heating temperature, time and pH) (Van Loey et al., 1998; Rudra et al., 2008). In order to minimise the green colour loss and produce high quality vegetable products, high temperature short time (HTST) blanching combined with pH adjustment is recommended (Belitz et al., 2008). Alkaline pH adjusted with the addition of MgCO₃ improves chlorophyll retention in spinach puree when combined with HTST processing (Gupte et al., 1964).



Figure 5.8: Chlorophylls content of fresh/blanched spinach juice and CRF. **a)** mg/g DW **b)** mg/g lipid. Results are expressed as means \pm SD (n=3). Data was analysed by one-way ANOVA according to Tukey's test (p<0.05) to access the differences of the samples. *Abbreviations: FJ Fresh dried juice; FC Fresh dried CRF; BJ Blanched dried dried juice; BC Blanched dried CRF. Chl.A Chlorophyll a; Chl.B Chlorophyll b; Total Chl. Total Chlorophylls (a+b)*

5.3.4.3. Carotenoids and vitamin E content of fresh/blanched spinach juice/CRF powders

Carotenoids are essential pigments present in fruits and vegetables occurring alongside chlorophylls, that are responsible yellow to red pigmentation (Maoka, 2020). The carotenoid content in fresh/blanched spinach juice and CRF is shown in Figure 5.9, and data are expressed as mg/g DW and mg/g of total lipids. β -carotene and lutein are the predominant carotenoids in spinach, and both occur at greater concentrations in fresh juice/CRF samples than in the blanched samples. Approximately 92% and 69% of the total carotenoid content (beta-carotene and lutein) were decreased in blanched juice and CRF samples on a dry mass basis, whereas 68% and 58% loss on lipid biomass basis. Carotenoid pigments are sensitive to high temperature during blanching, and this can cause isomerisation of carotenoids from trans-isomer to cis-isomers (Lu et al., 2020). In the current work, carotenoid loss was possibly attributed to leaching out into water during blanching and/or degradation during heat treatment. Similar losses are reported in the blanched leaves of both Hypochaeris and Hyoseris, in which the blanching water had a significantly higher (p < 0.05) carotenoid content than the blanched leaves (Sicari et al., 2020). Moreover, the behaviour of carotenoids during thermal processing depends on the food matrix, since evidence shows that some carotenoids like those in carrot are more stable than that of spinach and broccoli (Kopas-Lane et al., 1995; Zhang et al., 2004; Lešková et al., 2006). This should be taken into account that carotenoid stability varies in different types of vegetables due to their different physical state and location in food matrix (Schieber et al., 2005).

Vitamin E was also assessed in the current work. Vitamin E consists of four tocopherols (α -, β -, γ -, and δ -) and the corresponding tocotrienols (α -, β -, γ -, and δ -) which contain unsaturated side chains. α -Tocopherol is considered one of the most important biological antioxidants in cell membranes (Seppanen et al., 2010). Lower levels of α -tocopherol were observed in this work with diminished levels occurring in blanched samples (0.04-0.57 mg/g) compared to fresh ones (0.36-0.89 mg/g) on dry mass basis. As highlighted above, the loss of α -tocopherol could possibility be due to leaching. α -tocopherol seems retained stable in blanched CRF (BC) when normalized to lipid biomass content. The level of α -tocopherol in raw and heat processed vegetables varies in different cases. As reported, higher vitamin E levels are present in boiled broccoli, compared to raw products (Chun et al., 2006).





Figure 5.9: Carotenoids and α -tocopherol content of fresh/blanched spinach juice and CRF. **a**) mg/g DW **b**) mg/g lipid. Results are expressed as means \pm SD (n=3). Data was analysed by one-way ANOVA according to Tukey's test (p<0.05) to access the differences of the samples. *Abbreviations: FJ Fresh dried juice; FC Fresh dried CRF; BJ Blanched dried juice; BC Blanched dried CRF.* β -car. β -carotene; Lut. Lutein; α -Toc. α -tocopherol.

5.3.4.4. Galactolipids and free fatty acid content of fresh/blanched spinach juice/CRF powders

The galactolipids are predominant lipids in the thylakoid membrane of plants, and characterised as monogalactosyldiacylglycerol (MGDG) and digalactosyldiacylglycerol (DGDG) compounds that represent up to 80% of the total lipids in some plants (Block et al., 1983). MGDG and DGDG play major roles in determining the physicochemical properties of the thylakoid membrane, and is responsible for the photosynthetic electron transfer reactions, and stabilization of the photosystem protein complexes in chloroplasts (Kobayashi, 2016). It appears that maintaining a constant MGDG/DGDG ratio in thylakoid membranes is crucial for the stability and functional integrity of photosynthetic membranes (Rocha et al., 2018). The impact of blanching on galactolipids content and free fatty acid release in dried spinach juice/CRF are shown in Figure 5.10 and Table 5.3. The results are expressed as mg/g of total lipids extracted from dried spinach juice/CRF samples. On average, 0.1g of dried fresh samples contains lipids around 0.0122-0.0297 g, as for blanched samples, lipids content was dropped to 0.0026-0.0185 g (Table 7.3). The blanched spinach juice and CRF (BJ and BC) showed higher MGDG content than DGDG with a ratio of 1.3-2, which is in agreement with a higher proportion of MGDG (50%) than DGDG (30%) in thylakoid membranes, especially in the inner envelope membrane (Seiwert et al., 2017). While the galactolipids in FC showed a different trend, the ratio of MGDG/DGDG was 0.6:1, with higher DGDG being found than MGDG. It seems that MGDG could be degraded by endogenous galactolipase more rapidly than that of DGDG. This has been observed in other reports, in that MGDG disappeared faster than DGDG in barley leaves and tomato leaves (Michalski et al., 1980; Gut et al., 1988). It could be confirmed by the significant high released of free fatty acid (FFA) like α linolenic acid (ALA) in FC, with the value of 96.91 mg/g lipid. While low levels of FFA (ALA) were found in FJ, BJ and BC with the values ranging from 0-0.02 mg/g lipid. The low amounts of FFA in blanched samples suggests that spinach leaves avoid lipolysis with inactivated endogenous enzymes. This observation is supported by previous reports, that spinach leaves blanched at 100 °C for 3 min prior to storage at

37 °C was able to abolish galactolipase and sulfolipase activities (Hirayama et al., 1969). In contrast, greater amount of ALA occurred in FC as compared with FJ confirmed significant lipolysis occurring during CRF preparation. The fresh spinach CRF were collected by centrifugation of the spinach juice for 10 min at 4°C, (10,000 rpm for 3 times), prior to being frozen at -80°C and subsequently freeze dried. Whereas spinach juice was directly frozen at -80°C before freeze drying. Consequently, this possibly explains the high level of ALA accumulated in FC could originate from the hydrolysis started by activated/liberated endogenous enzymes during fresh CRF preparation (Kaniuga, 2008). As can be seen from Figure 5.10 (a), bands for acyl-MGDG and MGMG with higher intensity was observed in FC. It is known that MGDG can be acylated with a third fatty acid to the galactose head group form acyl-MGDG in plant leaf homogenates (Persson et al., 2000; Nilsson et al., 2015). This enzymatic fatty acid exchange was inactivated by blanching spinach leaves in this study. Moreover, the galactolipids from non-heated samples were more easily to hydrolyse by endogenous enzymes compared with heated samples, thus the MGDG was hydrolysed to MGMG and release FFA in FC samples. Similar results have been in fresh/steam sterilised pea vine CRF, that 3-fold lower FFA and 1.8-3.4 fold higher of MGDG and DGDG levels found in steam sterilised samples (Wattanakul et al., 2022).

In addition, significantly reduced galactolipids content was observed in BJ (44.07 mg/g of MGDG and 22.32 mg/g DGDG) compared with FJ (77.26 mg/g of MGDG and 75.64 mg/g DGDG). This decrease of galactolipids in dried spinach juice could be explained by the loss of total lipids during blanching and possible interaction of the

sugar moiety in galactolipids with blanching water (Cho et al., 2001).

In general, blanching spinach leaves prior to juicing can inactivate endogenous enzymes activity, prevent galactolipids enzymatic change and degradation. While the blanching treatment can also cause significant loss of galactolipids in fresh spinach samples by leaching into hot water.



Figure 5.10: HPTLC analysis of lipid extracts from freeze-dried fresh/blanched spinach juice and CRF. Polar lipid separation was visualised in a) and non-polar lipid separation was visualised in b). As reference standards, pure MGDG, DGDG and α -

linolenic acid (ALA) were used. Abbreviations: FJ Fresh dried juice; FC Fresh dried CRF; BJ Blanched dried juice; BC Blanched dried CRF.

mg/g Lipid	MCDC	DCDC		Dried lipid
	MGDG	DGDG	FFA(ALA)	(mg/g DW)
FJ	77.26±3.30 ^b	75.64±5.82 ^a	0.02 ± 0.00^{b}	122±7°
FC	42.33±1.83 ^c	76.01±5.36 ^a	96.91±7.02 ^a	297±18 ^a
BJ	44.07±7.49 ^c	22.32±5.04 ^b	0.00 ± 0.00^{b}	26 ± 6^d
BC	91.25±3.12 ^a	67.77±3.35 ^a	0.00 ± 0.00^{b}	185±3 ^b

Table 5.3: Galactolipids and free fatty acid contents in freeze-dried fresh/blanched spinach juice and CRF.

The concentration of galactolipids were expressed as mg/g of lipid extracted from dried samples. The dried lipid weight extracted from freeze-dried spinach juice/CRF is shown in the table and expressed as mg/g DW. Data was analysed by one-way ANOVA according to Tukey's test (p < 0.05) to access the differences of the samples. Means (n=3) designated with different letters in the same row are significantly different (p < 0.05). Abbreviations: FJ Fresh dried juice; FC Fresh dried CRF; BJ Blanched dried juice; BC Blanched dried CRF.

5.3.5.Physical properties of fresh/blanched spinach juice/CRF powders

Table 5.4 provides some physical properties of freeze dried fresh/blanched spinach juice and CRF samples. Freeze drying was found to efficiently reduce the moisture

content and water activity in food products, decreases the perishability, improves value and enhances the shelf-life (Agoreyo et al., 2011). It is noted that all samples showed a low moisture content and water activity with the value in range of 1.71-8.05% and 0.11-0.21 respectively. The recommended moisture content and water activity for food powder products is below 10% and 0.3, our results suggest a guaranteed microbiological safety of spinach samples during storage (Tze et al., 2012b).

The solubility is defined as the maximum amounts of substance dissolved in a given amount of water at specific temperature, it plays a key role in manufacturing food powder products as ingredient for consumers with desired functionalities (Fang et al., 2007). Higher solubility was observed in juice samples (75.70-82.84%) as compared with CRFs (11.32-30.94%). The juice is more soluble than CRF, as the CRF preparation steps was to collect the pellet by centrifuging juice and discard the supernatant, and thus contains higher lipids and proteins.

In addition, blanching pretreatment seems to increase the solubility in juice samples and decrease in CRFs. Solubility might be affected by several factors, such as particle microstructure, particle size and physical-chemical properties of substances (Anandharamakrishnan et al., 2010; Kaptay, 2012). In this study, the higher solubility of dried juice samples was observed compared with CRF samples. Moreover, the blanching process of spinach juice seemed to have increased the solubility of dried juice samples. This is associated with the soluble ash and carbohydrates ratio presented in samples, which are highest in BJ, followed by FJ, BC and FC (data shown in Table 5.2). As for dried CRFs, significantly reduced solubility may possibility be affected by protein denaturation, and/or aggregation or other structural changes along with increased particle size during blanching and high lipid percentage (Table 5.2), (Borremans et al., 2020). Similarly, water blanching and steam blanching led to significantly reduced solubility in dried cauliflower leaf compared with unblanched samples (Mythili et al., 2021). As reported by Baldelli et al. (2022), powders with high-fat content showed large agglomerates, furthermore, an increase in 10% of fat percentage increased 60 µm of the mean diameter. The relationship between particle size and solubility is that solubility increased with decreased particle size due to enlarged surface area (Sun et al., 2012).

The dispersibility of food powder is dependent on its ability to disperse in solution, this index is important in industrial settings (Sharma et al., 2012). The dispersibility of fresh spinach samples was 85.08-82.14%, which reduced to 64.18-79.39% in blanched samples, all the samples showed a good dispersibility within the range of 67.05–99.98% which is ideal dispersibility for food powders (Jaya et al., 2004). Similarly, dispersibility also related to particle size and the presence of fat, increased particle size and free fat content in the powder further reduces the dispersibility. In this consequence, protein denaturation and/or aggregation or other structural changes during blanching leads to increased particle size in blanched samples and thus reduce the dispersibility (particle size data were not shown). Moreover, lower dispersibility in blanched samples could be linked to their fat percentage (date shown in Table 5.2).

Properties	FJ	FC	BJ	BC
Moisture content/%	8.05±0.65ª	3.87±1.37 ^b	6.28±0.38 ^a	1.71±0.37 ^c
Water activity	0.12±0.01 ^c	0.18 ± 0.00^{b}	0.11±0.00 ^c	0.21±0.00 ^a
Solubility/%	75.70 ± 1.18^{b}	30.94±0.74 ^c	82.84±17.6 ^a	11.32 ± 0.02^{d}
Dispersibility/%	85.08±0.29 ^a	79.39±3.16 ^b	82.14±2.05 ^{ab}	64.18±0.62 ^c

Table 5.4: Physical properties of freeze dried fresh/blanched spinach juice and CRF

Data was analysed by one-way ANOVA according to Tukey's test (p<0.05) to access the differences of the samples. Means (n=3) designated with different letters in the same row are significantly different (p<0.05). *Abbreviations: FJ Fresh dried juice; FC Fresh dried CRF; BJ Blanched dried juice; BC Blanched dried CRF.*

5.4. Conclusion

A combination of blanching and freezing has been established as the most suitable preservation method for vegetables, therefore, blanching spinach leaves prior to freeze drying was discussed in this chapter in order to make the spinach chloroplast products with minimum quality loss and enhanced nutrients stability.

Overall, blanching spinach leaves (100 °C 30s) before juicing can efficiently inactivate the endogenous enzymes (POD), preventing hydrolysis of galactolipids (MGDG and DGDG) and release of FFA in spinach juice. The blanched juice was then freeze dried to obtain the powdered spinach juice/CRF with enhanced shelf stability. However, significant quality loss was observed in both blanched spinach juice and CRF powders. Blanching is typically an open system, meaning that there is significant loss of mass or volume as water-soluble compounds, nutrients, and other components leach out into the blanching water. The visual green colour loss and nutrients loss was mostly due to the leaching of water soluble nutritional components to the blanching water and the degradation of thermal sensitive compounds. The total mass/volume losses of blanched spinach leaves/juice compared with control were measured, which was around 7% and 14% respectively (Appendix Table 5.1). Also, the yield of blanched juice was significantly reduced (Appendix Table 5.2). Given this fact, blanched spinach juice/CRF will be not considered for further investigation during *in vitro* static digestion model.

Pasteurisation (chapter 4) primarily focuses on preserving product quality in a closed system, while blanching involves both cooking the spinach leaves and leaching compounds into an open system, leading to quality changes in both the final product and the blanching water. The nutrients loss of freeze-dried spinach juice and CRF after pasteurisation and blanching is shown in Appendix Table 5.3. The foremost reduction in macronutrient content subsequent to thermal treatment occurs in total lipids. Specifically, pasteurized juice exhibits a 3-9% decrease, whereas blanched juice experiences a substantial 76% reduction. Pasteurised CRF displays a loss ranging from 28-36%, while blanched CRF records a reduction of 32%. The micronutrients reduction in terms of total chlorophylls, carotenoids, VE and ALA was also found, and significantly lost obeseved in blanched juice. This study provides impact of technical difference between pasteurisation and blanching on the nutrients loss of spinach juice/CRF products.

Chapter VI

6. *In vitro* digestion of galactolipids and proteins from spinach juice/CRF

6.1. Introduction

Galactolipids are the most abundant lipids on earth since they make up the majority of membranes in chloroplasts which are ubiquitous in the biosphere as key organelles in plants and algae (Hölzl et al., 2006). The lipid bilayer of the thylakoid membrane in plant chloroplasts and cyanobacterial cells is predominantly composed of four unique lipid classes; Monogalactosyl diacylglycerol (MGDG) and digalactosyl diacylglycerol (DGDG) containing one or two saturated and/or unsaturated fatty acids linked to the glycerol moiety, and account for 70-80% of the total lipids in plant cells (Figure 2.4) (Christensen, 2009). Trigalactosyldiacylglycerol (TGDG) and sulfoquinovosyldiacylglycerol (SQDG) present in lower proportion, a sulfolipid with a sulfoquinovose residue instead of galactose (Figure 2.4) (Amara et al., 2010). MGDG and DGDG are two main galactolipids that essential for facilitating the photosynthesis light reaction in the thylakoid membrane and maintaining chloroplast morphology and for plant survival under abiotic stresses such as phosphate starvation and freezing (Li et al., 2018).

The most abundant and commonly reported galactolipid in plants comprise of large amounts of α -linolenic acid (ALA; 18:3 n-3), which, on a molar basis, represents about 60%-95% of the total fatty acids in some higher plant galactolipids, and hexadecatrienoic acid (16:3 n-3) accounts for 8.5 - 19.2% of total fatty acids found in

MGDG (Whitaker, 1986). However, the digestion of galactolipids in humans has been overlooked since green leaf tissue only contributes a small portion of the total acylglycerolipid (such as triacylglycerol in oils and fats) in the human diet (Sahaka, 2020). Studies addressing the digestion of galactolipids in humans are important, as it is a good source of omega-3 fatty acids and the dismantling of the main lipid in thylakoid membranes should promote the release of these nutrients *viz*. carotenoids and tocopherol.

The digestion of galactolipids are associated with lipolytic enzymes such as pancreatic lipase-related protein 2 (PLRP2) together with pancreatic carboxyl ester hydrolase/bile salt stimulate lipase (CEH/BSSL) (N'Goma, 2012; Sahaka, 2020) . PLRP2 has been identified as an important factor of galactolipid digestion in humans, and also found in monogastric herbivores such as guinea pig, rat, rabbit and sheep (Caro et al., 2008). Both human and animal PLRP2 can hydrolyse MGDGs and DGDGs from spinach leaves to FFAs, MGMGs, DGMGs and water-soluble galactose-containing compounds (Andersson et al., 1995). In addition, the presence of bile salt micelles and colipase with human PLRP2 can hydrolyse long-chain tri-, di-, and monoglycerides (Xiao, 2011). Together, CEH/BSSL and PLRP2 had a synergistic effect, increasing cellular uptake and re-esterification 4-fold compared with the sum of each lipase alone (Andersson, 2011). Human pancreatic juice (HPJ) has been identified as the source of these enzymes (PLRP2 and CEH/BSSL), however its use is restricted due to ethical issues. An alternative to HPJ, has been in the use of porcine pancreatic extracts (PPE). While, PPE has lower galactolipase and cholesterol esterase activities than HPJ, which

indicates a lower level of PLRP2 and CEH/BSSL in this enzyme preparation that can not be determined by mass spectrometry (Salhi et al., 2020).

In addition, lipid digestion starts in the stomach by the action of gastric lipase which contributes about 10 - 25% of the overall total triacylglycerides (TAGs) digested. Interestingly gastric lipase remains active in intestine for a period of time and therefore serves to aid liberation of FFA in the GI tract (Thomas et al., 2012). Fatty acids liberated by human gastric lipase can trigger the subsequent activity of pancreatic lipase on lipid substrates (Brodkorb et al., 2019). Therefore, addition of gastric lipase is recommended in gastric phase during in vitro digestion model by INFOGEST 2.0 protocol. Rabbit gastric extracts (RGE) which contain both gastric lipase and pepsin activity is now commercially available, and in combination with PPE appears to be a good substitute for human gastric and pancreatic lipases (Capolino et al., 2011). There is hardly any published work on the digestion of galactolipids in chloroplasts under conditions that mimic the human gastrointestinal conditions (Wattanakul et al., 2019). Wattanakul and colleagues showed for the first time that human simulated digestion of galactolipids in spinach chloroplast obtained by different food processing protocols. The refinement of the INFOGEST in vitro digestion model to include RGE was also employed. The rate of lipolysis and the composition of the lipolysis products were assessed as a function of time during digestion after lipid extraction and analysis by high performance thinlayer chromatography (HPTLC).

6.2. In vitro digestion model

In vitro digestion model



Oral phase:

Food:SSF (1:1), pH 7, 37°C, 2min

Gastric phase: Oral phase:SGF (1:1), pH 5, 37°C, 2h Digestive enzymes: RGE + Pepsin (Lipase activity 20.4U/ml; Pepsin activity 2000U/ml)

Intestinal phase: Gastric phase:SIF (1:1), pH 6.5, 37°C, 2h Digestive enzymes cocktail: PPE (Lipase activity 2000U/ml, trypsin activity>100U/ml; porcine bile (10mM)

Sampling 1ml at end of each stage and labelled as G0, G and I respectively. Then extract lipid from these samplings for further analysis of HPTLC.

Figure 6.1: *In vitro* digestion model with modifications based on INFOGEST 2.0 (Brodkorb et al., 2019). Simulated Salivary Fluids (SSF). Simulated Gastric Fluids (SGF). Simulated Intestinal Fluids (SIF). Rabbit Gastric Extracts (RGE). Porcine Pancreatic Extracts (PPE).



6.3. Sample preparation summary

-80 °C overnight before freeze drying

Figure 6.2: Spinach juice/CRF samples preparation steps. (1) Spinach juice was prepared with a twin-gear juicer and held at room temperature before further processing, making 3L spinach juice usually takes 2h at room temperature (RT) and each sample was prepared on a different day but from the same batch of leaves. (2) Fresh juice was kept in an ice bucket (0°C) while spray drying, the whole drying process took roughly 10h. (3) Fresh juice was directly frozen at -80°C overnight before freeze drying. (4) Fresh juice was centrifuged before freezing at -80°C overnight. (5) Spinach juice was pasteurised at mild and intense conditions (70°C 15s and 90 °C 5 min), while the fresh juice was kept in cold room (4 °C 24h) before pasteurisation. (6) Heated juice was directly frozen at -80°C overnight before freeze drying. The temperature was directly frozen at -80°C overnight before freeze drying. (7) Heated CRF were collected by centrifuging and then kept in -80 °C freezer overnight before freeze drying. Samples include spray dried juice (SJ), freeze dried juice (FJ), freeze dried CRF (FC),

mild pasteurised freeze-dried juice (MJ), mild pasteurised freeze-dried CRF (MC), intense pasteurised freeze-dried juice (IJ), intense pasteurised freeze-dried CRF (IC).

6.4. Lipid extraction and measurement from spinach juice/CRF before and during *in vitro* digestion

The lipids and protein concentration from initial dried spinach juice/CRF powders is shown in Table 6.1. The lipid and protein content of fresh freeze-dried and spraydried spinach juice are similar. All samples analysed and used in these digestion studies, apart from SJ, were freeze-dried, so we assume a similar degree of lipid extraction in each one to allow comparison across the samples. On comparing the FJ and FC lipid % figures it appears that FC contains approximately 2.5x more total lipid per unit dry weight than FJ; this is to be expected as the centrifugation step concentrates chloroplasts, the major store of lipids in a spinach plant cell and separates them from carbohydrates.

Further interrogation of the data in Table 6.1 reveals no significant impact of heating juice on the lipid and protein concentration as expressed % DW of powder derived from freeze-dried juice. However, in CRF samples, a significant increase of proteins and decrease of lipids (% DW) was observed after pasteurisation. This is likely due to the high heat cause the break of lipid-proteins complex in pasteurised spinach juice, while after centrifugation, the protein-rich pasteurised CRFs were collected, the lipids released in the supernatant were discarded.

Table 6.1: Lipid and protein concentration from spinach chloroplast juice/CRF

powders before digestion

/%DW	SJ	FJ	FC	MJ	MC	IJ	IC
Proteins	38.3±0.6 ^e	39.6±0.4 ^{de}	50.6±0.5°	41.2 ± 0.1^{d}	58.6±0.2ª	38.5±0.9 ^e	54.9 ± 1.0^{b}
Lipids	13.0±1.9°	11.8±0.4 ^c	28.3±0.4 ^a	11.5±0.5°	20.3±0.7 ^b	10.7±0.5°	18.1±0.3 ^b

Data was analysed by one-way ANOVA according to Tukey's test (p<0.05) to access the differences of the samples. Means (n=3) designated with different letters in the same row are significantly different (p<0.05). Samples include spray dried juice (SJ), freeze dried juice (FJ), freeze dried CRF (FC), mild pasteurised freeze-dried juice (MJ), mild pasteurised freeze-dried CRF (MC), intense pasteurised freeze-dried juice (IJ), intense pasteurised freeze-dried CRF (IC).

In order to investigate the galactolipids change and free fatty acid release during digestion, we extracted total lipid from 1ml sampling at end of oral phase (G0), gastric phase (G) and intestinal phase (I).

The quantification of galactolipids and free fatty acid in this study was based on the lipid extracted from the 1 ml sampling during digestion (Table 6.2). However, the 1ml sampling during digestion includes not only the spinach chloroplasts but also the enzymes and other constituent that can carry lipid into the incubation system. Thus, control experiments were carried out by replacing the spinach chloroplast samples with ultra-pure water and undergoing the same digestion process as other samples. The amount of lipids in the sample derived from the chloroplast-enriched powders was calculated by subtracting the control values from the total lipid extracted from each sample (Table 6.2). Overall, there seems to be little shift in the amount of lipid in the juice samples during digestion, suggesting relatively efficient lipid extraction at each stage. As seen in Table 6.2, the significant reduction of lipid in intestinal phase was observed, which probably due to the sampling issue at the end of intestinal phase; The addition of PPE to the intestinal phase can form clumps and stringy structures which ultimately affect the sampling process and lipid analysis. Moreover, it could also be due to the sampling loss occurs at the end of gastric phase. (There were 20 ml samples left at end of gastric incubation, and 6 ml samples were sampling out for analysis, another 14 ml SIF was added in the digestive tube for intestinal incubation. It is therefore indicate a 30% of total lipid loss in intestinal tube for further sampling).

Lipid weight	SJ	FJ	FC	MJ	MC	IJ	IC
G0(mg/ml)	3.7±0.5 ^b	4.4 ± 0.4^{b}	16.5±1.9 ^a	5.5±0.2 ^b	9.3±1.0 ^b	4.7±0.2 ^a	11.2±1.5 ^a
G (mg/ml)	5.3±0.2 ^a	5.7±0.4 ^a	18.0±0.5 ^a	6.2±0.2 ^a	12.2±1.1 ^a	5.2±0.4 ^a	9.1±0.6 ^a
I (mg/ml)	4.0±0.3 ^b	4.8±0.5 ^{ab}	9.3±1.2 ^b	3.8±0.4 ^c	5.4±1.2 ^c	3.6±0.5 ^b	5.7±0.8 ^b

Table 6.2: Lipid concentration from 1ml sampling during digestion.

Data was analysed by one-way ANOVA according to Tukey's test (p<0.05) to access the differences of the samples. Means (n=3) designated with different letters in the same column are significantly different (p<0.05). Samples include spray dried juice (SJ), freeze dried juice (FJ), freeze dried CRF (FC), mild pasteurised freeze-dried juice (MJ), mild pasteurised freeze-dried CRF (MC), intense pasteurised freeze-dried juice (IJ), intense pasteurised freeze-dried CRF (IC). G0: End of oral phase, G: End of gastric phase, I: End of intestinal phase. The lipid weight from spinach chloroplasts of G0, G and I were calculated by subtracting the lipids weight from control.

Figure 6.3 illustrates the lipids content from initial dry powders (%) and from 1ml sampling during digestion at G0 (mg/ml). All the freeze-dried samples showed a similar trend of lipid concentration, which suggests a constant and reliable extraction efficiency. Spray dried juice showed slightly lower extraction from 1ml sampling at G0 compared with initial powders.



Figure 6.3: The lipid concentration from spinach chloroplast before digestion. Samples include spray dried juice (SJ), freeze dried juice (FJ), freeze dried CRF (FC), mild pasteurised freeze-dried juice (MJ), mild pasteurised freeze-dried CRF (MC), intense pasteurised freeze-dried juice (IJ), intense pasteurised freeze-dried CRF (IC). Initial: dried spinach chloroplast powders (%). G0: 1ml sampling at the end of oral phase (mg/ml).

6.5. Limitations about HPTLC analysis

The interference, troubleshooting for calculation, and sampling/lipid extraction

efficiency was observed throughout the *in vitro* digestion, and this section will mainly discuss the issues in polar lipids.

The interference in HPTLC analysis may occur when some lipids contained in enzymes. In order to subtract the lane/background noise of MGDG and DGDG, we performed control experiments without the spinach chloroplasts, only water with the addition of digestive enzymes (RGE in gastric phase, porcine bile and PPE in intestinal phase) undergoing the same incubation environments. According to the HPTLC analysis of the control plates (Figure 6.4), a yellow faint band around Rf 0.6 can be observed during gastric phase, this was due to the addition of RGE, since it was the only component contained in the gastric phase in control. As the position of the yellow lipid band did not overlap with the bands for any of target galactolipids, this band 'noise' can be ignored. However, many bands showed up in intestinal phase (Rf 0.23, 0.3, 0.4), which may interfere the quantification of some galactolipids. (The band next to the MGDG at Rf 0.5 was too faint therefore it was ignored in this study). The yellow band next to the DGDG at Rf 0.23 was quite wide and strong and would interfere with the DGDG bands. This yellow bands were partially merged with DGDG and MGMG. Due to the addition of the RGE, PPE and bile extracts in the control experiments, we suspect that the interference bands could result from lipids from these preparations. These interference bands were not observed in Wattanakul's work where RGE, PPE and bile salt (sodium taurodeoxycholate, NaTDC) were used in the control experiments (Wattanakul et al., 2019). The only difference was the source of bile, therefore, the interference bands in intestinal phase could be attributed to the porcine bile extract.

Unlike the NaTDC which is the pure bile salt, the porcine bile extract contains glycine and taurine conjugates of hyodeoxycholic acid and other bile salts, so the interference bands could be caused by the lipid contamination from the bile extract. In consequence, the DGDG band may in effect be partly or wholly composed of contaminating material. The identification and quantification of DGDG is based on the peak integration of the target peak areas (as shown in appendix Figure 6.2). The DGDG peaks are contaminated by bile extract and partially merged with MGMG peaks. Thus, the peak area of DGDG will also include the partially merged interference peak area, in this instance, the subtraction of interference from DGDG is unfeasible, thus the quantification of DGDG is ignored in this study.

The sampling and lipid extraction efficiency was checked in all samples during digestion. The expected lipid content in all samples at each phase is calculated and shown (see appendix Table 6.1). When compared the expected lipid amount with the extracted value, approximately 30% of lipid loss is seen in intestinal phase. This is due to the sampling (6 ml) out from total gastric digestion tube (20 ml), thus 30% of total lipid from gastric phase was lost when entering into intestinal phase. According to this, the galactolipids and FFA content during intestinal phase in all samples will be adjusted to account for its original content. The MGDG and FFA can be calculated and expressed as "mg MGDG/g CRF lipid", the CRF lipid extracted from 1 ml sampling during digestion is shown in Table 6.2. Meantime, the sampling issues occur at the end of digestion with the addition of PPE, which allows clumps and stringy structures formation. The sampling was done by using the 1 ml pipette, the jelly-like mixture
samplings may be stuck in the tips and cause the underestimate of total extracted lipid and further effect the calculation of galactolipids data.



Figure 6.4: HPTLC analysis of control samples during *in vitro* digestion. G0: end of oral phase; G120: end of gastric phase; I120: end of intestinal phase. Polar lipids

standards: monogalactosyl diglycerides (MGDG) and digalactosyl diglycerides (DGDG). The bands from this control plates indicate the interference from the addition of RGE and bile extract.

6.6. Results and discussion

As described in Chapter 3 and 4, different processing procedures were used to produce the spinach chloroplast materials. Both spray drying and freeze drying can be applied to produce dried spinach chloroplast powders with high levels of nutrients retained. Thermal pre-treatment (pasteurisation of spinach juice) combined with freeze drying is able to inactivate the endogenous enzymes, especially peroxidase (POD), which can produce spinach chloroplast materials with enhanced stability. In this chapter, we aim to investigate the effect of spinach chloroplast material processing on the digestion of galactolipids in a simulated human digestion model.

6.6.1.Galactolipid digestion trends in different chloroplastenriched powders using band intensity from HPTLC separation of lipids

The total lipid extracted from 1ml sampling at the end of each stage along with the standards were transferred to the silica gel thin layer chromatography (TLC) plate. The high-performance TLC (HPTLC) technique was used for the analysis of both polar and non-polar lipids. In this study, polar lipids include monogalactosyl diacylglycerol (MGDG) and digalactosyl diacylglycerol (DGDG), nonpolar lipids include monoacylglycerol (MAG), 1,2 diacylglycerol (1,2DAG) l, 1,3 diacylglycerol (1,3DAG) and triacylglycerol (TAG) and α -linolenic acid (ALA) were used as reference. Figure

6.6 illustrates all the HPTLC analysis of the galactolipids and FFA release in different treated spinach chloroplasts during *in vitro* digestion.



Figure 6.6: HPTLC analysis of the galactolipids and FFA in spinach chloroplasts during *in vitro* digestion. Samples were incubated in oral phase (2min), gastric phase (2 hours) and intestine phase (2 hours) at 37 °C. Samples were collected at time 0 of gastric phase, G0 (panel A, D), at the end of 2h gastric phase (G) (panel B, E) and at the end of 2h intestinal phase (I) (panel C, F). Panel A-C reveals polar lipids (galactolipids) digestion of spinach chloroplast and control samples during *in vitro* digestion. Panel D-F reveals non-polar lipids (MAG, DAG, TAG and FFA) release of spinach chloroplast and control samples during *in vitro* digestion. Pure monogalactosyldiacylglycerol (MGDG) and digalactosyldiacylglycerol (DGDG) standards show bands at Rf 0.48 and Rf 0.28, respectively. Mixed lipid standard (MAG,

1,2-DAG, 1,3-DAG & TAG) show bands between Rf 0.1-0.9; and α -linolenic acid (ALA) standard show band at Rf 0.6.

According to the HPTLC analysis (Figure 6.6 A, B and C), both the MGDG and DGDG from spinach chloroplast materials were hydrolysed and converted to lysogalactolipid (MGMG and DGMG) throughout the whole digestion. In the absence of the pure MGMG and DGMG standards, the bands relating to these products of galactolipid digestion (Figure 6.6 C) were presumed from the work carried out by Wattanakul and co-workers (Wattanakul et al., 2019), we hypothesise that the bands showed at Rf 0.3 and 0.1 are MGMG and DGMG respectively. The decrease of MGDG was in correlation with the increased band intensity of MGMG.

Qualitatively, the MGDG and DGDG remains relatively stable during gastric phase (Figure 6.6 A, B), along with the weakly released FFA (Figure 6.6 D, E), which confirms that the galactolipid hydrolysis is not predominant in gastric conditions. This finding is in line with the fact that gastric lipase in RGE has no galactolipase activity reported by Sahaka (2020). Nevertheless, gastric lipase shows high activity on TAG at sn-3 position at pH 5, and lower activity on DAG and MAG. Also, the acylated galactosyldiacylglycerols (acyl-MGDG and acyl-DGDG) remain relatively stable during gastric phase in all samples, which is in agreement with the minimal hydrolysed galactolipids. Based on other researchers, acyl-MGDG and acyl-DGDG, which is acylated forms of MGDG and DGDG have also been found by thin layer

chromatography in various plants such as spinach (Vu et al., 2014). The band assignment for acylated galactolipids is based on recent work that separate acylated galactosyldiacylglycerol in spinach with mobile phase chloroform: methanol: water (47.5:12:1.25) using HPTLC which is identical to this study (Sutcharit et al., 2023). Therefore, as a reference we hypothesise that the bands at Rf 0.8 and 0.4 represents the presence of acyl-MGDG and acyl-DGDG.

With the addition of bile extract and PPE in the intestinal phase, the galactolipids (MGDG and DGDG) from all spinach chloroplast materials were significantly hydrolysed to MGMG and DGMG, respectively (Figure 6.6 C). High amounts of FFA were also released (Figure 6.6 F). The hydrolysis of galactolipids (MGDG and DGDG) to their galactose constituents such as MGMG and DGMG will result in the release of FFA; ALA was found the most abundant FFA released from sn-1 position of MGDG and DGDG in spinach leaves (Amara et al., 2010). From the HPTLC plates, the bands of MGDG were almost gone, while DGDG remains throughout intestinal phase in all samples. This result clearly shows that the hydrolysis of MGDG is faster than DGDG during the intestinal phase in the presence of PPE. This is confirmed by Wattanakul et al. (2019), that PPE has a higher affinity to MGDG than DGDG. Similarly, a significant decrease in acyl-MGDG and acyl-DGDG were also observed in all samples during intestinal incubation. The acyl-MGDG was completely hydrolysed while the acyl-DGDG remains. The acylation and deacylation reaction is a reversible process. The PLRP2 can deacylated the acylated galactolipids to galactolipids and release FFA

(Sahaka, 2020).

Due to the co-migration of metabolites and the presence of interference bands (Rf 0.27-0.4) during intestinal incubation as described in section 6.5, the measurement of DGDG, MGMG and acyl-DGDG though band intensity is compromised. The DGMG bands were too faint and was excluded for quantification analysis. Therefore, the MGDG and the main released FFA (ALA) are chosen for the markers for the galactolipids digestion in this study. The intensity of the bands for MAG and TAG were too low to be identified or quantified and the band for 1,2-DAG, 1,3-DAG were partially merged together and may comigrate with each other. But the HPTLC analysis does allow qualitative data worthy of comment.

6.6.2.Quantification of galactolipid digestion and FFA release in different chloroplast-enriched powders using band intensity from HPTLC separation of lipids

6.6.2.1. MGDG and FFA content in the oral phase

The galactolipid (MGDG) and FFA changes from different treated spinach chloroplast materials during *in vitro* digestion was quantified and expressed as mg of MGDG/FFA per gram of chloroplast lipid as shown in Figure 6.7 and 6.8 (time 0, 2 and 4h represents G0, G and I respectively). As discussed in section 6.5, the sampling loss happens during the whole digestion process, the reliability of the sampling method may have been compromised by the presence of large particles in some samples, and this will cause an underestimate of the overall data that MGDG and FFA obtained.

Initially, the amount of MGDG in all samples was in the range of 3.3-96.5 mg/g lipid at G0. Since no digestive enzymes were added at G0 (2 min, 37 °C), hardly any digestion occurred. Thus, we suspect the distinct variation of MGDG and FFA concentrations in spinach juice and CRF samples is mainly caused by its composition difference and the preparation steps (Figure 6.2). The MGDG concentration in FC was significantly lower than FJ despite its higher total lipid content (Table 6.2). This could be attributed to the preparation steps for the spinach materials, spinach juice was directly frozen (-80°C) before drying, whereas the fresh spinach CRF was collected by centrifuging the spinach juice 10 min, 4°C, 10,000 rpm for 3 times, and then frozen at -80°C prior to freeze drying, during which significant lipolysis or acylation of galactolipids seemed to occur, with higher band intensity seen for acyl-MGDG and acyl-DGDG observed in FC (Figure 6.6A). Compared with FJ, SJ showed significantly lower MGDG, since during spray drying the fresh spinach juice (3L) was added to the feed tank gradually which the whole drying process took almost 10h. In this situation, the loss of galactolipids in SJ may arise from the autoxidation reaction started by activated/liberated endogenous enzymes (lipases, lipoxygenases and galactolipase) or the enzymatic transacylation (higher band intensity of acyl-MGDG and acyl-DGDG) (Figure 6.6A) (Duden et al., 1981; Kaniuga, 2008). The relatively low MGDG content at G0 was observed in pasteurised juice (MJ and IJ). This could be also caused by the preparation steps, that the spinach juice was pasteurised 1 day after juicing (kept at 4°C). This gives more time for the endogenous enzymes to act and synthesise acylated form

of galactolipids, and the increased level of acyl-MGDG and acyl-DGDG in pasteurised juice confirmed the enzymatic transacylation. As for the CRF materials, it is worth noticing that MC contains the highest concentration of MGDG (83.6 mg/g lipid) followed by FC (31.6 mg/g lipid) and IC (11.0 mg/g lipid). This is due to the fact that lipid concentration was drastically reduced in pasteurised CRFs (MC and IC) compared with non-pasteurised CRF (FC). As shown in Table 6.2, the lipid concentration from 1 ml sampling at G0 is 16.5 mg in FC and decreased to 9.3-11.2 mg in MC and IC respectively. Moreover, as can be seen in the HPTLC analysis of galactolipids (Figure 6.6A), the MGDG at G0 shows similar band intensity between FC and MC and significantly distinguished in IC. So, when it was calculated by mg/g lipid, the concentration of MGDG in MC was relatively high. One assumes that despite the refrigerated storage overnight, the mild pasteurisation tends to largely retain the GLs levels in the subsequent dried CRF powders and intense pasteurisation seems like decrease their levels. Meanwhile, it is known that pasteurisation process causes the denaturation and aggregation of lipid-protein complex, therefore the pasteurised CRF pellet was able to retain most of the chloroplast membranes with the evidence that the supernatant shows the bright yellow colour. The lipids of the photosynthetic membranes in the chloroplasts are dominated by galactolipids. While the fresh CRF pellet was collected by discarding the green supernatant after centrifugation for 3 times which suggests some loss of photosynthetic membranes which mainly consists of GLs. This could explain the low MGDG in FC powders compared with MC.

As shown in Figure 6.8, high levels of FFA (ALA) can be observed in all cases (G0). The FFA levels in all the initial samples were mainly due to the lipid hydrolysis during preparation with the activate endogenous enzymes. Thermal processing of the green biomass inhibits the activity of lipolytic enzymes and so reduces the release of FFA (Suzuki et al., 2005), and prevent the subsequent fatty acid oxidation and degradation. While, in this study the high accumulation of FFA in pasteurised samples is because of the long storage (24h, 4°C) before heating, during which significantly lipid hydrolysis occurs. This was confirmed in Chapter 5, by blanching spinach leaves prior to juicing, from which step the endogenous enzymes were inactivated immediately and the lipid hydrolysis was prevented in the following process. The FFA(ALA) release between fresh and blanched CRF powders was compared, with the value of 96.91 and 0 mg/g lipid respectively (data shown in Table 5.3), which provides strong evidence that the FFA accumulation can be completely avoided by rapidly heat processing of the biomass.



Figure 6.7: MGDG content in different treated spinach chloroplasts during *in vitro* digestion. Time 0 represents G0, the end of 2 min oral phase, without adding any enzyme. In theory, no digestion happens yet at this time point, therefore, it can be seen as undigested CRFs; Time 2 h represents G, end of 2 h gastric incubation; Time 4h represents I, end of 2h intestinal incubation. Samples include spray dried juice (SJ), freeze dried juice (FJ), freeze dried CRF (FC), mild pasteurised freeze-dried juice (MJ), mild pasteurised freeze-dried CRF (MC), intense pasteurised freeze-dried juice (IJ), intense pasteurised freeze-dried CRF (IC).



Figure 6.8: Release of free fatty acid (α-linolenic acid, ALA) in different treated spinach chloroplasts during *in vitro* digestion. Time 0 represents G0, the end of 2 min oral phase, without adding any enzyme. In theory, no digestion happens yet at this time point, therefore, it can be seen as undigested CRFs; Time 2 h represents G, end of 2 h gastric incubation; Time 4h represents I, end of 2h intestinal incubation. Samples include spray dried juice (SJ), freeze dried juice (FJ), freeze dried CRF (FC), mild

FFA

pasteurised freeze-dried juice (MJ), mild pasteurised freeze-dried CRF (MC), intense pasteurised freeze-dried juice (IJ), intense pasteurised freeze-dried CRF (IC).

6.6.2.2. MGDG and FFA content in gastric phase (G)

During gastric phase, approximately 68%, 13% and 17% of MGDG have been digested in FJ, FC and MC (Figure 6.7), and about 13-54% of FFA released in all samples (Figure 6.8). Similar results were also reported by other researchers. According to Wattanakul et al. (2019), in freeze dried CRF from blanched spinach leaves in the presence of RGE in gastric phase (30min 37°C), the decline of galactolipids in the gastric phase is low, with MGDG decreasing from 25-26 to 22-23 mg/g CRF and DGDG decreased from 18-22 to 17-18 mg/g CRF. Similarly, Sutcharit found that 15% of MGDG in spinach CRF were hydrolysed during the gastric incubation with RGE (2h 37°C) (Sutcharit et al., 2023). The significant lower digestion of MGDG in FC compared with FJ could possibly be due to its physicochemical and structural difference. It is probably that the aggregated particle structure of FC hindered the hydrolysis of lipid-protein complex during digestion compared with porous particle structure of FJ.

For SJ, IJ and IC treatments, instead of being digested, MGDG in the gastric phase showed a significant increasing trend (SJ [G₀]: $12.5\pm3.2 \rightarrow$ [G] 35.3 ± 1.6 mg/g lipid CRF; IJ [G₀]: $3.3\pm0.6 \rightarrow$ [G] 53.1 ± 2.0 mg/g lipid CRF; IC [G₀]: $11.0\pm1.3 \rightarrow$ [G] 38.8 ± 3.9 mg/g lipid CRF). This could be due to the inactivation of acylation enzymes in chloroplast after high temperature during spray drying and intense pasteurisation. And acyl-MGDG contained in initial samples were deacylated to MGDG during gastric incubation (Figure 6.6 B). As shown in Figure 6.6 A and B, qualitatively, acyl-MGDG is contained in all samples before digestion and decreased diversely after 2h gastric incubation. Interestingly, the slightly higher band intensity of MGDG was observed in SJ, IJ and IC at time point of gastric phase (G) compared with G0, while the anticipated rise in MGDG was not detected in others, which confirms our hypothesis. According to Heinz et al. (Heinz et al., 1978), the acyl-MGDG-forming enzyme was contained in chloroplast envelope membranes, which has activity to convert MGDG to acyl-MGDG. This acylation enzyme activity was activated in disrupted tobacco leaves under homogenization with acidic conditions or in incubation at 40 °C (Matsuzaki et al., 1982). Contrary to tobacco leaves, the acylation in spinach leaves is even stronger, since the acyl-MGDG made up 20% of the leaf lipids after 3 hours incubation at 4 °C (Heinz, 1973). Thus, we suspect that freeze drying or mild pasteurisation will still maintain acylation enzyme activity in spinach chloroplasts, further after incubated at 37 °C for 2 hours at an acidic pH of 5 during the gastric phase, part of the MGDG was hydrolysed to MGMG, while some of the MGDG was converted to acyl-MGDG.

6.6.2.3. MGDG and FFA content in intestinal phase (I)

During the intestinal phase, galactolipids were drastically digested in all samples, and the MGDG were significantly decreased and high amounts of FFA produced (Figure 6.7 and Figure 6.8). However, as mentioned in section 6.5, approximately 30% of lipid loss occurs after sampling (6 ml) from gastric phase (20 ml) once entering into intestinal phase. Thus, the intestinal data listed above (Figure 6.7 and 6.8) was adjusted to its original content in order to minimise the impact of sampling issue.

Analysis revealed that significant MGDG hydrolysis was found in all samples throughout the 2 hours incubation during the intestine phase for example, 46% and 68% of MGDG were digested in FC and MC at the end of intestinal digestion, and MGDG was fully hydrolysed all other samples. The results indicate that 243-590 mg of FFA/g of lipid CRF were released in all samples at the intestinal phase. Moreover, there seem to be minimal differences in the digestion of galactolipids between the different chloroplast samples. During the whole in vitro digestion process, the main lipid digestion of spinach chloroplast materials occurred in the intestinal phase. This was due to the presence of PPE, as previously confirmed by Salhi et al. (2020). This study found that PPE has low galactolipase and cholesterol esterase activities. The low galactolipase and cholesterol esterase activities of PPE indicated that pancreatic lipase-related protein 2 (PLRP2) and carboxyl ester hydrolase/bile salt stimulated lipase (CEH/BSSL) are present at low levels in this enzyme preparation. Importantly, PLRP2 as the essential enzymes for galactolipid digestion was recently found in porcine pancreatic juice and reported to have the highest galactolipase activity. CEH/BSSL also known as cholesterol esterase, which is the dominant lipases in the intestine involved in lipid digestion, can hydrolyse both galactolipids and cholesterol ester. As reported by Minekus (Minekus et al., 2014), PPE is a mixture of digestive enzymes with broad activities which includes protease, pancreatic- α -amylase, pancreatic lipase and colipase. The activities of galactolipase and cholesterol esterase are not found in classic native

human (nHPL) and porcine (nPPL) pancreatic lipase (Amara et al., 2009). Therefore, we hypothesise that the activity of galactolipase found in PPE could originate from both PLRP2 and CEH/BSSL. This hypothesis is also confirmed by the result of Wattanakul et al. (2019), who compared the addition of PPE and HPJ (Human Pancreatic Juice) during the intestinal digestion phase. The observation from their work is that galactolipids of CRF were hydrolysed at a slower rate when RGE and PPE were combined for *in vitro* digestion compared with RGE and HPJ. As the galactolipase activity of HPJ is higher than PPE, with 8.40 and 0.28 U/mg on MDDG-C8 at pH 6 respectively (Salhi et al., 2020). The higher galactolipase activity of HPJ is associated with the higher extent of PLRP2 (Andersson et al., 1996). In addition, PPE has a higher galactolipase activity when MGDG is the substrate compared with DGDG, while the rate of hydrolysis of MGDG and DGDG being similar in HPJ (Wattanakul et al., 2019).

6.6.3.Protein changes during heat treatment and *in vitro* digestion determined by SDS-PAGE

SDS-PAGE was performed to evaluate the protein changes from different treated spinach chloroplasts during *in vitro* digestion, the electropherogram is presented in Figure 6.9. Rubisco protein (54kDa) was chosen as a maker for protein changes as it is the major protein present in chloroplasts. Previous research has reported that heat treatment can break the lipid-protein complex in the food matrix and thus increase the bioaccessibility of lipophilic nutrients (Platel et al., 2016). This study was designed to establish if the method of chloroplast material preparation affects the profile of protein

digestion. Starting materials were dispersed in water then treated as described in the methods (see Section 2.3.4.4); samples from the digestion study were also treated as described. An aliquot equivalent to 20 μ g of protein was loaded onto the SDS-PAGE gel in each case.

Before digestion (Figure 6.9A), a gradual decrease in the Rubisco band intensity was observed in pasteurised juice which suggests the denaturation and aggregation of insoluble protein with lower solubility caused by heat treatment. Typically, temperatures higher than 60°C are able to achieve the selective precipitation of proteins including the Rubisco (Opdensteinen et al., 2021). Despite the higher protein concentration in FC compared with FJ (Table 6.1), the band intensity of Rubisco seemed lower. This could be due to the lower solubility of FC (43%) than FJ (74%) powders, see chapter 3, that lowers protein in FC following solubilization. Also, we assume the intense band near the top of the lane (above 250 kDa) in FC is made up of protein aggregates of Rubisco. This band disappeared in MC and IC, this can be explained by the protein dissociated and re-arranged to form aggregates with structure changed (lower molecular mass) when expose to heat (Wang et al., 2017).

During digestion (Figure 6.9 B, C and D), most of the visible bands were noted below a molecular weight of 20 kDa at the end point of each gastric and intestinal phase. This indicates that most of the digested protein can be attributed to protein hydrolysis by pepsin and pancreatic enzymes (Rao et al., 2020). Qualitatively, larger amount of digested protein below 20 kDa was detected in pasteurised samples (Panel C and D), since the Rubisco was denatured and more resistant to hydrolysis (Wang et al., 2014). This finding concludes that heat treatment may inhibit the digestibility of proteins from spinach juice/CRF materials.



Figure 6.9: Sodium Dodecyl Sulfate-Polyacrylamide Gel Electrophoresis (SDS-PAGE) analysis of spinach chloroplast protein during *in vitro* digestion (M: Standard molecular mass; Panel A: initial samples before digestion. Panel B, C and D: spinach chloroplast samples during digestion. Lane 1-3, SJ, SJ-G, SJ-I; Lane 4-6, FJ, FJ-G, FJ-I; Lane 7-9,

FC, FC-G, FC-I; Lane 10-12, MJ, MJ-G, MJ-I; Lane 13-15, MC, MC-G, MC-I; Lane 16-18, IJ, IJ-G, IJ-I; Lane 4-6, IC, IC-G, IC-I. 54kDa, Rubisco)

6.6.4. Overall discussion

Several investigations in our group have focused on the process of galactolipids digestion in chloroplasts rich fraction (CRF) from spinach during *in vitro* digestion. Wattanakul and colleagues used a two-step static *in vitro* digestion model for studying the galactolipids digestion of blanched spinach CRF (Wattanakul et al., 2019). The model is maintained at 37 °C and starts from stomach phase at pH 5 with the presence of RGE (20 U/ml) incubated for 30 min, and then followed with an intestinal phase at pH 6, with the addition of human pancreatic juice (HPJ) or PPE and bile salt (NaTDC 4mM) incubated for 60min. The results have shown that blanching spinach leaves (85 °C, 3min) can knock out endogenous galactolipase activity inside the plant materials, avoiding the loss of galactolipids and the release of FFA.

Sutcharit's research also focuses on the galactolipids digestion of fresh freezedried spinach CRF and use the same digestion model as this study with slightly difference of gastric pH (half gastric emptying pH 5.5 for liquid sample) (Sutcharit et al., 2023). The main findings are identical to this study, that the MGDG, DGDG, acyl-MGDG and acyl-DGDG were detected in spinach chloroplast samples before digestion. And the major hydrolysis of both non-acylated and acylated galactolipids occurred in the intestinal phase, which accompanied by a significant increase in MGMG and DGMG values.

Despite the rapid advances achieved in our group, there are still significant gaps in our understanding of how various foods with different compositions, physical properties and structures prepared with different processing procedures are affected during gastric-intestinal (GI) tract. Therefore, the key purpose of this study was to establish the impact of spinach chloroplast processing or preparation procedures in the digestion of nutrients (galactolipids and proteins). As the various spinach chloroplast structures caused by different food processing technologies (spray drying vs freeze drying, heat treatment vs non-heat treatment) may be disintegrated dissimilarly during gastric-intestinal tract. Understanding the influence of food processing procedures on the digestion of nutrients in spinach chloroplast samples within the in vitro digestion model will assist the development of new products with enhanced functionalities. As well known, the complex food structure is a major limiting factor for the digestion of nutrients (Chen, 2009). Thermal treatment can stimulate the breakdown of protein-lipid complex and enhance the release of lipophilic nutrients (Gomes et al., 2021). In this context, the production and processing steps for spinach chloroplast in this study was aim to preserve their quality, address their limitations and promote their functionality for health attributes.

6.7. Conclusion

In this study, we have shown that both mild and intense pasteurisation (70 °C 15s and 90 °C 5min) of spinach juice inactivate endogenous enzyme activity. Therefore, it is recommended that thermal treatment should be applied in spinach juice/CRF powders production to avoid lipid hydrolysis, accumulation of FFA and further oxidation of polyunsaturated fatty acids.

During the simulated *in-vitro* digestion, the galactolipids from spinach juice/CRF could be digested. Moreover, thermal pretreatment can enhance the digestibility of GLs from spinach juice/CRF with higher digestion rate especially found in intense pasteurised samples. GLs maintained relatively stable in gastric phase with the presence of RGE, and predominately digested in small intestinal phase with the main enzyme PLRP2 and CEH/BSSL which is found in PPE. Furthermore, PPE hydrolyses MGDG more extensively than DGDG. We have also shown that the acylated galactolipids (acyl-MGDG and acyl-DGDG) were also digested throughout the digestion and mainly digested by the enzymes in PPE. After digestion, the main FFA released from spinach chloroplast samples during is ALA, and the increasing rate is low in gastric phase, but considerably raising upon entering the intestinal phase.

In addition, it can be seen that proteins in spinach juice/CRF samples were digested by pepsin and PPE during digestion, and majority of protein hydrolysis occurred in gastric phase. However, heat pretreatment was found inhibit the digestion of proteins with larger amount of digested protein below 20 kDa was detected in pasteurised samples during digestion compared with fresh samples. This suggests that heat treatment could lead to a reduction of protein digestion due to their denaturation. Overall, this work indicated that thermal pretreatment can stimulate the digestion of galactolipids from spinach juice/CRF but hinder the digestion of proteins.

Chapter VII

7. Bioaccessibility of lipophilic nutrients from spinach juice/CRF during *in vitro* digestion

7.1. Introduction

Spinach chloroplasts are rich in proteins and lipophilic nutrients such as carotenoids (β -carotene and lutein), α -tocopherol (vitamin E), chlorophylls (a and b), and omega-3 fatty acids (constituents of galactolipids). The consumption of these nutrients is associated with health benefits in the human body, as they can decrease oxidative stress, lower the relative risk of cancer and chronic diseases, and promote brain, cardiovascular, and eye health (Roberts & Moreau, 2016; Lasya, 2022). It is therefore important to research how the digestive process impacts on these nutrients derived from chloroplast products. In this chapter, the release of carotenoids, vitamin E and galactolipids from spinach chloroplast-rich materials, produced by different food processing technologies, during simulated human digestion was evaluated.

Bioaccessibility is a definition used to describe the availability of nutrients from foods needed for use by the cells and tissues of our bodies. An understanding of bioaccessibility is useful as it allows researchers a means to ascertain nutritional efficiency of food and food formula in the development of products aimed at improving human health (Fernández-García et al., 2009). In essence, bioaccessibility relates to the proportion of a compound that is released from a food matrix in the gastrointestinal tract, that then becomes available for intestinal absorption and distributed systemically. Currently, this process has been investigated using several validated *in vitro* models (Brodkorb et al., 2019).

Interestingly, it has been reported that the bioaccessibility of carotenoids and vitamin E from fruits and vegetables are generally quite low (Reboul et al., 2006; Lemmens et al., 2014). This is mainly attributed to two main causes, the limited release from food matrix (Aguilera, 2019; Vitali Čepo et al., 2020) and the low solubility in gastrointestinal fluids (Savjani et al., 2012). During food preparation, processing and/or mastication the fractured plant cell can promote the release of carotenoids from food plant tissues. Following release from the food matrix, the major limiting factor is solubility of carotenoids/vitamin E in digesta given their hydrophobic character (Faulks et al., 2005). In this situation, improving carotenoids and vitamin E bioaccessibility of fruits and plants food by specific methodologies and technologies has becoming a worldwide topic (Failla et al., 2005). In Chapter 3 and 4, we recovered intact chloroplasts without the confinement of cell wall by juicing spinach leaves using a twin-gear juicer, and then produce dried powders with different processing protocols, spray drying or freeze drying; with or without pasteurisation. The bioaccessibility of carotenoids and α -tocopherol from these materials was investigated in this study. The impact of food processing, e.g., drying techniques and thermal pre-treatment on the bioaccessibility of these lipophilic nutrients was measured and compared.

It is known that various factors influence nutrient bioaccessibility including dietary factors such as fat, fibre, dosage of carotenoid, location of the carotenoids in the

plant tissue, heat treatment, particle size of food, carotenoid species, interactions among carotenoids, isomeric form and molecular linkage and subject characteristics. (Priyadarshani, 2017). In this study, we focused on the impact of the food matrix and its microstructural characteristics on the release of selected nutrients. The microstructure characteristics are a result of the application of specific unit operations such as thermal processing, mechanical processing, etc (Rodríguez-Roque et al., 2016; Sun et al., 2018). Thus, the hypothesis of this study was that the thermal pre-treatment e.g. pasteurisation of spinach juice prior to CRF production will improve the bioaccessibility of lipophilic nutrients. Also, the spray drying as a cheaper, faster and more efficient alternative to freeze drying was also used to produce dried spinach juice fine powders. The experiment will also investigate whether the drying method will modulate bioaccessibility of spinach bioactive compounds without the confinement of plant cell wall during digestion. All these seven spinach chloroplasts productions, SJ (Spray dried juice); FJ (Freeze dried juice); FC (Freeze dried CRF); MJ (Mild pasteurised juice); MC (Mild pasteurised CRF); IJ (Intense pasteurised juice); IC (Intense pasteurised CRF) was sieved to include 250 µm or less and their composition before and during digestion were measured and compared with an in vitro digestion model as described in section 6.2. After digestion, centrifugation was used to separate the micelles from total digesta, and the micelles phase need to be passed through a 0.2 µm PTFE filter. It is assumed that the lipophilic nutrients are solubilised in aqueous phase of micelles and only the filtered micelles can be considered being able to pass

through the mucous layer in the small intestine and absorbed by the epithelium cells (Yang et al., 2013).

7.2. Results and discussion

Samples preparation steps are shown in Section 6.3.

7.2.1.Lipophilic nutrients of spinach chloroplasts before *in vitro* digestion

The concentration (mg/g DW and mg/g lipid) of lipid micronutrients in the chloroplast samples were measured before digestion (Figure 7.1). The initial samples before digestion were prepared as liquid samples (juice/CRF powders mixed with water 1.5:3.5, w/w). The addition of water makes it a consumable food as suggested by the INFOGEST method in 2019: high-solid foods must be reconstituted in liquids to reflects real food or meal (Brodkorb et al., 2019).

Spinach CRF contained higher lipophilic nutrients content (carotenoids and α -tocopherol) than juice, and pasteurisation can significantly reduce these compositions in CRFs but maintained relatively stable in juice (Figure 7.1a). The highest nutrients content can be found in FC with β -carotene and lutein was 3.05 and 4.29 mg/g DW, and α -tocopherol was 0.41mg/g DW. While, more than 50 % of these nutrients were lost in mild and intense pasteurised CRFs. The explanations were discussed in section 3.2.1.3 and 4.2.2.4 in detail.

The lipophilic nutrients content (wet extraction) shown in Figure 7.1 differs from

the previous results reported for the dry extracted material described in chapter 3 and 4, showed an overall decreasing trend in wet system. This was suspected to be caused by differences in the extraction efficiency between solid and liquid samples. Since the addition to water in dried samples prior to extraction may decrease the extraction efficiency and will lead to the reduced amount of lipophilic nutrients including carotenoids and α -tocopherol. Based on the research by Balasubramanian et al. (2013), a biomass moisture content higher than 5% reduces the lipid extraction efficiency. It is hypothesized that water molecules in wet biomass form a hydrophilic outer layer of chloroplast material prevents the penetration of non-polar solvents from reaching lipids, and further hinders the lipid extraction process. The efficiency of extraction, measurements, and identification is crucial in carotenoids and α -tocopherol analysis. The method used to extract carotenoids in this study includes extraction with chloroform and methanol from dry powders and liquid samples (with or without addition of water in initial samples), drying under gas nitrogen, dissolving with acetone, separating on HPLC, identification and quantification. Identification is a prerequisite step for accurate quantification, it can be achieved by the retention times attained through co-chromatography with authentic carotenoid standards (Butnariu, 2016).



Figure 7.1: Lipophilic nutrient composition of spinach juice/CRF before digestion. Data were expressed as a MEAN±SD of triplicate independent experiments. Different letters indicate significant statistical differences according to one way ANOVA and Tukey test (p<0.05). Samples include spray dried juice (SJ), freeze-dried juice (FJ), freeze-dried CRF (FC), mild pasteurised freeze-dried juice (MJ), mild pasteurised freeze-dried CRF (MC), intense pasteurised freeze dried juice (IJ), intense pasteurised freeze dried CRF (IC).

7.2.2.Behaviour of spinach juice/CRF during *in vitro* static digestion model

The *in vitro* static digestion was performed in an incubator at 37 °C containing the digestion tubes that were rotating at a speed of 20 rpm. Rotation was used to provide a continuous force allowing for the samples to mix with the digestive fluids and enzymes. This process was set to mimic the oral, gastric, and intestinal digestion in human. The transformation/behaviour of spinach chloroplasts during in-vitro static digestion model was imaged by light microscope at ×40 magnification, the images were taken at the end of oral phase (2 min), gastric phase (2h) and intestinal phase (2h). The impact of *in vitro* digestion on the different treated spinach chloroplasts at each digestion phase was revealed from the following pictures shown in Figure 7.2.

For all samples, the initial 1.5g weight of spinach chloroplast powders were mixed with 3.5 mL of water to the final volume of 5 mL, added 5 mL SSF incubate for 2 min and then 10 mL SGF (without enzymes), the oral phase sampling was taken at this time in order to make the dilution factor same as the gastric phase. As for the gastric and intestinal phase, the sampling was taken after another 2 and 4 hours' incubation. Then 1M NaHCO₃ was added to the sampling of both oral and gastric phase separately (1:1, v/v), to raise the pH to 8. The gastric lipase activity was stopped and the samples were diluted by half, which in agreement with the dilution factor of the intestine phase (Egger et al., 2016). Therefore, in all of the samples, the dilution factors are the same, the reduction in number of chloroplasts and reduced particle size was not due to the dilution effect only the addition of enzymes at each stage.

In the oral phase, the samples were incubated at a short time (2 min) without adding enzyme, which indicates undigested status as initial samples. For un-pasteurised samples, the SJ showed smaller particle size of chloroplasts compared to FJ, and FJ was smaller than FC. This confirms that spray drying usually produces particles with a mean particle size below 10 μ m (Daniel et al., 2017), in this study, the mean particle size of spray dried juice was 9.81±0.22 μ m. In contrast, the freeze-dried samples showed larger particle size (FJ 66.44±0.38 μ m and FC 113.95±4.44 μ m), this was influenced by the material structure and grinding. As for pasteurised samples, the particle size increased with the heating temperature and time in both juice and CRF samples (data shown in Table 4.1 and Table 4.3). Moreover, the pasteurised samples showed darker brown colour due to the transformation of chlorophylls (green) to pheophytins (olive brown). The formation of pheophytins is initiated by the release of cellular acids and synthesis of new acids during heating, the magnesium atom in the chlorophyll porphyrin ring was replaced by two hydrogen ions (Erge et al., 2008).

In the gastric phase, all dried juice samples (SJ, FJ, MJ, IJ) were digested to significantly smaller particles by the addition of RGE and pepsin, while CRFs (FC, MC, IC) still maintain most of its large particles/aggregates. At the end of the intestine phase, with the addition of porcine pancreatin extracts (PPE) and bile salts, all the samples were digested to small particles (in digesta). The digesta was centrifuged and filtered through 0.2µm PTFE filter to obtain the micelle. The micelle is formed from bile salt

and lipid digestion products (free fatty acids, monoglycerides and fat-soluble substances such as vitamins and cholesterol). The micelles are water soluble small fat droplets and enable the transportation of lipid digestion productions to small intestinal epithelium for absorption (Cheng et al., 2020). Therefore, the formation of micelles is a critical step which can affect digestibility.

The same structure reduction of spinach juice and CRF powders during digestion was also be found iby Authors in our group (Gedi et al., 2017) (Wattanakul et al., 2019).



Figure 7.2: Microscope images of spinach juice/CRF with different treatment during *in-vitro* static digestion model, includes spray dried juice (SJ), freeze dried juice (FJ), freeze dried CRF (FC), mild pasteurised freeze- dried juice (MJ), mild pasteurised 192

freeze-dried CRF (MC), intense pasteurised freeze-dried juice (IJ), intense pasteurised freeze-dried CRF (IC). The spinach chloroplast powders were incubated in oral (2min), gastric (2 h) and intestinal (2 h) phase with *in vitro* static digestion model.

7.2.3.Carotenoids release from chloroplast material during simulated human digestion

7.2.3.1. Isomerisation of β -carotene both before and after digestion

The carotenoids in all samples before and after *in vitro* digestion (both total digesta and micelle at the end of intestinal phase) were analysed by HPLC. Two standards were used in this experiment, β -carotene (all-trans/all-E) and lutein respectively. Naturally, the majority β -carotene exists in the all-trans form (Hiranvarachat et al., 2008). However, the trans-to-cis isomerisation of β -carotene was found occur in both food processing especially thermal treatment and during digestion. The major cis-form of β carotene is 15-15'-di-cis- β -carotene, 13-cis- β -carotene and 9-cis- β -carotene (Marx et al., 2003; Pénicaud et al., 2011).

In this study, prior to the *in vitro* digestion, all samples showed two small peaks of cis- β -carotene (around RT 21 and 23 min) which represents 13-cis- β -carotene and 9-cis- β -carotene respectively (Figure 7.3). The isomers of β -carotene were identified by comparison with Wattanakul's publication (Wattanakul et al., 2022). This isomerisation is mainly caused by the processing procedure, apart from heat treatment, β -carotene can also be affected by mechanical treatment. Firstly, the juicing step in preparation can cause the loss of tissue integrity by disrupting cellular structure and the activated

endogenous enzymes can promote the release of organic acids (e.g. oxalic acid), which will result in the formation of cis-form β -carotenes. Secondly, the spinach juice is exposed to heat and oxygen during pasteurisation which will further stimulate the trans to cis (E-Z) isomerisation process (Pénicaud et al., 2011). The level of isomerisation is increased with the heating temperature and time as shown in Figure 7.3. The isomerisation or degradation of carotenes can both occur during processing and storage, therefore, in order to minimise non-enzymatic oxidation on storage, the CRFs were vacuum packed in aluminum bags and kept in -20 °C. Any nutrient losses in predigested samples discussed in this work are likely due to the processing steps prior to freezing.

After *in vitro* digestion, there was an increasing trend of cis- β -carotenes levels observed in all samples, comparing their peak area and dilution factor (data not shown). This suggests that the isomerisation of trans to cis- β -carotene also occurs during *in vitro* digestion. Moreover, greater levels of cis-isomers were found in pasteurised samples after digestion. In addition, the food matrix and degree of processing play important role in carotenoid isomerisation and bioaccessibility (Aherne et al., 2010). An increase of cis-isomers and decrease of trans-isomers of carotenoids in carrots during *in-vitro* digestion has been reported by other researchers; Tao et al. (2021) found that all trans- β -carotene reduced 75% and degraded to 15-cis- and 9-cis- β -carotene during *in vitro* digestion and the predominant degradation occurs in intestinal phase compared with oral and gastric phase. Moreover, recent studies have proved that cis- β -carotene is preferred to be micellarised than all-trans- β -carotene during the small intestinal phase; while cellular uptake appears higher for the all-trans-form (Bohn et al., 2019). Both all trans and cis isomers of β -carotene are transferred across the brush-border surface of the enterocyte from mixed micelles and absorbed in humans by passive diffusion (Ferruzzi et al., 2006).

In conclusion, these findings confirmed that the trans-to-cis isomerisation of β carotene from spinach juice/CRF occurs during both thermal treatment and simulated digestion.









Figure 7.3: Chromatogram of β -carotene in spinach juice/CRF before and after 198
digestion. Peak 1:lutein; Peak 2:13-cis- β -carotene; Peak 3: all-trans- β -carotene; Peak 4: 9-cis- β -carotene. The scale of chromatogram is not consistent and the peak areas should not be compared directly to each other due to different dilution factors. The slight shift in retention time of all β -carotenes before and after digestion in this study were due to the retention time drift in HPLC overtime, and the standards were run in between every five samples in order to confirm the target compound.

7.2.3.2. Micellarisation rate of carotenoids after *in vitro* digestion

In this study, micellarisation rate after *in vitro* digestion shows the ability of nutrients from spinach juice/CRF to be incorporated into mixed micelles and potentially absorbed by intestinal cells.

The micellarisation rate of carotenoids from fresh and pasteurised spinach juice/CRF after *in vitro* digestion is shown as concentration of β -carotene/lutein in micelle in relative to their concentration in total digesta in terms of percentage (%) (see Figure 7.4) and mg/100g DW (see Table 7.2). In fresh samples, higher β -carotene and lutein micellarisation rate was obtained in juice compared with CRF, with 11% and 30% in SJ, 12% and 13% in FJ, while only 3% and 7% in FC. The data also suggests that spray drying of spinach juice resulted in higher lutein micellarisation rate than freeze drying. In addition, even though carotenoids were found to be higher in FC than SJ and FJ (Figure 3.3), their micellarisation rate was relatively lower.

It is known that multiple factors interfere with the carotenoids micellarisation rate, including food matrix (particle size, location of carotenoids in tissue), carotenoids type (species, interactions among carotenoids, isomeric form, molecular linkages), heat pretreatment, digestion environment (Priyadarshani, 2017). In this study, we suspect the reason could be the particle size and morphology differences between samples, CRF showed more compacted structure with the largest particle size, and spray dried juice showed more porous structure with the lowest particle size. According to Lemmens et al. (2010), the significantly reduced particle size for both raw and cooked carrots showed enhanced ability of carotenoids to be incorporated into the micelle phase during digestion. On the other hand, dried juice contained lower lipids (%) which makes it more easily to be solubilised in GI tract (Table 3.1 and Table 4.2).

In addition, pasteurisation prior to freeze drying in this study was found to promote the micellarisation rate of carotenoids (β -carotene and lutein) in both dried juice and CRF after digestion. Pasteurised CRF still showed lower carotenoids micellarisation rate than juice. It is worth noticing that micellarisation rate of lutein significantly increased in pasteurised juice and β -carotene significantly increased in pasteurised CRF. Micellarisation depends on the degree of food processing, high processing degree can maximise the amount of compound which is soluble from the matrix (Fernández-García et al., 2012). Several researchers have shown that β -carotene is generally less efficiently transferred to the micelles than lutein (Ryan et al., 2008), however the transfer efficiency varies among foods (Granado-Lorencio et al., 2007). The type of food matrix and the composition of final digesta can enhance or inhibit its transfer efficiency. It is also known that carotenoids interact during incorporation and absorption and then type and the relative amounts of carotenoids present in the final digesta (i.e., free versus ester forms) may become relevant (Tyssandier et al., 2001). The transfer of carotenoids may be prevented because they are deeply embedded in the food tissues and because of interfacial characteristics (i.e., surface-active proteins) or electric charges (ζ -potential) (Rich et al., 2003).



Carotenoids micellarisation rate

Figure 7.4: Carotenoids micellarisation rate of different processed spinach juice/CRF during digestion. Data were expressed as a MEAN±SD of triplicate independent experiments. Different letters indicate significant statistical differences according to one way ANOVA and Tukey test (p<0.05). Two separate groups of tests were conducted, with upper case letters represent group "all-trans- β -carotene" and lower case letters represent group "lutein". Samples include spray dried juice (SJ), freeze-dried juice (FJ), freeze-dried CRF (FC), mild pasteurised freeze-dried juice (MJ), mild pasteurised freeze dried CRF (MC), intense pasteurised freeze dried juice (IJ), intense pasteurised freeze dried CRF (IC).

7.2.3.3. Carotenoids bioaccessibility

In this study, "bioaccessibility" was defined as the proportion of nutrient concentration in micelle phase compared to the initial nutrient concentration in samples before digestion. This factor reflects how much nutrients is available for absorption by the human body, and the stability of the nutrient during digestion and liberation of the nutrient into the micelle phase. It provides a valuable indication to help select food source and dosage to ensure nutritional efficiency of food products as their bioactive compounds claiming health benefits (Chen & Roca, 2019).

The carotenoids bioaccessibility from the spinach juice/CRF during *in vitro* digestion is shown in Figure 7.5 expressed as % and in Table 7.2 expressed as dry weight (mg/100g, DW). In fresh samples, higher carotenoids bioaccessibility was found in dried juice (SJ and FJ) than CRFs (FC). The lowest carotenoids bioaccessibility was obtained in FC (β -carotene 0.5% and lutein 2%). In addition, both mild and intense pasteurisation conditions can enhance the carotenoid bioaccessibility in freeze dried spinach juice and CRF during digestion. While the most significant increase was observed after intense pasteurisation, with the highest carotenoids bioaccessibility was observed in the IC (β -carotene 44% and lutein 10%), followed by IJ (β -carotene 21% and lutein 8%). This result suggests that intense pasteurisation was sufficient enough to break the bond of carotenoids in spinach chloroplasts, consequently enhance the micellarisation of carotenoids.

It is widely believed that thermally processing can enhance the carotenoids

bioaccessibility, this improvement could be due to the thermal treatment separating the carotenoids from its complex with proteins, thus the carotenoids transferred more easily to micelles (Palmero et al., 2014). The enhanced bioaccessibility of β -carotene and lutein after thermal treatment was also found in cooked carrots by some researchers (Hornero-Méndez & Mínguez-Mosquera, 2007; Lemmens et al., 2011b; Knockaert et al., 2012). Furthermore, *in vivo* studies conducted by Rock et al proved that β -carotene plasma levels in subjects were 3 times higher after consuming of thermally processed carrot and spinach compared with fresh ones (Rock et al., 1992; Rock et al., 1998).



Figure 7.5: Carotenoids bioaccessibility of different processed spinach juice/CRF during *in vitro* digestion. Data were expressed as MEAN±SD of triplicate independent experiments. Different letters indicate significant statistical differences according to one way ANOVA and Tukey test (p<0.05). Two separate groups of tests were conducted, with upper case letters represent group "all-trans- β -carotene" and lower case letters represent group "lutein". Samples include spray dried juice (SJ), freeze-dried juice

(FJ), freeze-dried CRF (FC), mild pasteurised freeze-dried juice (MJ), mild pasteurised freeze-dried CRF (MC), intense pasteurised freeze dried juice (IJ), intense pasteurised freeze dried CRF (IC).

7.2.4.α-Tocopherol release from spinach juice/CRF during simulated human digestion

The micellarisation rate and bioaccessibility of α -tocopherol in spinach juice/CRF during *in vitro* digestion is shown in Table 7.1 and 7.2 (expressed as % and ug/g DW). From our results, the level of α -tocopherol retained in micelle or digesta phase after digestion in some samples might be too low to be detected (ND), therefore, these individual data were not shown.

In this study, SJ showed more efficiently micellarisation during digestion compared with FJ. As can be seen, micellarisation rate and bioaccessibility of α tocopherol in SJ was 20.5% and 10.5% respectively, and no data detected in FJ. This is probably due to the porous structure and significantly reduced particle size of spray dried samples in comparison to FJ. Moreover, spray drying as a food processing technique has been proved to both preserve the nutrient over time and enhance its stability, bioaccessibility and release of these bioactive compounds. So, it is considered as a suitable technique for encapsulation which may result in a stable powder (Ezhilarasi et al., 2013; Katouzian et al., 2016; Mujica-Álvarez et al., 2020).

Among freeze dried spinach juice/CRF samples, the micellarisation rate and bioaccessibility of α -tocopherol was found to be enhanced in pasteurised samples. The

values of micellarisation rate and bioaccessibility in MJ was twice as IJ (no data in FJ). In contrast, the micellarisation rate and bioaccessibility was only detected in IC with the value of 7.8% and 15.1% respectively (no data in FC and MC). This indicates that the release and micellarisation of α -tocopherol in spinach juice/CRF during digestion can be promoted by pasteurisation. As previous studies have demonstrated that thermal processing by disrupting the food matrix can enhance α -tocopherol transfer into digesta and micelles (Gounaris et al., 1983).

Apart from the food matrix, another important factor for α -tocopherol bioaccessibility is the amount and type of oil/fat in a meal. Wattanakul et al. (2022) compared the α -tocopherol bioaccessibility in freeze-dried fresh pea vine haulm CRF with thermal processed or the addition of oil, and the results revealed that both of these methods can promote its bioaccessibility but more conspicuous with the presence of oil.

Table 7.1: α-tocopherol digestibility of different processed spinach juice/CRF during *in vitro* digestion

/%	SJ	FJ	FC	MJ	MC	IJ	IC
Micellarisation rate	20.5±1.2 ^a	ND	ND	10.0±2.9 ^b	ND	5.0±1.8 ^b	7.8±1.3 ^b
Bioaccessibility	10.5±0.8 ^{ab}	ND	ND	8.9±2.9 ^{ab}	ND	5.0±2.3 ^b	15.1±3.2 ^a

Data was analysed by one-way ANOVA according to Tukey's test (p<0.05) to access the differences of the samples. Means (n = 3) designated with different letters in the

same row are significantly different (p<0.05). ND means not detected. Samples include spray dried juice (SJ), freeze dried juice (FJ), freeze dried CRF (FC), mild pasteurised freeze dried juice (MJ), mild pasteurised freeze dried CRF (MC), intense pasteurised freeze dried juice (IJ), intense pasteurised freeze dried CRF (IC).

Table 7.2: Carotenoids and α -tocopherol content of different processed spinach

	•••	ODE	1 •	•	• .	1	
4	11100/	1 1 1 1	dumma	110	T71 100	digastia	•
		1 N P	(11111110)		VIIIO	moesnoi	
		UIU.	uuiiic	111	1110	argontor	
			0			0	

All-trans-β-carotene			lutein			α -tocopherol			
mg/100g DW	undigested	micelle	digesta	undigested	micelle	digesta	undigested	micelle	digesta
SJ	78±3b	9±2cd	74±5cd	83±3e	6±1c	20±2d	26±0b	3±0a	13±1c
FJ	72±5b	8±1cd	68±9d	126±5cd	6±2c	49±5c	24±2bc	ND	ND
FC	305±38a	2±0d	54±5	429±35a	7±2bc	107±9a	41±1a	ND	ND
MJ	70±10b	13±0bc	103±5	121±8cde	10±1b	43±2bc	15±2e	$1\pm 0b$	13±1c
MC	92±5b	3±0d	27±2e	184±5b	8±1bc	61±12b	12±2e	ND	1±0d
IJ	85±4b	18±2b	123±14b	95±3de	7±0bc	32±3c	20±1d	$1\pm 0b$	20±2b
IC	110±16b	48±7a	270±21a	145±13bc	14±2a	101±10a	21±1cd	3±0a	40±3a

Data was analysed by one-way ANOVA according to Tukey's test (p<0.05) to access the differences of the samples. Means (n =3) designated with different letters in the same column are significantly different (p<0.05). Samples include spray dried juice (SJ), freeze dried juice (FJ), freeze dried CRF (FC), mild pasteurised freeze dried juice (MJ), mild pasteurised freeze dried CRF (MC), intense pasteurised freeze dried juice (IJ), intense pasteurised freeze dried CRF (IC). DW means dry weight.

7.3. Conclusion

Overall, in this chapter, the powdered spinach juice/CRF obtained through different

food processing technologies (spray drying vs. freeze drying, pasteurisation) were used during *in vitro* digestion model. The impact of food processing on digestion of lipophilic nutrients including carotenoids and α -tocopherol within simulated GI tract were discovered. It appears that enhanced bioaccessibility of lipophilic nutrients in spinach juice can be achieved by spray drying rather than freeze drying. Another aspect is that combination of pasteurisation of spinach juice prior to freeze drying can efficiently break the lipid-protein complex and stimulate the release of lipophilic nutrients, which further leads to higher proportion of micellisation during *in vitro* digestion. Given this context, the optimum pasteurisation conditions for spinach juice and CRF is mild pasteurisation (70 °C 15s) and intense pasteurisation (90 °C 5min) respectively. This study provides insight of production of spinach juice/CRF powders with the aim of achieving better bioaccessibility of lipophilic nutrients during simulated human GI tract.

Chapter VIII

8. Main Findings, Conclusion and Suggestion for future

work

8.1. Thoughts on main findings



Figure 8.1: Flowchart of experimental design

Tests on the nutritional credentials of spinach juice/CRF powder products confirmed that they contain various macronutrients including proteins, lipids, minerals, carbohydrates, and micronutrients such as carotenoids, chlorophylls, α -tocopherol and essential fatty acids. Not surprisingly, spinach CRF contains higher levels of lipophilic nutrients compared with spinach juice. Moreover, both spray drying and freeze drying procedures retain the nutrients of spinach juice to a similar extent. Further, thermal preservation techniques such as pasteurisation of juice and blanching of leaves which inactivate endogenous enzyme, eliminate/reduce the potential for pathogen contamination and growth all prevent quality loss during storage. However, thermal treatment of leaves prior to juicing, or of juice from fresh leaves causes some loss in nutrients in spinach juice/CRF powders to different extents. Significant loss of total lipids (28-36%) and lipophilic micronutrients was observed only in pasteurised CRF, but relatively stable in pasteurised juice. On the other hand, hot water blanching of spinach leaves (100°C 30s) causes 76% and 32% loss of total lipids in both freeze dried juice and CRF powders, along with the decrease of lipophilic nutrients. From these observations it seems that pasteurising juice is better than blanching leaves in terms of preparing spinach leaf-derived powders that are stable and rich in nutrients.

As a matter of fact, healthy food design should not only consider the composition, but also food structure, texture, shelf stability, dynamics and absorption of food components during digestion (Mengucci et al., 2022). Food structure can be a key indicator when developing food products that control the extent and rate of food lipids release during the digestion process (Guo et al., 2017). Whereas, it has always been overlooked in previous studies. For this reason, the correlations between the microstructure of powdered spinach juice/CRF samples and their nutrients digestibility was studied. The *in vitro* digestion model had recently been developed by researchers across Europe (Quadram Institute), and this new INFOGEST 2.0 was adopted in the current work (Sams, 2016; Brodkorb et al., 2019). It allows researchers to follow structure changes, nutrients release and digestion, solubilization of food components along the different phases of GI tract (mouth, stomach and small intestine). Following the *in vitro* digestion model, the investigation was conducted to test the impact of processing procedure on the digestibility of nutrients from chloroplasts.

The structure of spinach juice/CRF samples in this study is the result of combinations of extrusion, heating and drying preparations. The liberation of nutrients from the food matrix and the intestinal micellization are main factors that affect bioaccessibilities during simulated digestion model (Eriksen et al., 2017). It is assumed that the liberation of chloroplasts from their cell via juicing removes one major barrier to digestion of chloroplasts and the release of their nutrients, that is the cell wall (Li et al., 2019b). In addition, the morphology and particle size of dried spinach juice/CRF powders is also an important food matrix factor for their digestibility. Several studies suggested that reduced particle size improves bioaccessibility of nutrients by increasing the surface area available for enzymes to act and thus stimulate release of bioactive compounds from food matrix (Hedren et al., 2002). Meanwhile, the heat treatment of plant biomass can soften the plant tissues, disrupt cell membranes and cell walls, more importantly break the protein-carotenoid complex (Priyadarshani, 2017). From the results reported in this thesis, the spray dried juice (SJ) showed porous spherical structures and small particle size (D_{4,3}, 9.81µm), and freeze dried juice (FJ) and freeze dried CRF (FC) showed smooth and compact plate shaped structures with larger particle size (D_{4,3}, 66.44 and 113.95 µm). A higher bioaccessibility of carotenoids

from SJ was observed compared with FJ, and lowest bioaccessibility was in FC despite its high nutritional value. In addition, pasteurised spinach juice/CRFs powders obtained by freeze drying had larger particles (D_{4,3}, 69.53-182.11 μ m) with rougher surface compared with fresh samples. The results reveal that higher bioaccessibility of carotenoids was found from pasteurised samples. Overall, these findings confirmed that food matrix did play a vital role in releasing and uptake of nutrients in GI tract to different extents and from different aspects.

In addition, thermal treatment enhanced the digestion of galactolipids but hindered proteins digestion derived from spinach chloroplasts. The majority of these galactolipids, MGDG and DGDG respectively, were digested in the small intestinal phase. This finding was confirmed by the increase in MGMG and DGMG bands intensity and released FFA (ALA) throughout the digestion process. In contrast to lipid digestion, the majority of spinach chloroplast protein hydrolysis occurred in the gastric phase in the presence of pepsin. Moreover, a higher amount of digested proteins (peptides below 20 kDa) was detected in pasteurised samples compared with fresh samples during digestion. This reveals that heat treatment impede the protein hydrolysis. These findings are closely linked to the bioaccessibility results, that thermal treatment of plant-based food can increase permeability of plant tissues and enhance the hydrolysis of nutrients (Pallares et al., 2018). It has been suggested that protein-rich food may cause slower rates of lipid digestion (Salentinig, 2019). The protein digestion occurs mainly in gastric and intestinal phase, facilitating the subsequent release of fat aggregates from the degraded juice/CRF bolus (Žolnere et al., 2019).

The absorption of lipophilic nutrients (i.e., carotenoids) is affected not only by the disruption of the food matrix that stimulate their release, but also by the emulsification into lipid droplets and incorporation in mixed micelles (Wang et al., 2012b) . During the intestinal stage of digestion, lipophilic bioactive compounds are incorporated with other lipids (i.e., cholesterol, phospholipids) and lipid digestion products (i.e., free fatty acids, monoacylglycerols, lysophospholipids) into mixed micelles, then becoming available for uptake in the epithelial cells of the small intestine (Reboul, 2019a). That is the reason why Wattanakul et al. (2022) added rapeseed oil during digestion with the purpose of improving the bioaccessibility of lipophilic nutrients in pea vine haulm CRF. In this present study, porcine bile extracts were used in small intestinal phase during simulated GI tract for the formation of mixed micelles where the digestive enzymes can act on. The accumulation of FFA was incorporated into mixed micelles at the start of small intestinal phase which facilitated carotenoid micellarisation.

Overall, the findings of this study support the concept of preparing nutrient-rich powders from underutilized green biomass and highlight the impact of processing parameters on the materials properties, and hence digestibility, of these powders.

8.2. Conclusions

Overall, this thesis was set out to investigate the impact of food processing procedures on the nutritional and functional properties of spinach chloroplast materials.

The static *in vitro* digestion model was also developed to gain a detailed insight into the influence of food processing on the digestibility of lipophilic nutrients and proteins in chloroplast materials. The main findings are as follows:

- Chloroplast can be liberated from the confinement of the plant cell wall by disrupting the green biomass using a twin-gear juicer. This offers a simple, clean and effective procedure that circumvents the need for solvents or chemicals.
- Both spray drying and freeze drying methods produce dried spinach chloroplast materials which efficiently remove the moisture and stabilise the nutrients level for a long period of storage until further utilisation.
- Spinach chloroplasts contain high levels of various macro and micronutrients, including total proteins, lipids, carotenoids, chlorophylls, α-tocopherol and αlinolenic acid. Chloroplast rich fractions (CRF) contain higher total lipids and their lipophilic nutrients than spinach juice.
- 4. Heat treatment of green biomass or fresh juice such as blanching and pasteurisation can effectively inactivate endogenous enzymes and prevent the free fatty acid release due to the inhibition of endogenous galactolipases.
- 5. Blanching spinach leaves promotes a significant loss in lipophilic nutrients in both juice and CRF. However, pasteurisation of spinach juice from fresh leaves retains the levels of these nutrients in spinach juice, but not in CRF.

- Compared with fresh spinach juice/CRF powders, both mild (70°C 15s) and intense (90°C 5min) pasteurised spinach samples have improved bioaccessibility of lipophilic nutrients (carotenoids and α-tocopherol) as determined using *in vitro* digestion model.
- Pasteurisation of spinach juice prior to making juice/CRF powders enhances the digestibility of galactolipids but hinders protein digestion.
- 8. The majority of the galactolipids and acylated galactolipids digestion occurs in the intestinal phase during simulated digestion with the galactolipase active enzymes PLRP2 and CEH/BSSL which is found in PPE. High levels released free fatty acids especially α-linolenic acid, were noted in the current work. Moreover, the galactolipase activity in PPE hydrolyses MGDG more extensively than DGDG.
- 9. The majority of protein in spinach juice/CRF hydrolysed in gastric phase with the addition of pepsin during *in vitro* digestion. Compared with fresh spinach juice/CRF powders, larger amount of digested protein below 20 kDa was detected in pasteurised samples during digestion.

8.3. Suggestion for future work

Several limitations associated with the current study can be overcome. These optimization procedures are beyond the scope of the current project but will be detailed below for future reference.

As shown in previous studies, thermal processing has been widely used for food

preservation, and the nutrients bioaccessibility in thermal processed foods are positively promoted (Lemmens et al., 2011a; Barba et al., 2017). Therefore, pasteurisation and blanching of spinach biomass were compared in this study. The pasteurisation steps followed in this study involved samples being left in a cold room for a period of time (roughly 4 °C, 24h) prior to undergoing the heating steps. Even at 4°C, it is anticipated that undesirable accumulation of FFA due to the hydrolysis of galactolipids by endogenous enzymes in fresh spinach juice occurs. This problem can be overcome using blanched spinach leaves prior to juicing. The loss of galactolipids in blanched samples was merely due to leaching into the boiling liquor. However, the yield of blanched CRF is extremely low. In this consequence, the future work for the preparation of thermal processed chloroplast materials could be directly pasteurising fresh juice.

Despite the benefits of thermal treatment on plant-based foods, the product quality and the micronutrients content can be negatively affected. Therefore, the development of non-thermal technologies should be considered in future studies. The development of these approaches will be critical and needed to facilitate the production of extracts that contain thermally labile bioactive compounds (Jayathunge et al., 2017; Lorenzo et al., 2019; Ozkan et al., 2022). Recently, pulse electric fields (PEF), ultrasounds (US), high pressure processing (HPP) and high pressure homogenization (HPH) have been proposed for the purpose as a means to preserve the levels of thermally labile compounds found in plant tissues (López-Gámez et al., 2021). Moreover, emerging evidence suggests that some of these technologies can enhance plant bioactive nutrients bioaccessibility as a result of modifying the structure of plant tissues.

Critical in the development of any food-based product is the confirmation that any nutrient present is bioaccessible, and available for cellular uptake. To this end, the bioaccessibility of micronutrients in this study was investigated using an in vitro digestion model according to the INFOGEST protocol (Sams, 2016; Brodkorb et al., 2019). There is a major limitation in interpreting galactolipids digestion data by using HPTLC analysis as described in section 5.5. The interference bands on HPTLC plates next to target galactolipid will hinder the integration of their peak area, and cause troubleshooting for calculation. The interference bands in this study could be originated from the lipid contamination from the porcine bile extract. These inference bands were not observed in a previous study reported from our research group (Wattanakul et al., 2019). The key difference here being that in the work of Wattanakul, the porcine bile extract was replaced by the bile salt sodium taurodeoxycholate (NaTDC). Consequently, the inference bands (yellow/grey colour) were partially or wholly merged with the DGDG bands which leads to the difficulty in the identification and quantification of DGDG. Another major problem is the sampling efficiency during the intestinal phase. The addition of PPE within the intestinal phase step caused clumps to form, and gelatinization of the reaction fluid inside the digestion tube. The formation of clumps and gelatinization of the mixture hindered the sampling efficiency and made analysis of samples difficult. This phenomenon happens due to the mechanical forces generated

by the rotating tubes, as these forces are not strong enough to break the PPE and mix the cocktails homogenously. This step clearly needs some improvement and possibly the use a propeller inside of the tube revolver to facilitate. The propeller would provide a more evenly and thoroughly mix reaction fluid to aid interactions between the food matrix with the PPE. This hypothesis was confirmed in the work of Sutcharit et al. (2023) who developed a modified digestion device containing an installed propeller, pH-stat and thermometer. In addition, it is worth considering an alternative to the PPE in future work to develop a more reliable analytical model. PPE is commonly used and recommended for use in the INFOGEST model. However, as reported by Salhi et al. (2020), PPE is not an ideal substitution of the lipolytic enzymes contained in human pancreatic juice (HPJ). Indeed, as compared galactolipase activities of PPE with human (HPJ) and porcine (PPJ) pancreatic juice, it was found lowest in PPE $(0.31 \pm 0.01 \text{ U/mg})$ and higher levels in PPJ (1.78±0.02 U/mg) then HPJ (7.73±0.2 U/mg) at pH 8 (Salhi et al., 2020). The galactolipase activity relies on the pancreatic lipase-related protein 2 (PLRP2) and carboxylester hydrolase/bile salt-simulated lipase (CEH/BSSL) (Sias et al., 2004; Kergomard et al., 2023). The low galactolipase activities of PPE indicate that the PLRP2 and CEH/BSSL are still contained but in low levels and were not detected by mass spectrometry (Salhi et al., 2020). Given this evidence, PPJ or HPJ should be considered in future digestion studies for better testing the bioaccessibility of the dietary lipids.

In this study, we have used in vitro static model, and the enzyme selection and

incubation conditions are adapted by the study objective. Whilst the static model is widely used with the advantages of simple and cost-effective to operate, the model can only maintain initial conditions at each phase in GI tract, it is therefore not a comparable simulation of complicate in vivo conditions (Thuenemann, 2015). Dynamic models are more complicated but more physiologically relevant than static models (Dupont et al., 2015). Recently, several *in vitro* dynamic digestion models have been developed, such as the TIM-1, DIDGI and ESIN, they are able to closely mimic the food digestion during the GI tract (Dupont et al., 2019). These models can reproduce the dynamic transit times, digestive enzymes secretion rate, dynamic pH changes, and peristaltic movements such as shearing and grinding which can largely affect the breakdown of large particles and release of nutrients of food matrix during digestion process (Sensoy, 2021). The data obtained from these models are validated against in vivo data, which showed that they accurately reproduced the gradual transition of ingested food through the GI tract (Barroso et al., 2015). Until now, only a limited number of studies have focused on the comparison of static and dynamic in vitro digestion models on the digestibility of carotenoids and proteins. Lee et al. (2018a) compared the static digestion model and dynamic model as a means to estimate the bioaccessibility of lutein in high-fat foods. Similarly, Egger et al. (2019) determined the milk proteins hydrolysis in skim milk powders with both static (INFOGEST) and dynamic (DIDGI) models. The results from both studies exhibited strong similarities to in vivo data from the pig. However, the dynamic model showed better physiological situation at the level of free

amino acid release. Nonetheless, *in vitro* digestion protocols have their limitations to completely mirror the complexity of the *in vivo* situation, which leaves vast potential for future studies (Bohn et al., 2018). An *in vivo* study has been conducted in our group by Gedi et al. (2019), by including the dried spinach leaf powders and chloroplast rich fraction (CRF) partially as fishmeal content in a fish feeding trail for their growth and nutrients uptake. The results showed that the zebrafish fed with CRF showed higher lutein and PUFA (alpha-linolenic acid, C18:3 n-3 and hexadecanoic acid, C16:3) which is in agreement with *in vitro* data.

Furthermore, the recovery of total lipids and biomolecules from spinach could be an important challenge for sustainable resource exploitation. This gives insight for the utilisation of natural green plants as nutritional supplements, has an additional advantage in terms of environmental protection through food processing waste minimization. For this purpose, a mass balance of the processing could be carried out determining the mass of lipid and lipophilic nutrients recovered from the fresh/heat treated spinach and comparing to the total dry starting mass.

Overall, there is growing interest in the development of plant-based food materials which contain high nutritional quality coupled with the presence of functional food ingredients. Future research may explore emerging technologies (non-thermal treatments) for the production of quality stable spinach chloroplasts. The parameters of static *in vitro* digestion model could be further updated in accordance with the specific research target for better mimicking the GI environment, including the transition time, digestive enzymes, pH, mechanical forces, etc. Furthermore, both dynamic *in vitro* digestion model and *in vivo* digestion is worth to be conducted for understanding the food digestion behaviours during human GI tract in order to design healthier food products. Despite the expensive cost and ethically limitations of human feeding trials, it is still worth to be conducted by consuming chloroplasts during daily meal. This may provide more reliable data for the bioacessibility of chloroplast nutrients and comparable with the fish feeding trial data and test the accuracy of *in vitro* studies.

IMPACT OF THE COVID-19 PANDEMIC

The first national lockdown (March 2020- September 2020) occurred when I was in 2nd year of my research. I went back to China during May-August 2020, and returned back to lab in December 2020.

Following nearly a year of complete closure, the laboratories in my building remained at minimum capacity (25%) until 01/06/21. The chemistry laboratory (where I conducted around 90 % of my practical work) was limited to a maximum occupancy of 2 people at this time. Given that I was one of around 20 lab users, this significantly limited the amount of practical work I was able to complete. After 01/06/21, the maximum occupancies of some labs in my building increased, but my primary laboratory remained at a maximum occupancy of 2 persons.

On 27/07/21 our safety team announced plans for a phased approach, planning to return to 100% occupancy before the start of term. The building did not resume 100% capacity until 30/09/21. This disrupted my work further. Consequently, closures resulting from the pandemic forced me to push back experimental work. This compounded with supply chain delays, which repeatedly delayed experiments.

In September, the lab reopened with more restrictions, only 1 working slot (8:30am – 12:30pm or 13:00pm – 17:00pm) can be booked per day and maximum 3 days week per person. A lot of paper work are required to be completed and approved, including procedure risk assessments (PRA, for every lab-related activity), activity risk assessment (ARA, for every lab-related activity), permission to work (PTW), moodle

induction course, moodle quiz (covid-19), guide to working during covid-19 pandemic.

The pandemic also affected me personally on my mental health. During March and April 2020, I was isolated in my flat, experiencing daily anxiety, stress and weightloss. May - August 2020, upon returning to China, I endured a five-week quarantine in a hotel. During this period, I can only read papers, write paper manuscripts and have online meetings with my supervisor and colleagues. Concerns about my PhD studies weighed heavily on my mind. September 2020, I went back to campus to get prepared for laboratory work. However, after the 2nd year viva, my internals doubted my ability to complete PhD study with suggestions for a shift to an MPhil study. Seeking another chance to prove myself, I got approval from both my supervisor and internals, with the final decision deferred until a re-viva in May 2021. I retuned back to my lab experimental work in December and since that day I worked at least 5 days a week in the lab until June 2022. Since 2021, regular consultations with a therapist became necessary, and I grappled with persistent anxiety, depression, and insomnia, a struggle that persists to this day.

Appendices

Chapter 5:

	Spinach	Spinach after	Juice	Sieved juice ¹	Yield
	(g)	blanching (g)	(ml)	(ml)	(ml/kg)
Control	500	N/A	350	300	600
Blanched ²	500	465	300	235	470

Table 5.1: Yield of fresh and blanched spinach juice (ml/kg)

1: Both fresh and blanched spinach juice was filtered through a 75 µm sieve.

2: Blanching condition (100 °C 30s)

Table 5.2: Yield of freeze dried fresh and blanched spinach juice/CRF powders (%

DW)

Yield/% DW	Juice	CRF
Control	4	0.74
Blanched	1.1	1.1

The yield of freeze dried fresh and blanched spinach juice and CRF powders was calculated by the mass of powders passed through 250 μ m sieve compared with raw spinach.

 Table 5.3: Main nutrients loss of spinach juice/CRF under pasteurisation and blanching

conditions

Nutrianta laga/0/	Juice				CRF		
INULLIEILIS IOSS/ 70	MJ	IJ	BJ	MC	IC	BC	
Total lipids	3	9	76	28	36	32	
Total chlorophylls	8	39	93	55	70	59	
Total carotenoids	/	1	92	51	57	70	
VE	85	50	89	71	24	36	
ALA	/	/	92	24	29	55	

Abbreviations: MJ Mild pasteurised freeze dried juice; MC Mild pasteurised freeze

dried CRF; IJ Intense pasteurised freeze dried juice; IC Intense pasteurised freeze dried

CRF. BJ Blanched dried juice; BC Blanched dried CRF. Total Chl. Total Chlorophylls (*a+b*); *Total carotenoids: beta-carotene and lutein.*

Chapter 6:



Figure 6.1: The total lipid content extracted from 1 ml sampling during digestion. Samples include spray dried juice (SJ), freeze dried juice (FJ), freeze dried CRF (FC), mild pasteurised freeze-dried juice (MJ), mild pasteurised freeze-dried CRF (MC), intense pasteurised freeze-dried juice (IJ), intense pasteurised freeze-dried CRF (IC). Control experiments are spinach chloroplasts replaced with pure water by mass and undergoing the same digestion process as other samples. G0: End of oral phase, G: End of gastric phase, I: End of intestinal phase.



Figure 6.2: Band intensity of different treated spinach chloroplasts at the end of *in vitro* digestion. Peak 1 stands for the area of DGDG, peak 2 stands for the interference band overlapped with the bands for DGDG and MGMG.



Figure 6.3: Representative of HPTLC chromatography for the quantification of polar lipid (galactolipid) in spinach chloroplast materials. A) HPTLC plate after developing and before staining with thymol solution B) HPTLC plate after staining with thymol solution and charring at 110 °C 10min. Samples loading volume at 15 μL. Polar lipids standards: monogalactosyl diglycerides (MGDG) and digalactosyl diglycerides (DGDG).



Figure 6.4: Representative of HPTLC chromatography for the quantification of nonpolar lipid (fatty acid) in spinach chloroplast materials. A) HPTLC plate after developing and before staining with copper acetate-phosphoric acid solution B) HPTLC plate after staining with copper acetate-phosphoric acid solution and charring at 180 °C 15min. Samples loading volume at 3 μ L. Nonpolar lipids standards: mix lipid standard and alpha-linolenic acid. Mix lipid standard consist of monoacylglycerol (MAG), 1,2 diacylglycerol (1,2DAG) 1, 1,3 diacylglycerol (1,3DAG) and triacylglycerol (TAG).

Both polar and non-polar lipids were analysed first without staining which allowed qualitative visualisation of pigments such as carotenoids and chlorophylls. (Figure 6.3 A and Figure 6.4 A).

HPTLC separation coupled to derivatization by thymol-sulfuric acid reagent and scanning densitometry was applied in this experiment for the quantification of polar lipids (galactolipids) present in spinach chloroplasts and galactosylated lipolysis products resulting from *in-vitro* digestion (Figure 6.3 B). Thymol solution is used to selectively stain the galactolipids (MGDG & DGDG) avoiding the interference of other pigments, such as chlorophylls that may occur during the densitometric analysis by the HPTLC. Initially, thymol solution was used for detection of carbohydrates, that a sugar solution turns carmine red when mixed with a solution of thymol with 15–20% alcohol and excess sulphuric acid (Molisch, 1886). In 2021, Sahaka and co-workers used the thymol-sulfuric acid reagent for the qualitative analysis of galactolipids (Sahaka et al., 2021). In this experiment, the galactolipids was detected on thin layer chromatography (TLC) plates by charring in the presence of thymol solution at 110 °C for 10 min, after which the bands of samples and standards started to turn brown with higher sensitivity. MGDG and DGDG was used as markers to monitor the breakdown of chloroplast membrane during the in vitro digestion.

As shown in Figure 6.3 A, there are some yellow pigments showed on the plate before staining with thymol staining and charring. We suspect these yellow pigments are carotenoids in different forms (i.e. beta-carotene or lutein). The carotenoid content in different samples is shown in Figure 4.4, CRFs contain higher carotenoid concentrations than juice; significant apparent losses of carotenoids is only observed in CRFs after pasteurisation (48-61% losses, DW) but remained relatively stable in juice. This may be due to real losses of carotenoids, but we have already observed the dilution of lipids in CRF samples due to the increase of proteins in the CRF pellet caused by precipitation of denatured proteins after heat treatment. The intensity of these yellow bands follows the same trend, which confirms our hypothesis.

Carotenoids will stay yellow and not interact with thymol solution which stains galactose rings in galactolipids and appears burgundy red after heat treatment. However, some of these carotenoid pigments commigrated with the acyl-MGDG bands at Rf 0.8, which will still influence the band intensity quantification of acyl-MGDG. Therefore, with the lack of pure standard of acyl-MGDG and difficulty in its band intensity quantification, we can only do qualitative analysis.

As for non-polar lipids, copper acetate-phosphoric acid solution was utilised for the quantitative analysis of free fatty acid release from spinach chloroplast materials during *in-vitro* digestion (Figure 6.4 B). This method is able to quantitatively detect unsaturated lipids, including phospholipids, neutral lipids, fatty acids, and hydrocarbons (such as squalene)(Churchward et al., 2008). The use of copper acetatephosphoric acid solution as a fluorometric staining reagent offers reproducible improvements over existing fluorescent reagents such as Nile Red or molybdenum blue reagent after charring, at the same time provides more significantly enhanced sensitivity of detection, ease of use, and lower cost (Fewster et al., 1969). Copper acetate in phosphoric acid is able to develop a strong intensity of staining regardless of the degree of unsaturation, the higher unsaturation result in higher staining intensity. In this experiment, the lipids were detected on thin layer chromatography (TLC) plates by charring in the presence of copper salts at 180 °C for 15min, after which the bands of samples and standards started to turn into black with higher sensitivity.

 Table 6.1: Calculation of the expected amount of lipids in spinach chloroplast samples

 during digestion.

			Exposted	Extracted		Total (a	Apparent
G0	Sample/g	Lipid/%	lipid/g	lipid/g in	Volume/ml	lipid/20ml)	extraction
				G0			efficiency/%
SJ	1.5	13	0.195	0.0037	20	0.074	38
FJ	1.5	12	0.18	0.0044	20	0.088	49
FC	1.5	28	0.42	0.0165	20	0.33	79
MJ	1.5	12	0.18	0.0055	20	0.11	61
MC	1.5	20	0.3	0.0093	20	0.186	62
IJ	1.5	11	0.165	0.0047	20	0.094	57
IC	1.5	18	0.27	0.0112	20	0.224	83

	Expected	Extracted		Total (g lipid/20ml)	Apparent
G	lipid/g in	lipid/g in G	Volume/ml		extraction
	G				efficiency/%
SJ	0.195	0.0053	20	0.106	54
FJ	0.18	0.0057	20	0.114	63
FC	0.42	0.018	20	0.36	86

MJ	0.18	0.0062	20	0.124	69	
MC	0.3	0.012	20	0.24	80	
IJ	0.165	0.0052	20	0.104	63	
IC	0.27	0.0091	20	0.182	67	
						_

I	Expected lipid/g in I	Extracted lipid/g in I	Volume/ml	Total (g lipid/28ml)	Apparent extraction efficiency/%
SJ	0.1365	0.004	28	0.112	82
FJ	0.126	0.0048	28	0.1344	107
FC	0.294	0.0093	28	0.2604	89
MJ	0.126	0.0038	28	0.1064	84
MC	0.21	0.0054	28	0.1512	72
IJ	0.1155	0.0036	28	0.1008	87
IC	0.189	0.0057	28	0.1596	84

(The expected lipid amount in intestinal phase is 30% lower from G, as 6ml sampling took out from 20ml of gastric tube.)










Figure 7.1: Chromatogram of α -tocopherol in spinach chloroplasts before and after digestion. The scale of chromatogram is not consistent and the peak areas should not be compared directly to each other due to different dilution factors. The slightly shift retention time of α -tocopherols before and after digestion in this study were due to the retention time drift in HPLC overtime, and the standards were set in between every five samples in order to confirm the target compound. (The initial/undigested samples, the α -tocopherol peak is around 17-18 min, while the digested ones are around 14-15 min).

References

Aamir, Ovissipour, Sablani, & Rasco. (2013). Predicting the Quality of Pasteurized Vegetables Using Kinetic Models: A Review. International Journal of Food Science, 2013, 271271. doi:10.1155/2013/271271

Abdallah, & Kamal. (2018). Influence of process variables on physical characteristics of spray freeze dried cellulose nanocrystals. Cellulose, 25(10), 5711-5730. doi:10.1007/s10570-018-1975-0

Abdel-Aal el, Akhtar, Zaheer, & Ali. (2013). Dietary sources of lutein and zeaxanthin carotenoids and their role in eye health. Nutrients, 5(4), 1169-1185. doi:10.3390/nu5041169

Abu Al-Qumboz, & Abu-Naser. (2019). Spinach Expert System: Diseases and Symptoms. International Journal of Academic Information Systems Research (IJAISR), 3(3), 16-22.

Actis-Goretta, Leveques, Rein, Teml, Schäfer, Hofmann, . . . Williamson. (2013). Intestinal absorption, metabolism, and excretion of (–)-epicatechin in healthy humans assessed by using an intestinal perfusion technique. The American journal of clinical nutrition, 98(4), 924-933.

Adams. (2007). The Principles of Freeze-Drying. In J. G. Day & G. N. Stacey (Eds.), Cryopreservation and freeze-drying protocols (pp. 15-38). Totowa, NJ: Humana Press.

Agoreyo, Akpiroroh, Orukpe, Osaweren, & Owabor. (2011). The effects of various drying methods on the nutritional composition of Musa paradisiaca, Dioscorea rotundata and Colocasia esculenta. Asian Journal of Biochemistry, 6(6), 458-464.

Aguilar, Rodríguez-Herrera, Montaňez-Sáenz, de la Luz Reyes-Vega, & Contreras-Esquivel. (2004). Blanching at low temperatures: A thermal bioprocess applied to fruits and vegetables to improve textural quality. Food Science and Biotechnology, 13(1), 104-108.

Aguilera. (2019). The food matrix: implications in processing, nutrition and health. Critical Reviews in Food Science and Nutrition, 59(22), 3612-3629. doi:10.1080/10408398.2018.1502743

Aherne, Daly, Jiwan, O'Sullivan, & O'Brien. (2010). Bioavailability of β -carotene isomers from raw and cooked carrots using an in vitro digestion model coupled with a human intestinal Caco-2 cell model. Food Research International, 43(5), 1449-1454. doi:https://doi.org/10.1016/j.foodres.2010.04.026

Akinyele, Keshinro, & Akinnawo. (1990). Nutrient losses during and after processing of pineapples and oranges. Food Chemistry, 37(3), 181-188.

Alegbeleye, Odeyemi, Strateva, & Stratev. (2022). Microbial spoilage of vegetables, fruits and cereals. Applied Food Research, 2(1), 100122. doi:https://doi.org/10.1016/j.afres.2022.100122

Ali. (2022). High-performance liquid chromatography (HPLC): a review. Ann. Adv. Chem, 6, 010-020.

Aman, Schieber, & Carle. (2005). Effects of Heating and Illumination on Trans–Cis Isomerization and Degradation of β -Carotene and Lutein in Isolated Spinach Chloroplasts. Journal of Agricultural and Food Chemistry, 53(24), 9512-9518. doi:10.1021/jf050926w

Amara, Barouh, Lecomte, Lafont, Robert, Villeneuve, . . . Carrière. (2010). Lipolysis of natural long chain and synthetic medium chain galactolipids by pancreatic lipase-related protein 2. Biochimica et Biophysica Acta (BBA)-Molecular and Cell Biology of Lipids, 1801(4), 508-516.

Amara, Lafont, Fiorentino, Boullanger, Carrière, & De Caro. (2009). Continuous measurement of

galactolipid hydrolysis by pancreatic lipolytic enzymes using the pH-stat technique and a medium chain monogalactosyl diglyceride as substrate. Biochimica et Biophysica Acta (BBA)-Molecular and Cell Biology of Lipids, 1791(10), 983-990.

Amin, Karim, Yusof, & Muhammad. (2021). Effects of Enzymatic Liquefaction, Drying Techniques, and Wall Materials on the Physicochemical Properties, Bioactivities, and Morphologies of Zinc-Amaranth (Amaranthus viridis L.) Powders. Int J Food Sci, 2021, 1819104. doi:10.1155/2021/1819104

Amit, Uddin, Rahman, Islam, & Khan. (2017). A review on mechanisms and commercial aspects of food preservation and processing. Agriculture & Food Security, 6(1), 51. doi:10.1186/s40066-017-0130-8

An, Zhao, Wang, Wu, Xu, & Xiao. (2016). Comparison of different drying methods on Chinese ginger (Zingiber officinale Roscoe): Changes in volatiles, chemical profile, antioxidant properties, and microstructure. Food Chemistry, 197, 1292-1300.

Anandharamakrishnan, Rielly, & Stapley. (2010). Spray-freeze-drying of whey proteins at subatmospheric pressures. Dairy Science & Technology, 90(2), 321-334.

Anderson, Conway, & Peplinski. (1970). Gelatinization of Corn Grits by Roll Cooking, Extrusion Cooking and Steaming. Starch - Stärke, 22(4), 130-135. doi:10.1002/star.19700220408

Andersson. (2011). BSSL and PLRP2: key enzymes for lipid digestion in the newborn examined using the Caco-2 cell line. J Lipid Res, 52(11), 1949-1956. doi:10.1194/jlr.M015685

Andersson, Bratt, Arnoldsson, Herslöf, Olsson, Sternby, & Nilsson. (1995). Hydrolysis of galactolipids by human pancreatic lipolytic enzymes and duodenal contents. J Lipid Res, 36(6), 1392-1400.

Andersson, Carrière, Lowe, Nilsson, & Verger. (1996). Pancreatic lipase-related protein 2 but not classical pancreatic lipase hydrolyzes galactolipids. Biochimica et Biophysica Acta (BBA)-Lipids and Lipid Metabolism, 1302(3), 236-240.

Anthon, & Barrett. (2002). Kinetic Parameters for the Thermal Inactivation of Quality-Related Enzymes in Carrots and Potatoes. Journal of Agricultural and Food Chemistry, 50(14), 4119-4125. doi:10.1021/jf011698i

Arroqui, Rumsey, Lopez, & Virseda. (2001). Effect of different soluble solids in the water on the ascorbic acid losses during water blanching of potato tissue. Journal of Food Engineering, 47(2), 123-126.

Atalar, Gul, Saricaoglu, Besir, Gul, & Yazici. (2019). Influence of thermosonication (TS) process on the quality parameters of high pressure homogenized hazelnut milk from hazelnut oil by-products. J Food Sci Technol, 56(3), 1405-1415. doi:10.1007/s13197-019-03619-7

Azima, Azben, Refdi, Aulia, & Syukri. (2020). Effect of different cooking methods on the content of vitamin C, phenolics and minerals in several green leafy vegetables. Pak J Nutr, 19, 160-165.

Bahçeci, Serpen, Gökmen, & Acar. (2005). Study of lipoxygenase and peroxidase as indicator enzymes in green beans: change of enzyme activity, ascorbic acid and chlorophylls during frozen storage. Journal of Food Engineering, 66(2), 187-192.

Bai, Herforth, & Masters. (2022). Global variation in the cost of a nutrient-adequate diet by population group: an observational study. The Lancet Planetary Health, 6(1), e19-e28. doi:https://doi.org/10.1016/S2542-5196(21)00285-0

Bajpai. (2014). 7 - Chemicals Used in Deinking and Their Function**Some excerpts taken from Bajpai, 2006 with kind permission from Pira International, UK. In P. Bajpai (Ed.), Recycling and Deinking of Recovered Paper (pp. 121-137). Oxford: Elsevier.

Balasubramanian, Yen Doan, & Obbard. (2013). Factors affecting cellular lipid extraction from marine microalgae. Chemical Engineering Journal, 215-216, 929-936. doi:https://doi.org/10.1016/j.cej.2012.11.063

Baldelli, Oguzlu, Liang, Subiantoro, Woo, & Pratap-Singh. (2022). Spray freeze drying of dairy products: Effect of formulation on dispersibility. Journal of Food Engineering, 335, 111191. doi:https://doi.org/10.1016/j.jfoodeng.2022.111191

Barba, Mariutti, Bragagnolo, Mercadante, Barbosa-Cánovas, & Orlien. (2017). Bioaccessibility of bioactive compounds from fruits and vegetables after thermal and nonthermal processing. Trends in Food Science & Technology, 67, 195-206. doi:https://doi.org/10.1016/j.tifs.2017.07.006

Barbara, & Anne. (2000). Processing Effects on Dietary Antioxidants from Plant Foods. HortScience HortSci, 35(4), 580-584. doi:10.21273/HORTSCI.35.4.580

Barber. (1976). The Intact chloroplast / edited by J. Barber. Amsterdam: Elsevier Scientific Pub. Co. Barrett, & Lloyd. (2012). Advanced preservation methods and nutrient retention in fruits and vegetables. J Sci Food Agric, 92(1), 7-22. doi:10.1002/jsfa.4718

Barrett, & Theerakulkait. (1995). Quality indicators in blanched, frozen, stored vegetables. Food Technology, 49, 62-65.

Barroso, Cueva, Peláez, Martínez-Cuesta, & Requena. (2015). The Computer-Controlled Multicompartmental Dynamic Model of the Gastrointestinal System SIMGI. In K. Verhoeckx, P. Cotter, I. López-Expósito, C. Kleiveland, T. Lea, A. Mackie, T. Requena, D. Swiatecka, & H. Wichers (Eds.), The Impact of Food Bioactives on Health: in vitro and ex vivo models (pp. 319-327). Cham (CH): Springer

Copyright 2015, The Author(s).

Bassetto, Jacomino, Pinheiro, & Kluge. (2005). Delay of ripening of 'Pedro Sato' guava with 1methylcyclopropene. Postharvest Biology and Technology, 35(3), 303-308. doi:https://doi.org/10.1016/j.postharvbio.2004.08.003

Batziakas, Stanley, Batziakas, Brecht, Rivard, & Pliakoni. (2020). Reducing postharvest food losses in organic spinach with the implementation of high tunnel production systems. Agronomy for Sustainable Development, 40(6), 42. doi:10.1007/s13593-020-00648-x

Belitz, Grosch, & Schieberle. (2008). Food chemistry: Springer Science & Business Media.

Bergquist, Gertsson, & Olsson. (2006). Influence of growth stage and postharvest storage on ascorbic acid and carotenoid content and visual quality of baby spinach (Spinacia oleracea L.). Journal of the Science of Food and Agriculture, 86(3), 346-355. doi:10.1002/jsfa.2373

Bernat, Chafer, Rodríguez-García, Chiralt, & González-Martínez. (2015). Effect of high pressure homogenisation and heat treatment on physical properties and stability of almond and hazelnut milks. LWT-Food Science and Technology, 62(1), 488-496.

Bernhardt, & Schlich. (2006). Impact of different cooking methods on food quality: Retention of lipophilic vitamins in fresh and frozen vegetables. Journal of Food Engineering, 77(2), 327-333.

Berti, Faber, & Smuts. (2014). Prevention and control of micronutrient deficiencies in developing countries: current perspectives.

Bhatta, Stevanovic Janezic, & Ratti. (2020). Freeze-Drying of Plant-Based Foods. Foods, 9(1), 87. Retrieved from https://www.mdpi.com/2304-8158/9/1/87

Bicudo, Jó, Oliveira, Chaimsohn, Sierakowski, Freitas, & Ribani. (2015). Microencapsulation of Juçara (Euterpe edulis M.) Pulp by Spray Drying Using Different Carriers and Drying Temperatures. Drying Technology, 33(2), 153-161. doi:10.1080/07373937.2014.937872

Block, Dorne, Joyard, & Douce. (1983). Preparation and characterization of membrane-fractions enriched in outer and inner envelope membranes from spinach-chloroplasts. 2. Biochemical-characterization. Journal of Biological Chemistry, 258(21), 3281-3286.

Block, Douce, Joyard, & Rolland. (2007). Chloroplast envelope membranes: a dynamic interface between plastids and the cytosol. Photosynthesis Research, 92(2), 225-244. doi:10.1007/s11120-007-9195-8

Blondeau, Lipsky, Bourourou, Duncan, Gorelick, & Marini. (2015). Alpha-linolenic acid: an omega-3 fatty acid with neuroprotective properties-ready for use in the stroke clinic? Biomed Res Int, 2015, 519830. doi:10.1155/2015/519830

Bohn, Carriere, Day, Deglaire, Egger, Freitas, . . . Ménard. (2018). Correlation between in vitro and in vivo data on food digestion. What can we predict with static in vitro digestion models? Critical Reviews in Food Science and Nutrition, 58(13), 2239-2261.

Bohn, Desmarchelier, El, Keijer, van Schothorst, Rühl, & Borel. (2019). β -Carotene in the human body: metabolic bioactivation pathways – from digestion to tissue distribution and excretion. Proceedings of the Nutrition Society, 78(1), 68-87. doi:10.1017/S0029665118002641

Borgstrom, Dahlqvist, Lundh, & Sjovall. (1957). Studies of intestinal digestion and absorption in the human. J Clin Invest, 36(10), 1521-1536. doi:10.1172/jci103549

Bornhorst. (2016). Engineering digestion: multiscale processes of food digestion. Journal of Food Science, 81(3), R534-R543.

Borremans, Bußler, Sagu, Rawel, Schlüter, & Leen. (2020). Effect of Blanching Plus Fermentation on Selected Functional Properties of Mealworm (Tenebrio molitor) Powders. Foods, 9(7), 917. Retrieved from https://www.mdpi.com/2304-8158/9/7/917

Bouchnak, Brugière, Moyet, Le Gall, Salvi, Kuntz, . . . Rolland. (2019). Unraveling Hidden Components of the Chloroplast Envelope Proteome: Opportunities and Limits of Better MS Sensitivity. Mol Cell Proteomics, 18(7), 1285-1306. doi:10.1074/mcp.RA118.000988

Bouchnak, Moyet, Salvi, Kuntz, & Rolland. (2018). Preparation of chloroplast sub-compartments from Arabidopsis for the analysis of protein localization by immunoblotting or proteomics. JoVE (Journal of Visualized Experiments)(140), e58581.

Brigelius-Flohé, & Traber. (1999). Vitamin E: function and metabolism. The FASEB Journal, 13(10), 1145-1155. doi:https://doi.org/10.1096/fasebj.13.10.1145

Brodkorb, Egger, Alminger, Alvito, Assunção, Ballance, ... Carrière. (2019). INFOGEST static in vitro simulation of gastrointestinal food digestion. Nature Protocols, 14(4), 991-1014.

Bunea, Andjelkovic, Socaciu, Bobis, Neacsu, Verhé, & Camp. (2008). Total and individual
carotenoids and phenolic acids content in fresh, refrigerated and processed spinach (Spinacia
oleraceaoleraceaL.).FoodChemistry,108(2),649-656.doi:https://doi.org/10.1016/j.foodchem.2007.11.056

Burrows, & Connor. (2020). Chloroplast Distribution in the Stems of 23 Eucalypt Species. Plants

(Basel), 9(12). doi:10.3390/plants9121814

Burton, & Traber. (1990). Vitamin E: antioxidant activity, biokinetics, and bioavailability. Annu Rev Nutr, 10, 357-382. doi:10.1146/annurev.nu.10.070190.002041

Butnariu. (2016). Methods of Analysis (Extraction, Separation, Identification and Quantification) of Carotenoids from Natural Products. Journal of Ecosystem & Ecography, 6. doi:10.4172/2157-7625.1000193

Calín-Sánchez, Lipan, Cano-Lamadrid, Kharaghani, Masztalerz, Carbonell-Barrachina Á, & Figiel. (2020). Comparison of Traditional and Novel Drying Techniques and Its Effect on Quality of Fruits, Vegetables and Aromatic Herbs. Foods, 9(9). doi:10.3390/foods9091261

Çalışkan Koç, & Nur Dirim. (2017). Spray Drying of Spinach Juice: Characterization, Chemical Composition, and Storage. J Food Sci, 82(12), 2873-2884. doi:10.1111/1750-3841.13970

Calvo-Lerma, Fornés-Ferrer, Heredia, & Andrés. (2018). In Vitro Digestion of Lipids in Real Foods: Influence of Lipid Organization Within the Food Matrix and Interactions with Nonlipid Components. Journal of Food Science, 83(10), 2629-2637. doi:https://doi.org/10.1111/1750-3841.14343

Canjura, Schwartz, & Nunes. (1991). Degradation Kinetics of Chlorophylls and Chlorophyllides. Journal of Food Science, 56(6), 1639-1643. doi:https://doi.org/10.1111/j.1365-2621.1991.tb08660.x

Caparino, Tang, Nindo, Sablani, Powers, & Fellman. (2012). Effect of drying methods on the physical properties and microstructures of mango (Philippine 'Carabao' var.) powder. Journal of Food Engineering, 111(1), 135-148. doi:https://doi.org/10.1016/j.jfoodeng.2012.01.010

Capolino, Guérin, Paume, Giallo, Ballester, Cavalier, & Carrière. (2011). In Vitro Gastrointestinal Lipolysis: Replacement of Human Digestive Lipases by a Combination of Rabbit Gastric and Porcine Pancreatic Extracts. Food Digestion, 2. doi:10.1007/s13228-011-0014-5

Caro, Eydoux, Cherif, Lebrun, Gargouri, Carrière, & Caro. (2008). Occurrence of pancreatic lipaserelated protein-2 in various species and its relationship with herbivore diet. Comparative biochemistry and physiology. Part B, Biochemistry & molecular biology, 150, 1-9. doi:10.1016/j.cbpb.2008.01.007

Carpentier, Knaus, & Suh. (2009). Associations between Lutein, Zeaxanthin, and Age-Related Macular Degeneration: An Overview. Critical Reviews in Food Science and Nutrition, 49(4), 313-326. doi:10.1080/10408390802066979

Carrière, Renou, Ville, Grandval, & Laugier. (2001). Advantage of expressing the variations in some digestive parameters as a function of gastric emptying instead of time. Digestion, 64(1), 46-53.

Castenmiller, West, Linssen, van het Hof, & Voragen. (1999). The food matrix of spinach is a limiting factor in determining the bioavailability of β -carotene and to a lesser extent of lutein in humans. The Journal of Nutrition, 129(2), 349-355.

Chauhan, & Patil. (2013). Effect of packaging material on storage ability of mango milk powder and the quality of reconstituted mango milk drink. Powder Technology, 239, 86-93. doi:https://doi.org/10.1016/j.powtec.2013.01.055

Chen. (2009). Food oral processing—A review. Food hydrocolloids, 23(1), 1-25.

Chen, & Roca. (2019). Cooking effects on bioaccessibility of chlorophyll pigments of the main edible seaweeds. Food Chemistry, 295, 101-109.

Chen, Song, & Kirsch. (1983). Effects of blanching, freezing and storage on folacin contents of spinach. Nutrition Reports International (USA).

Chen, Zhuo, Chen, Wu, Hsieh, Liu, & Chu. (2014). Multiphoton imaging to identify grana, stroma thylakoid, and starch inside an intact leaf. BMC plant biology, 14, 175. doi:10.1186/1471-2229-14-175

Cheng, Mah, & Seluakumaran. (2020). Fat Digestion: Bile Salt, Emulsification, Micelles, Lipases, Chylomicrons. In H. M. Cheng, K. K. Mah, & K. Seluakumaran (Eds.), Defining Physiology: Principles, Themes, Concepts. Volume 2: Neurophysiology and Gastrointestinal Systems (pp. 63-65). Cham: Springer International Publishing.

Chinnaswamy, Rielly, & Stapley. (2007). Effects of Process Variables on the Denaturation of Whey Proteins during Spray Drying. Drying Technology, 25, 799-807. doi:10.1080/07373930701370175 Cho, Lee, Park, & Lee. (2001). Effects of Heat Pretreatment on Lipid and Pigments of Freeze-Dried Spinach. Journal of Food Science, 66(8), 1074-1079. doi:https://doi.org/10.1111/j.1365-2621.2001.tb16083.x

Chranioti, Chanioti, & Tzia. (2016). Comparison of spray, freeze and oven drying as a means of reducing bitter aftertaste of steviol glycosides (derived from Stevia rebaudiana Bertoni plant)– Evaluation of the final products. Food Chemistry, 190, 1151-1158.

Christensen. (2009). Galactolipids as potential health promoting compounds in vegetable foods. Recent Pat Food Nutr Agric, 1(1), 50-58. doi:10.2174/2212798410901010050

Chun, Lee, Ye, Exler, & Eitenmiller. (2006). Tocopherol and tocotrienol content of raw and processed fruits and vegetables in the United States diet. Journal of Food Composition and Analysis, 19, 196-204. doi:10.1016/j.jfca.2005.08.001

Churchward, Brandman, Rogasevskaia, & Coorssen. (2008). Copper (II) sulfate charring for high sensitivity on-plate fluorescent detection of lipids and sterols: quantitative analyses of the composition of functional secretory vesicles. J Chem Biol, 1(1-4), 79-87. doi:10.1007/s12154-008-0007-1

Ciurzyńska, Lenart, & Gręda. (2014). Effect of pre-treatment conditions on content and activity of water and colour of freeze-dried pumpkin. LWT-Food Science and Technology, 59(2), 1075-1081. Cruz, & Casal. (2013). Validation of a fast and accurate chromatographic method for detailed quantification of vitamin E in green leafy vegetables. Food Chemistry, 141(2), 1175-1180. doi:https://doi.org/10.1016/j.foodchem.2013.03.099

Cubas, Lobo, & Gonzalez. (2008). Optimization of the extraction of chlorophylls in green beans (Phaseolus vulgaris L.) by N,N-dimethylformamide using response surface methodology. Journal of Food Composition and Analysis, 21, 125-133. doi:10.1016/j.jfca.2007.07.007

Daniel, Ana Colette, Vitor, José Domingos, Maria, & Pedro. (2017). Spray Drying: An Overview. In P. Rosario & M. Teresa (Eds.), Biomaterials (pp. Ch. 2). Rijeka: IntechOpen.

Daza, Fujita, Fávaro-Trindade, Rodrigues-Ract, Granato, & Genovese. (2016). Effect of spray drying conditions on the physical properties of Cagaita (Eugenia dysenterica DC.) fruit extracts. Food and Bioproducts Processing, 97, 20-29. doi:https://doi.org/10.1016/j.fbp.2015.10.001

De Corcuera, Cavalieri, & Powers. (2004). Blanching of foods. Encyclopedia of agri, food and biological engineering. Marcel Dekker, New York City, NY, USA, 1-5.

Derbyshire. (2018). Micronutrient Intakes of British Adults Across Mid-Life: A Secondary Analysis

of the UK National Diet and Nutrition Survey. Front Nutr, 5, 55. doi:10.3389/fnut.2018.00055 Derrien, Badr, Gosselin, Desjardins, & Angers. (2017). Optimization of a green process for the extraction of lutein and chlorophyll from spinach by-products using response surface methodology (RSM). LWT - Food Science and Technology, 79, 170-177. doi:https://doi.org/10.1016/j.lwt.2017.01.010

Dhiman, & Prabhakar. (2021). Micronization in food processing: A comprehensive review of mechanistic approach, physicochemical, functional properties and self-stability of micronized food materials. Journal of Food Engineering, 292, 110248. doi:https://doi.org/10.1016/j.jfoodeng.2020.110248

Ding, Yu, Boiarkina, Depree, & Young. (2020). Effects of morphology on the dispersibility of instant whole milk powder. Journal of Food Engineering, 276, 109841. doi:https://doi.org/10.1016/j.jfoodeng.2019.109841

Dordevic, Kushkevych, Jancikova, Zeljkovic, Zdarsky, & Hodulova. (2020). Modeling the effect of heat treatment on fatty acid composition in home-made olive oil preparations. Open Life Sciences, 15(1), 606-618. doi:doi:10.1515/biol-2020-0064

Dörmann. (2013). Galactolipids in Plant Membranes. In eLS.

Dorne, & Heinz. (1989). Position and pairing of fatty acids in phosphatidylglycerol from pea leaf chloroplasts and mitochondria. Plant Science, 60(1), 39-46. doi:https://doi.org/10.1016/0168-9452(89)90041-1

Dorne, Joyard, & Douce. (1990). Do thylakoids really contain phosphatidylcholine? Proc Natl Acad Sci U S A, 87(1), 71-74. doi:10.1073/pnas.87.1.71

Duden, & Fricker. (1981). Enzymatic degradation of polar lipids in deep-frozen parsley. Zeitschrift für Ernährungswissenschaft, 20(3), 172-181. doi:10.1007/BF02021562

Dumbrava, Poenaru, & Moldovan. (2016). Effects of some thermal treatments on chlorophyll and carotenoids concentration from spinach (Spinacia oleracea L.) and savoy cabbage (Brassica oleracea var. Sabauda L.) leaves.

Dupont, Alric, Blanquet-Diot, Bornhorst, Cueva, Deglaire, . . . Lelieveld. (2019). Can dynamic in vitro digestion systems mimic the physiological reality? Critical Reviews in Food Science and Nutrition, 59(10), 1546-1562.

Dupont, & Mackie. (2015). Static and dynamic in vitro digestion models to study protein stability in the gastrointestinal tract. Drug Discovery Today: Disease Models, 17, 23-27.

Egger, Ménard, Baumann, Duerr, Schlegel, Stoll, . . . Portmann. (2019). Digestion of milk proteins: Comparing static and dynamic in vitro digestion systems with in vivo data. Food Research International, 118, 32-39.

Egger, Ménard, Delgado-Andrade, Alvito, Assunção, Balance, . . . Portmann. (2016). The harmonized INFOGEST in vitro digestion method: From knowledge to action. Food Research International, 88, 217-225. doi:https://doi.org/10.1016/j.foodres.2015.12.006

Egger, Schlegel, Baumann, Stoffers, Guggisberg, Brügger, . . . Portmann. (2017). Physiological comparability of the harmonized INFOGEST in vitro digestion method to in vivo pig digestion. Food Research International, 102, 567-574.

Ellison. (2016). Carotenoids: Physiology. In B. Caballero, P. M. Finglas, & F. Toldrá (Eds.), Encyclopedia of Food and Health (pp. 670-675). Oxford: Academic Press.

Ercan, & Soysal. (2011). Effect of ultrasound and temperature on tomato peroxidase. Ultrasonics Sonochemistry, 18(2), 689-695.

Erdman Jr, Poor, & Dietz. (1988). Factors affecting the bioavailability of vitamin A, carotenoids, and vitamin E. Food technology (USA).

Erge, Karadeniz, Koca, & Soyer. (2008). Effect of heat treatment on chlorophyll degradation and color loss in green peas. GIDA, 33, 225-233.

Erge, Koca, & Soyer. (2009). EFFECT OF HEAT TREATMENT ON CHLOROPHYLL DEGRADATION AND COLOR LOSS IN GREEN PEAS BEZELYELERDE.

Eriksen, Luu, Dragsted, & Arrigoni. (2017). Adaption of an in vitro digestion method to screen carotenoid liberation and in vitro accessibility from differently processed spinach preparations. Food Chemistry, 224, 407-413. doi:https://doi.org/10.1016/j.foodchem.2016.11.146

Espín, García-Conesa, & Tomás-Barberán. (2007). Nutraceuticals: facts and fiction. Phytochemistry, 68(22-24), 2986-3008.

Eun, Maruf, Das, & Nam. (2020). A review of encapsulation of carotenoids using spray drying and freeze drying. Critical Reviews in Food Science and Nutrition, 60(21), 3547-3572.

Ezhilarasi, Karthik, Chhanwal, & Anandharamakrishnan. (2013). Nanoencapsulation Techniques for Food Bioactive Components: A Review. Food and Bioprocess Technology, 6(3), 628-647. doi:10.1007/s11947-012-0944-0

Failla, & Chitchumronchokchai. (2005). In vitro models as tools for screening the relative bioavailabilities of provitamin A carotenoids in foods: Citeseer.

Fang, Selomulya, & Chen. (2007). On Measurement of Food Powder Reconstitution Properties. Drying Technology, 26(1), 3-14. doi:10.1080/07373930701780928

Fante, & Noreña. (2015). Quality of hot air dried and freeze-dried of garlic (Allium sativum L.). Journal of Food Science and Technology, 52, 211-220.

Faulks, & Southon. (2005). Challenges to understanding and measuring carotenoid bioavailability. Biochimica et Biophysica Acta (BBA) - Molecular Basis of Disease, 1740(2), 95-100. doi:https://doi.org/10.1016/j.bbadis.2004.11.012

Fazaeli, Emam-Djomeh, Ashtari, & Omid. (2012). Effect of spray drying conditions and feed composition on the physical properties of black mulberry juice powder. Food and Bioproducts Processing, 90(4), 667-675.

Fernández-García, Carvajal-Lérida, Jarén-Galán, Garrido-Fernández, Pérez-Gálvez, & Hornero-Méndez. (2012). Carotenoids bioavailability from foods: From plant pigments to efficient biological activities. Food Research International, 46(2), 438-450.

Fernández-García, Carvajal-Lérida, & Pérez-Gálvez. (2009). In vitro bioaccessibility assessment as a prediction tool of nutritional efficiency. Nutrition Research, 29(11), 751-760. doi:https://doi.org/10.1016/j.nutres.2009.09.016

Ferro, Seigneurin-Berny, Rolland, Chapel, Salvi, Garin, & Joyard. (2000). Organic solvent extraction as a versatile procedure to identify hydrophobic chloroplast membrane proteins. ELECTROPHORESIS, 21(16), 3517-3526. doi:https://doi.org/10.1002/1522-2683(20001001)21:16<3517::AID-ELPS3517>3.0.CO;2-H

Ferruzzi, Böhm, Courtney, & Schwartz. (2002). Antioxidant and Antimutagenic Activity of Dietary Chlorophyll Derivatives Determined by Radical Scavenging and Bacterial Reverse Mutagenesis Assays. Journal of Food Science, 67(7), 2589-2595. doi:https://doi.org/10.1111/j.1365-2621.2002.tb08782.x

Ferruzzi, Lumpkin, Schwartz, & Failla. (2006). Digestive Stability, micellarization, and uptake of beta-carotene isomers by Caco-2 human intestinal cells. J Agric Food Chem, 54(7), 2780-2785. doi:10.1021/jf0530603

Fewster, Burns, & Mead. (1969). Quantitative densitometric thin-layer chromatography of lipids using copper acetate reagent. Journal of Chromatography A, 43, 120-126.

Finten, Agüero, Jagus, & Niranjan. (2016). High hydrostatic pressure blanching of baby spinach (Spinacia oleracea L.). LWT, 73, 74-79. doi:https://doi.org/10.1016/j.lwt.2016.05.043

Fissore, Pisano, & Barresi. (2014). Applying quality-by-design to develop a coffee freeze-drying process. Journal of Food Engineering, 123, 179-187.

Folch, Lees, & Sloane Stanley. (1957). A simple method for the isolation and purification of total lipides from animal tissues. Journal of Biological Chemistry, 226(1), 497-509.

Franceschinis, Salvatori, Sosa, & Schebor. (2014). Physical and Functional Properties of Blackberry Freeze- and Spray-Dried Powders. Drying Technology, 32(2), 197-207. doi:10.1080/07373937.2013.814664

Fritschi, & Ray. (2007). Soybean leaf nitrogen, chlorophyll content, and chlorophyll a/b ratio. Photosynthetica, 45(1), 92-98. doi:10.1007/s11099-007-0014-4

Fuchs, Süß, Teuber, Eibisch, & Schiller. (2011). Lipid analysis by thin-layer chromatography—A review of the current state. Journal of Chromatography A, 1218(19), 2754-2774. doi:https://doi.org/10.1016/j.chroma.2010.11.066

Gaidhani, Harwalkar, Bhambere, & Nirgude. (2015). Lyophilization/freeze drying–a review. World journal of pharmaceutical research, 4(8), 516-543.

Garcia-Amezquita, Welti-Chanes, Vergara-Balderas, & Bermúdez-Aguirre. (2016). Freeze-drying: The Basic Process. In B. Caballero, P. M. Finglas, & F. Toldrá (Eds.), Encyclopedia of Food and Health (pp. 104-109). Oxford: Academic Press.

Gaur, Shivhare, & Ahmed. (2006). Degradation of chlorophyll during processing of green vegetables: a review. Stewart Postharvest Review, 2(5), 1-8.

Gaur, Shivhare, Sarkar, & Ahmed. (2007). Thermal Chlorophyll Degradation Kinetics of Mint Leaves Puree. International Journal of Food Properties, 10(4), 853-865. doi:10.1080/10942910601136450 Gaziano. (2004). Vitamin E and cardiovascular disease: observational studies. Annals of the New York Academy of Sciences, 1031(1), 280-291.

Gedi, Briars, Yuseli, Zainol, Darwish, Salter, & Gray. (2017). Component analysis of nutritionally rich chloroplasts: recovery from conventional and unconventional green plant species. J Food Sci Technol, 54(9), 2746-2757. doi:10.1007/s13197-017-2711-8

Gedi, Magee, Darwish, Eakpetch, Young, & Gray. (2019). Impact of the partial replacement of fish meal with a chloroplast rich fraction on the growth and selected nutrient profile of zebrafish (Danio rerio). Food & Function, 10(2), 733-745. doi:10.1039/C8FO02109K

Ghosh, Sarkar, Das, & Chakraborty. (2022). Natural colorants from plant pigments and their encapsulation: An emerging window for the food industry. LWT, 153, 112527. doi:https://doi.org/10.1016/j.lwt.2021.112527

Gökmen, Savaş Bahçeci, Serpen, & Acar. (2005). Study of lipoxygenase and peroxidase as

blanching indicator enzymes in peas: change of enzyme activity, ascorbic acid and chlorophylls during frozen storage. LWT - Food Science and Technology, 38(8), 903-908. doi:https://doi.org/10.1016/j.lwt.2004.06.018

Gomes, & Sobral. (2021). Plant protein-based delivery systems: an emerging approach for increasing the efficacy of lipophilic bioactive compounds. Molecules, 27(1), 60.

Gonçalves, Pinheiro, Abreu, Brandão, & Silva. (2010). Carrot (Daucus carota L.) peroxidase inactivation, phenolic content and physical changes kinetics due to blanching. Journal of Food Engineering, 97(4), 574-581.

González-Peña, Ortega-Regules, Anaya de Parrodi, & Lozada-Ramírez. (2023). Chemistry, Occurrence, Properties, Applications, and Encapsulation of Carotenoids-A Review. Plants (Basel), 12(2). doi:10.3390/plants12020313

Goodstine, Zheng, Holford, Ward, Carter, Owens, & Mayne. (2003). Dietary (n-3)/(n-6) Fatty Acid Ratio: Possible Relationship to Premenopausal but Not Postmenopausal Breast Cancer Risk in U.S. Women. The Journal of Nutrition, 133(5), 1409-1414. doi:10.1093/jn/133.5.1409

Goula, & Adamopoulos. (2005). Stability of lycopene during spray drying of tomato pulp. LWT - Food Science and Technology, 38(5), 479-487. doi:https://doi.org/10.1016/j.lwt.2004.07.020

Gounaris, & Barber. (1983). Monogalactosyldiacylglycerol: The most abundant polar lipid in nature. Trends in Biochemical Sciences, 8(10), 378-381. doi:https://doi.org/10.1016/0968-0004(83)90366-3

Granado-Lorencio, Olmedilla-Alonso, Herrero-Barbudo, Blanco-Navarro, Pérez-Sacristán, & Blázquez-García. (2007). In vitro bioaccessibility of carotenoids and tocopherols from fruits and vegetables. Food Chemistry, 102, 641-648. doi:10.1016/j.foodchem.2006.05.043

Grant, Williams, Rowe, & Muir. (2005). Efficacy of various pasteurization time-temperature conditions in combination with homogenization on inactivation of Mycobacterium avium subsp. paratuberculosis in milk. Applied and Environmental Microbiology, 71(6), 2853-2861.

Grimm. (2001). Chlorophyll: Structure and Function. In eLS.

Gross. (1991). Pigments in Vegetables. New York: Springer US.

Grune, Lietz, Palou, Ross, Stahl, Tang, . . . Biesalski. (2010). Beta-carotene is an important vitamin A source for humans. J Nutr, 140(12), 2268s-2285s. doi:10.3945/jn.109.119024

Grzeszczuk, Jadczak, & Podsiadło. (2007). The effect of blanching, freezing and freeze-storage on changes of some chemical compounds content in New Zealand spinach. Journal of Fruit and Ornamental Plant Research, 66(1), 95-103.

Gul, Tak, Singh, Singh, Yousuf, & Wani. (2015). Chemistry, encapsulation, and health benefits of βcarotene - A review. Cogent Food & Agriculture, 1(1), 1018696. doi:10.1080/23311932.2015.1018696

Guo, Ye, Bellissimo, Singh, & Rousseau. (2017). Modulating fat digestion through food structure design. Progress in Lipid Research, 68, 109-118.

Gupte, EL-BISI, & FRANCIS. (1964). Kinetics of Thermal Degradation of Chlorophyll in Spinach Pureea. Journal of Food Science, 29(4), 379-382. doi:https://doi.org/10.1111/j.1365-2621.1964.tb01747.x

Gut, & Matile. (1988). Apparent induction of key enzymes of the glyoxylic acid cycle in senescent barley leaves. Planta, 176(4), 548-550. doi:10.1007/bf00397663

Gyamfi, Awuah, & Owusu. (2019). Lipid metabolism: an overview. The molecular nutrition of fats, 17-32.

Haas, Obernberger, Zehetner, Kiesslich, Volkert, & Jaeger. (2019). Impact of powder particle structure on the oxidation stability and color of encapsulated crystalline and emulsified carotenoids in carrot concentrate powders. Journal of Food Engineering, 263, 398-408. doi:https://doi.org/10.1016/j.jfoodeng.2019.07.025

Hall, & Rao. (1999). Photosynthesis: Cambridge University Press.

Hamdan, Lee, Wong, Fauzi, Zamri, & Lee. (2022). Prevention of Enzymatic Browning by Natural Extracts and Genome-Editing: A Review on Recent Progress. Molecules, 27(3). doi:10.3390/molecules27031101

Harhar, Gharby, Kartah, El Monfalouti, Guillaume, & Charrouf. (2011). Influence of Argan Kernel Roasting-time on Virgin Argan Oil Composition and Oxidative Stability. Plant Foods for Human Nutrition, 66(2), 163-168. doi:10.1007/s11130-011-0220-x

He, & Luo. (2007). Enzymatic browning and its control in fresh-cut produce.

Hedren, Diaz, & Svanberg. (2002). Estimation of carotenoid accessibility from carrots determined by an in vitro digestion method. European journal of clinical nutrition, 56(5), 425-430.

Heinz. (1973). Some properties of the acyl galactoside-forming enzyme from leaves. Pflanzenphysiol, 69.

Heinz, Bertrams, Joyard, & Douce. (1978). Demonstration of an Acyltransferase Activity in Chloroplast Envelopes. Zeitschrift für Pflanzenphysiologie, 87(4), 325-331. doi:https://doi.org/10.1016/S0044-328X(78)80278-5

Hiranvarachat, Suvarnakuta, & Devahastin. (2008). Isomerisation kinetics and antioxidant activities of β -carotene in carrots undergoing different drying techniques and conditions. Food Chemistry, 107(4), 1538-1546. doi:https://doi.org/10.1016/j.foodchem.2007.10.026

Hirayama, & Oido. (1969). Changes of Lipid and Pigment Compositions in Spinach Leaves during Their Storage. Nippon Nōgeikagaku Kaishi, 43(7), 423-428. doi:10.1271/nogeikagaku1924.43.423 Hirschberg. (1999). Production of high-value compounds: carotenoids and vitamin E. Current Opinion in Biotechnology, 10(2), 186-191.

Hodgson, & Rose. (1984). Fusion of Spinach Mesophyll Protoplasts with Carrot Root Parenchyma Protoplasts and the Effect on Spinach Chloroplasts. Journal of plant physiology, 115, 69-78. doi:10.1016/S0176-1617(84)80052-8

Hölzl, & Dörmann. (2019). Chloroplast Lipids and Their Biosynthesis. Annual Review of Plant Biology, 70(1), 51-81. doi:10.1146/annurev-arplant-050718-100202

Hölzl, Witt, Kelly, Zähringer, Warnecke, Dörmann, & Heinz. (2006). Functional differences between galactolipids and glucolipids revealed in photosynthesis of higher plants. Proceedings of the National Academy of Sciences, 103(19), 7512-7517. doi:doi:10.1073/pnas.0600525103

Hornero-Méndez, & Mínguez-Mosquera. (2007). Bioaccessibility of carotenes from carrots: Effect of cooking and addition of oil. Innovative Food Science & Emerging Technologies, 8(3), 407-412. Hoshina, Tomita, & Shioi. (1998). Antioxidant Activity of Chlorophylls: Its Structure-Activity Relationship. In G. Garab (Ed.), Photosynthesis: Mechanisms and Effects: Volume I–V: Proceedings of the XIth International Congress on Photosynthesis, Budapest, Hungary, August 17–22, 1998 (pp. 3281-3284). Dordrecht: Springer Netherlands.

Hu, Chen, & Ni. (2012). Effect of superfine grinding on quality and antioxidant property of fine green tea powders. LWT - Food Science and Technology, 45(1), 8-12. doi:https://doi.org/10.1016/j.lwt.2011.08.002

Hur, Lim, Decker, & McClements. (2011). In vitro human digestion models for food applications. Food Chemistry, 125(1), 1-12.

Hussain, & Khushnuma Maqbool. (2014). GC-MS: principle, technique and its application in food science.

Hussain, Maqbool, & Naseer. (2019). High performance thin layer chromatography: Principle, working and applications. International Journal of Pharmacy and Pharmaceutical Sciences, 4, 83-88.

Isengard. (2001). Water content, one of the most important properties of food. Food Control, 12(7), 395-400. doi:https://doi.org/10.1016/S0956-7135(01)00043-3

Ishida, Izumi, Wada, & Makino. (2014). Roles of autophagy in chloroplast recycling. Biochimica et Biophysica Acta (BBA)-Bioenergetics, 1837(4), 512-521.

Jaeger, Janositz, & Knorr. (2010). The Maillard reaction and its control during food processing. The potential of emerging technologies. Pathologie Biologie, 58(3), 207-213. doi:https://doi.org/10.1016/j.patbio.2009.09.016

Jaya, & Das. (2004). Effect of maltodextrin, glycerol monostearate and tricalcium phosphate on vacuum dried mango powder properties. Journal of Food Engineering, 63(2), 125-134.

Jayathunge, Stratakos, Cregenzán-Albertia, Grant, Lyng, & Koidis. (2017). Enhancing the lycopene in vitro bioaccessibility of tomato juice synergistically applying thermal and non-thermal processing technologies. Food Chemistry, 221, 698-705.

Jensen, & Leister. (2014). Chloroplast evolution, structure and functions. F1000Prime Rep, 6, 40. doi:10.12703/p6-40

Jinapong, Suphantharika, & Jamnong. (2008). Production of instant soymilk powders by ultrafiltration, spray drying and fluidized bed agglomeration. Journal of Food Engineering, 84(2), 194-205. doi:https://doi.org/10.1016/j.jfoodeng.2007.04.032

Jonsson. (1991). Thermal degradation of carotenes and influence on their physiological functions. Adv Exp Med Biol, 289, 75-82. doi:10.1007/978-1-4899-2626-5_6

Joseph, Shukitt-Hale, Denisova, Bielinski, Martin, McEwen, & Bickford. (1999). Reversals of Age-Related Declines in Neuronal Signal Transduction, Cognitive, and Motor Behavioral Deficits with Blueberry, Spinach, or Strawberry Dietary Supplementation. The Journal of Neuroscience, 19(18), 8114. doi:10.1523/JNEUROSCI.19-18-08114.1999

Joshi, Kumari, Chauhan, & Singh. (2021). Development of water spinach powder and its characterization. Journal of Food Science and Technology, 58(9), 3533-3539. doi:10.1007/s13197-021-05058-9

Joyard, Teyssier, Miege, Berny-Seigneurin, Maréchal, Block, . . . Douce. (1998). The biochemical machinery of plastid envelope membranes. Plant physiology, 118(3), 715-723.

Kaddumukasa, Imathiu, Mathara, & Nakavuma. (2017). Influence of physicochemical parameters on storage stability: Microbiological quality of fresh unpasteurized fruit juices. Food Sci Nutr, 5(6), 1098-1105. doi:10.1002/fsn3.500

Kaniuga. (2008). Chilling response of plants: importance of galactolipase, free fatty acids and free

radicals*. Plant Biology, 10(2), 171-184. doi:https://doi.org/10.1111/j.1438-8677.2007.00019.x Kannan. (2018). Scanning electron microscopy: Principle, components and applications. A textbook on fundamentals and applications of nanotechnology, 81-92.

Kaptay. (2012). On the size and shape dependence of the solubility of nano-particles in solutions. International journal of pharmaceutics, 430(1-2), 253-257.

Karaaslan, & Tunçer. (2008). Development of a drying model for combined microwave–fanassisted convection drying of spinach. Biosystems Engineering, 100(1), 44-52. doi:https://doi.org/10.1016/j.biosystemseng.2007.12.012

Karim, & Aronsson. (2014). The puzzle of chloroplast vesicle transport - involvement of GTPases. Front Plant Sci, 5, 472. doi:10.3389/fpls.2014.00472

Katouzian, & Jafari. (2016). Nano-encapsulation as a promising approach for targeted delivery and controlled release of vitamins. Trends in Food Science & Technology, 53, 34-48. doi:https://doi.org/10.1016/j.tifs.2016.05.002

Kergomard, Carrière, Paboeuf, Chonchon, Barouh, Vié, & Bourlieu. (2023). Interfacial adsorption and activity of pancreatic lipase-related protein 2 onto heterogeneous plant lipid model membranes. Biochimie. doi:https://doi.org/10.1016/j.biochi.2023.04.001

Khan, Mahesh, Semwal, & Sharma. (2015). Effect of spinach powder on physico-chemical, rheological, nutritional and sensory characteristics of chapati premixes. Journal of Food Science and Technology, 52(4), 2359-2365.

Khandpur, & Gogate. (2016). Evaluation of ultrasound based sterilization approaches in terms of shelf life and quality parameters of fruit and vegetable juices. Ultrasonics Sonochemistry, 29, 337-353. doi:https://doi.org/10.1016/j.ultsonch.2015.10.008

Kidmose, Knuthsen, Edelenbos, Justesen, & Hegelund. (2001). Carotenoids and flavonoids in organically grown spinach (Spinacia oleracea L) genotypes after deep frozen storage. Journal of the Science of Food and Agriculture, 81(9), 918-923. doi:10.1002/jsfa.902

Kim, Shang, Assefa, Keum, & Saini. (2018). Metabolite profiling of green, green/red, and red lettuce cultivars: Variation in health beneficial compounds and antioxidant potential. Food Research International, 105, 361-370. doi:https://doi.org/10.1016/j.foodres.2017.11.028

Kirchhoff, Mukherjee, & Galla. (2002). Molecular Architecture of the Thylakoid Membrane: Lipid Diffusion Space for Plastoquinone. Biochemistry, 41(15), 4872-4882. doi:10.1021/bi011650y

Klockow, & Keener. (2009). Safety and quality assessment of packaged spinach treated with a novel ozone-generation system. LWT - Food Science and Technology, 42(6), 1047-1053. doi:https://doi.org/10.1016/j.lwt.2009.02.011

Knecht, Sandfuchs, Kulling, & Bunzel. (2015). Tocopherol and tocotrienol analysis in raw and cooked vegetables: a validated method with emphasis on sample preparation. Food Chem, 169, 20-27. doi:10.1016/j.foodchem.2014.07.099

Knockaert, Lemmens, Van Buggenhout, Hendrickx, & Van Loey. (2012). Changes in β-carotene bioaccessibility and concentration during processing of carrot puree. Food Chemistry, 133(1), 60-67. doi:https://doi.org/10.1016/j.foodchem.2011.12.066

Kobayashi. (2016). Role of membrane glycerolipids in photosynthesis, thylakoid biogenesis and chloroplast development. Journal of Plant Research, 129(4), 565-580. doi:10.1007/s10265-016-0827-y

Koç, & Dirim. (2018). Spray dried spinach juice: powder properties. Journal of Food Measurement and Characterization, 12(3), 1654-1668. doi:10.1007/s11694-018-9781-9

Kopas-Lane, & Warthesen. (1995). Carotenoid photostability in raw spinach and carrots during cold storage. Journal of Food Science, 60(4), 773-776.

Kopec, & Failla. (2018). Recent advances in the bioaccessibility and bioavailability of carotenoids and effects of other dietary lipophiles. Journal of Food Composition and Analysis, 68, 16-30.

Kowdley, Mason, Meydani, Cornwall, & Grand. (1992). Vitamin E deficiency and impaired cellular immunity related to intestinal fat malabsorption. Gastroenterology, 102(6), 2139-2142.

Krupinska, & Humbeck. (2004). Photosynthesis and chloroplast breakdown. In Plant cell death processes (pp. 169-187): Elsevier.

Kuck, & Noreña. (2016). Microencapsulation of grape (Vitis labrusca var. Bordo) skin phenolic extract using gum Arabic, polydextrose, and partially hydrolyzed guar gum as encapsulating agents. Food Chemistry, 194, 569-576. doi:https://doi.org/10.1016/j.foodchem.2015.08.066

Kumar, Devasagayam, Bhushan, & Verma. (2001). Scavenging of reactive oxygen species by chlorophyllin: an ESR study. Free Radical Research, 35(5), 563-574.

Kung, Tung, & Maciorowski. (1991). Effect of a microbial inoculant (Ecosyl[™] and/or a glycopeptide antibiotic (vancomycin) on fermentation and aerobic stability of wilted alfalfa silage. Animal Feed Science and Technology, 35(1), 37-48. doi:https://doi.org/10.1016/0377-8401(91)90097-C

Laokuldilok, & Kanha. (2015). Effects of processing conditions on powder properties of black glutinous rice (Oryza sativa L.) bran anthocyanins produced by spray drying and freeze drying. LWT - Food Science and Technology, 64(1), 405-411. doi:https://doi.org/10.1016/j.lwt.2015.05.015

Lasya. (2022). Spinach and its health benefits: A review.

Lee, Cha, Vuong, Kim, & Pan. (2018a). Comparison of static and dynamic in vitro digestion models to estimate the bioaccessibility of lutein in lutein-rich foods. Applied Biological Chemistry, 61(4), 441-447.

Lee, Choi, Jeong, Lee, & Sung. (2018b). Effect of different cooking methods on the content of vitamins and true retention in selected vegetables. Food Sci Biotechnol, 27(2), 333-342. doi:10.1007/s10068-017-0281-1

Lee, Taip, & Abdullah. (2018c). Effectiveness of additives in spray drying performance: a review. Food Research, 2, 486-499. doi:10.26656/fr.2017.2(6).134

Leistner, & Gorris. (1994). Food Preservation by Combined Processes: Final Report: FLAIR Concerted Action No. 7, Subgroup B: European Commission, Directorate-General XII, Science, Research and Development.

Lemmens, Colle, Van Buggenhout, Palmero, Van Loey, & Hendrickx. (2014). Carotenoid bioaccessibility in fruit- and vegetable-based food products as affected by product (micro)structural characteristics and the presence of lipids: A review. Trends in Food Science & Technology, 38(2), 125-135. doi:https://doi.org/10.1016/j.tifs.2014.05.005

Lemmens, Colle, Van Buggenhout, Van Loey, & Hendrickx. (2011a). Quantifying the influence of thermal process parameters on in vitro β -carotene bioaccessibility: a case study on carrots. Journal of Agricultural and Food Chemistry, 59(7), 3162-3167.

Lemmens, Colle, Van Buggenhout, Van Loey, & Hendrickx. (2011b). Quantifying the Influence of

Thermal Process Parameters on in Vitro β -Carotene Bioaccessibility: A Case Study on Carrots. Journal of Agricultural and Food Chemistry, 59(7), 3162-3167. doi:10.1021/jf104888y

Lemmens, De Vleeschouwer, Moelants, Colle, Van Loey, & Hendrickx. (2010). β-Carotene Isomerization Kinetics during Thermal Treatments of Carrot Puree. Journal of Agricultural and Food Chemistry, 58(11), 6816-6824. doi:10.1021/jf100449t

Lešková, Kubíková, Kováčiková, Košická, Porubská, & Holčíková. (2006). Vitamin losses: Retention during heat treatment and continual changes expressed by mathematical models. Journal of Food Composition and Analysis, 19(4), 252-276.

Li, Cui, Lu, Wang, Liao, Hu, & Zhang. (2019a). Beneficial effects of a chlorophyll-rich spinach extract supplementation on prevention of obesity and modulation of gut microbiota in high-fat diet-fed mice. Journal of Functional Foods, 60, 103436. doi:https://doi.org/10.1016/j.jff.2019.103436

LI, MIAO, & JIANG. (2009). CHANGES IN QUALITY ATTRIBUTES OF LONGAN JUICE DURING STORAGE IN RELATION TO EFFECTS OF THERMAL PROCESSING. Journal of Food Quality, 32(1), 48-57. doi:https://doi.org/10.1111/j.1745-4557.2008.00235.x

Li, Tsao, & Deng. (2012). Factors affecting the antioxidant potential and health benefits of plant foods. Canadian journal of plant science, 92(6), 1101-1111.

Li, & Yu. (2018). Chloroplast Galactolipids: The Link Between Photosynthesis, Chloroplast Shape, Jasmonates, Phosphate Starvation and Freezing Tolerance. Plant and Cell Physiology, 59(6), 1128-1134. doi:10.1093/pcp/pcy088

Li, Yu, Dhital, Gidley, & Gilbert. (2019b). Starch branching enzymes contributing to amylose and amylopectin fine structure in wheat. Carbohydrate Polymers, 224, 115185. doi:https://doi.org/10.1016/j.carbpol.2019.115185

Liang, Urano, Liao, Hedrick, Gao, & Jones. (2017). A nondestructive method to estimate the chlorophyll content of Arabidopsis seedlings. Plant Methods, 13, 26. doi:10.1186/s13007-017-0174-6

Lichtenthaler, & Buschmann. (2001). Chlorophylls and Carotenoids: Measurement and Characterization by UV-VIS Spectroscopy. Current Protocols in Food Analytical Chemistry, 1(1), F4.3.1-F4.3.8. doi:10.1002/0471142913.faf0403s01

Lin, & Schyvens. (1995). Influence of blanching treatments on the texture and color of some processed vegetables and fruits. Journal of food processing and preservation, 19(6), 451-465.

Lippman, Klein, Goodman, Lucia, Thompson, Ford, . . . Hartline. (2009). Effect of selenium and vitamin E on risk of prostate cancer and other cancers: the Selenium and Vitamin E Cancer Prevention Trial (SELECT). Jama, 301(1), 39-51.

Liu, Hou, Yang, & Gao. (2015). Effects of antioxidants on the stability of β -Carotene in O/W emulsions stabilized by Gum Arabic. J Food Sci Technol, 52(6), 3300-3311. doi:10.1007/s13197-014-1380-0

Liu, Wang, Cao, & Liu. (2018). Combination of Gas Chromatography-Mass Spectrometry and Electron Spin Resonance Spectroscopy for Analysis of Oxidative Stability in Soybean Oil During Deep-Frying Process. Food Analytical Methods, 11(5), 1485-1492. doi:10.1007/s12161-017-1132-7

Liu, Zhang, & Hu. (2022). High efficient freeze-drying technology in food industry. Critical Reviews in Food Science and Nutrition, 62(12), 3370-3388. doi:10.1080/10408398.2020.1865261

Lokstein, Renger, & Götze. (2021). Photosynthetic Light-Harvesting (Antenna) Complexes-Structures and Functions. Molecules, 26(11). doi:10.3390/molecules26113378

Lopez, Sala, de la Fuente, Condon, Raso, & Burgos. (1994). Inactivation of peroxidase, lipoxygenase, and polyphenol oxidase by manothermosonication. Journal of Agricultural and Food Chemistry, 42(2), 252-256. doi:10.1021/jf00038a005

López-Gámez, Elez-Martínez, Martín-Belloso, & Soliva-Fortuny. (2021). Recent Advances toward the Application of Non-Thermal Technologies in Food Processing: An Insight on the Bioaccessibility of Health-Related Constituents in Plant-Based Products. Foods, 10(7). doi:10.3390/foods10071538

Lorenzo, Estévez, Barba, Thirumdas, Franco, & Munekata. (2019). Polyphenols: Bioaccessibility and bioavailability of bioactive components. In Innovative thermal and non-thermal processing, bioaccessibility and bioavailability of nutrients and bioactive Compounds (pp. 309-332): Elsevier.

Lu, Maidannyk, & Lim. (2020). Carotenoids degradation and precautions during processing. In Carotenoids: Properties, processing and applications (pp. 223-258): Elsevier.

Lucas, Zambiazi, & Costa. (2018). Biocompounds and physical properties of açaí pulp dried by different methods. LWT, 98, 335-340. doi:https://doi.org/10.1016/j.lwt.2018.08.058

Lund, Karmas, & Harris. (1988). Nutritional evaluation of food processing. In: Van Nostrand Reinhold Co., New York.

Lynch, & Barbano. (2020). Kjeldahl Nitrogen Analysis as a Reference Method for Protein Determination in Dairy Products. Journal of AOAC INTERNATIONAL, 82(6), 1389-1398. doi:10.1093/jaoac/82.6.1389

Lyu, Bi, Chen, Wu, Qiao, Hou, & Zhang. (2021). Bioaccessibility of carotenoids and antioxidant capacity of seed-used pumpkin byproducts powders as affected by particle size and corn oil during in vitro digestion process. Food Chemistry, 343, 128541. doi:https://doi.org/10.1016/j.foodchem.2020.128541

Maas, Schaldach, Littringer, Mescher, Griesser, Braun, . . . Urbanetz. (2011). The impact of spray drying outlet temperature on the particle morphology of mannitol. Powder Technology, 213(1), 27-35. doi:https://doi.org/10.1016/j.powtec.2011.06.024

MacLean, Newberry, Mojica, Khanna, Issa, Suttorp, . . . Morton. (2006). Effects of Omega-3 Fatty Acids on Cancer RiskA Systematic Review. JAMA, 295(4), 403-415. doi:10.1001/jama.295.4.403

Maesmans, Hendrickx, Weng, Keteleer, & Tobback. (1990). Endpoint definition, determination and evaluation of thermal processes in food preservation. Belgian journal of food chemistry and biotechnology, 45(5), 179-192.

Makshakova, Breton, & Perez. (2020). Unraveling the complex enzymatic machinery making a key galactolipid in chloroplast membrane: a multiscale computer simulation. Scientific Reports, 10(1), 13514. doi:10.1038/s41598-020-70425-z

Mangos, & Berger. (1997). Determination of major chlorophyll degradation products. Zeitschrift für Lebensmitteluntersuchung und -Forschung A, 204(5), 345-350. doi:10.1007/s002170050088 Mansor. (2019). Process optimisation, nutrient stability and digestibility of green powder from spinach juice. University of Nottingham,

Manzocco, Calligaris, Mastrocola, Nicoli, & Lerici. (2000). Review of non-enzymatic browning and antioxidant capacity in processed foods. Trends in Food Science & Technology, 11(9), 340-346.

doi:https://doi.org/10.1016/S0924-2244(01)00014-0

Maoka. (2020). Carotenoids as natural functional pigments. J Nat Med, 74(1), 1-16. doi:10.1007/s11418-019-01364-x

Marchand, Heydarizadeh, Schoefs, & Spetea. (2018). Ion and metabolite transport in the chloroplast of algae: lessons from land plants. Cellular and Molecular Life Sciences, 75(12), 2153-2176. doi:10.1007/s00018-018-2793-0

Marques, Silveira, & Freire. (2006). Freeze-Drying Characteristics of Tropical Fruits. Drying Technology, 24(4), 457-463. doi:10.1080/07373930600611919

Martínez, Pérez, Carballo, & Franco. (2013). Effect of blanching methods and frozen storage on some quality parameters of turnip greens ("grelos"). LWT - Food Science and Technology, 51(1), 383-392. doi:https://doi.org/10.1016/j.lwt.2012.09.020

Martins, & Silva. (2002). Modelling colour and chlorophyll losses of frozen green beans (Phaseolus vulgaris, L.). International Journal of Refrigeration, 25(7), 966-974. doi:https://doi.org/10.1016/S0140-7007(01)00050-0

Marx, Stuparic, Schieber, & Carle. (2003). Effects of thermal processing on trans–cis-isomerization of β -carotene in carrot juices and carotene-containing preparations. Food Chemistry, 83(4), 609-617. doi:https://doi.org/10.1016/S0308-8146(03)00255-3

Mathlouthi. (2001). Water content, water activity, water structure and the stability of foodstuffs. Food Control, 12(7), 409-417. doi:https://doi.org/10.1016/S0956-7135(01)00032-9

Matsuzaki, Koiwai, Kawashima, & Matsuyama. (1982). Formation of Acylmonogalactosyldiglyceride and Degradation of Polar Lipids in Leaf Homogenates of Tobacco. Agricultural and Biological Chemistry, 46(3), 723-729. doi:10.1080/00021369.1982.10865124

Mauzerall. (1976). Chlorophyll and photosynthesis. Philos. Trans. R. Soc., B, 273, 287-294.

McGlynn. (2003). Importance of food ph in commerical canning operations. Retrieved from Mdoda, Obi, Ncoyini-Manciya, Christian, & Mayekiso. (2022). Assessment of Profit Efficiency for Spinach Production under Small-Scale Irrigated Agriculture in the Eastern Cape Province, South Africa. Sustainability, 14(5), 2991. Retrieved from https://www.mdpi.com/2071-1050/14/5/2991

Mehta. (2015). Nutritional and Toxicological Aspects of the Chemical Changes of Food Components and Nutrients During Heating and Cooking. In P. C. K. Cheung & B. M. Mehta (Eds.), Handbook of Food Chemistry (pp. 897-936). Berlin, Heidelberg: Springer Berlin Heidelberg.

Mengucci, Ferranti, Romano, Masi, Picone, & Capozzi. (2022). Food structure, function and artificial intelligence. Trends in Food Science & Technology, 123, 251-263. doi:https://doi.org/10.1016/j.tifs.2022.03.015

Meydani, & Beharka. (1998). Recent developments in vitamin E and immune response. Nutrition Reviews, 56(1), S49.

Michalski, & Kaniuga. (1980). Photosynthetic apparatus in chilling-sensitive plants. VII. Comparison of the effect of galactolipase treatment of chloroplasts and cold-dark storage of leaves on photosynthetic electron flow. Biochimica et biophysica acta, 589 1, 84-99.

Milani, Basirnejad, Shahbazi, & Bolhassani. (2017). Carotenoids: biochemistry, pharmacology and treatment. Br J Pharmacol, 174(11), 1290-1324. doi:10.1111/bph.13625

Minekus, Alminger, Alvito, Ballance, Bohn, Bourlieu, . . . Dupont. (2014). A standardised static in vitro digestion method suitable for food–an international consensus. Food & Function, 5(6), 1113-

1124.

92, 275-287.

Minguez-Mosquera, Garrido-Fernandez, & Gandul-Rojas. (1989). Pigment changes in olives during fermentation and brine storage. Journal of Agricultural and Food Chemistry, 37(1), 8-11. Miqyass, Van Gorkom, & Yocum. (2007). The PSII calcium site revisited. Photosynthesis Research,

Mishra, Bacheti, & Husen. (2011). Medicinal uses of chlorophyll: A critical overview. Chlorophyll: Structure, function and medicinal uses, 177-196.

Mitra, Rauf, Tareq, Jahan, Emran, Shahriar, . . . Rebezov. (2021). Potential health benefits of carotenoid lutein: An updated review. Food and Chemical Toxicology, 154, 112328.

Mitra, Shrivastava, & Rao. (2015). Non-enzymatic browning and flavour kinetics of vacuum dried onion slices. International Agrophysics, 29(1).

Miyazawa, Nakagawa, & Sookwong. (2011). Health benefits of vitamin E in grains, cereals and green vegetables. Trends in Food Science & Technology, 22(12), 651-654.

Molisch. (1886). Zwei neue Zuckerreactionen. Monatshefte für Chemie und verwandte Teile anderer Wissenschaften, 7(1), 198-209. doi:10.1007/BF01516570

Morita, Kuroiwa, Kuroiwa, & Nozaki. (2008). High localization of ribulose-1,5-bisphosphate carboxylase/oxygenase in the pyrenoids of Chlamydomonas reinhardtii (Chlorophyta), as revealed by cryofixation and immunogold electron microscopy. Journal of Phycology, 33, 68-72. doi:10.1111/j.0022-3646.1997.00068.x

Mounir, Mujumdar, Zhang, Bhandari, & Fang. (2017). Advances in Drying Science and Technology: Handbook of Drying of Vegetables and Vegetable Products.

Mujica-Álvarez, Gil-Castell, Barra, Ribes-Greus, Bustos, Faccini, & Matiacevich. (2020). Encapsulation of vitamins A and E as spray-dried additives for the feed industry. Molecules, 25(6), 1357.

Mujumdar, & Menon. (2020). Drying of solids: principles, classification, and selection of dryers. In Handbook of industrial drying (pp. 1-39): CRC Press.

Mukherjee, & Chattopadhyay. (2007). Whirling bed blanching of potato cubes and its effects on product quality. Journal of Food Engineering, 78(1), 52-60.

Müller, & Krawinkel. (2005). Malnutrition and health in developing countries. Cmaj, 173(3), 279-286. doi:10.1503/cmaj.050342

Murcia, López-Ayerra, & García-Carmona. (1999). Effect of processing methods and different blanching times on broccoli: proximate composition and fatty acids. LWT-Food Science and Technology, 32(4), 238-243.

Mythili, Rajeswari, Bosco, & Kamatchi alias Rajalechumi. (2021). Impact of blanching treatments on the chemical composition, total dietary fiber, physicochemical, functional, and structural properties of underutilized cauliflower leaves (Brassica oleracea var. botrytis). Journal of food processing and preservation, 45(11), e15910. doi:https://doi.org/10.1111/jfpp.15910

N'Goma. (2012). Understanding the lipid-digestion processes in the GI tract before designing lipid-based drug-delivery systems. Therapeutic Delivery, 3(1), 105-124.

Nagao, Kotake-Nara, & Hase. (2013). Effects of Fats and Oils on the Bioaccessibility of Carotenoids and Vitamin E in Vegetables. Bioscience, Biotechnology, and Biochemistry, 77(5), 1055-1060. doi:10.1271/bbb.130025

Negri Rodríguez, Arias, Soteras, Sancho, Tacca, Aimaretti, . . . Szerman. (2021). Comparison of the quality attributes of carrot juice pasteurized by ohmic heating and conventional heat treatment. LWT, 111255. doi:https://doi.org/10.1016/j.lwt.2021.111255

Nicholls, & Ferguson. (2013). 1 - Chemiosmotic Energy Transduction. In D. G. Nicholls & S. J. Ferguson (Eds.), Bioenergetics (Fourth Edition) (pp. 3-12). Boston: Academic Press.

Nijdam, & Langrish. (2006). The effect of surface composition on the functional properties of milk powders. Journal of Food Engineering, 77(4), 919-925. doi:https://doi.org/10.1016/j.jfoodeng.2005.08.020

Nilsson, Johansson, Fahlberg, Kommuri, Töpel, Bodin, . . . Andersson. (2015). Acylated monogalactosyl diacylglycerol: prevalence in the plant kingdom and identification of an enzyme catalyzing galactolipid head group acylation in Arabidopsis thaliana. Plant J, 84(6), 1152-1166. doi:10.1111/tpj.13072

Nireesha, Divya, Sowmya, Venkateshan, Babu, & Vuppalapati. (2013). Lyophilization/Freeze Drying-An Review. 3, 87-98.

Njoroge, Matofari, Mulwa, & Anyango. (2015). Effects of blanching time/temperature combination coupled with solar-drying on the nutritional and microbial quality of indigenous leafy vegetables in Kenya. African Journal of Food Science and Technology, 6(7), 209-219.

Nowak, & Jakubczyk. (2020). The Freeze-Drying of Foods—The Characteristic of the Process Course and the Effect of Its Parameters on the Physical Properties of Food Materials. Foods, 9(10), 1488. Retrieved from https://www.mdpi.com/2304-8158/9/10/1488

Okoli, Nmorka, & Unaegbu. (1988). Technical Note: Blanching and storage of some Nigerian vegetables. International Journal of Food Science & Technology, 23(6), 639-641. doi:https://doi.org/10.1111/j.1365-2621.1988.tb01050.x

Okpala, & Ekechi. (2014). Rehydration characteristics of dehydrated West African pepper (Piper guineense) leaves. Food Sci Nutr, 2(6), 664-668. doi:10.1002/fsn3.149

Ölmez, & Kretzschmar. (2009). Potential alternative disinfection methods for organic fresh-cut industry for minimizing water consumption and environmental impact. LWT - Food Science and Technology, 42(3), 686-693. doi:https://doi.org/10.1016/j.lwt.2008.08.001

Onayemi, & Badifu. (1987). Effect of blanching and drying methods on the nutritional and sensory quality of leafy vegetables. Plant Foods for Human Nutrition, 37(4), 291-298. doi:10.1007/BF01092204

Opdensteinen, Lobanov, & Buyel. (2021). A combined pH and temperature precipitation step facilitates the purification of tobacco-derived recombinant proteins that are sensitive to extremes of either parameter. Biotechnology Journal, 16(4), 2000340. doi:https://doi.org/10.1002/biot.202000340

Ozkan, Stübler, Aganovic, Draeger, Esatbeyoglu, & Capanoglu. (2022). A comparative study on physicochemical properties and in vitro bioaccessibility of bioactive compounds in rosehip (Rosa canina L.) infusions treated by non-thermal and thermal treatments. Journal of food processing and preservation, 46(6), e16096.

Pallares, Miranda, Truong, Kyomugasho, Chigwedere, Hendrickx, & Grauwet. (2018). Processinduced cell wall permeability modulates the in vitro starch digestion kinetics of common bean cotyledon cells. Food & Function, 9(12), 6544-6554. Palmero, Lemmens, Hendrickx, & Van Loey. (2014). Role of carotenoid type on the effect of thermal processing on bioaccessibility. Food Chemistry, 157, 275-282. doi:https://doi.org/10.1016/j.foodchem.2014.02.055

Pandrangi, & Laborde. (2004). Retention of Folate, Carotenoids, and Other Quality Characteristics in Commercially Packaged Fresh Spinach. Journal of Food Science, 69, C702-C707. doi:10.1111/j.1365-2621.2004.tb09919.x

Park, Imm, & Ku. (2001). Improved Dispersibility of Green Tea Powder by Microparticulation and Formulation. Journal of Food Science, 66(6), 793-798. doi:10.1111/j.1365-2621.2001.tb15174.x

Park, Kim, & Park. (2012). Antidiabetic activity of fruits and vegetables commonly consumed in Korea: Inhibitory potential against α -glucosidase and insulin-like action in vitro. Food Science and Biotechnology, 21(4), 1187-1193. doi:10.1007/s10068-012-0155-5

Patel, Patel, & Suthar. (2009). Spray drying technology: an overview. Indian Journal of Science and Technology, 2(10), 44-47.

Patil, Chauhan, & Singh. (2014). Optimization of the spray-drying process for developing guava powder using response surface methodology. Powder Technology, 253, 230-236.

Peng, Tang, Barrett, Sablani, Anderson, & Powers. (2017). Thermal pasteurization of ready-to-eat foods and vegetables: Critical factors for process design and effects on quality. Critical Reviews in Food Science and Nutrition, 57(14), 2970-2995. doi:10.1080/10408398.2015.1082126

Pénicaud, Achir, Dhuique-Mayer, Dornier, & Bohuon. (2011). Degradation of β -carotene during fruit and vegetable processing or storage: reaction mechanisms and kinetic aspects: a review. Fruits, 66(6), 417-440. doi:10.1051/fruits/2011058

Perez-gago, & Krochta. (2001). Denaturation Time and Temperature Effects on Solubility, Tensile Properties, and Oxygen Permeability of Whey Protein Edible Films. Journal of Food Science, 66(5), 705-710. doi:https://doi.org/10.1111/j.1365-2621.2001.tb04625.x

Pérez-Marín, Torres, Entrenas, Vega, & Sánchez. (2019). Pre-harvest screening on-vine of spinach quality and safety using NIRS technology. Spectrochim Acta A Mol Biomol Spectrosc, 207, 242-250. doi:10.1016/j.saa.2018.09.035

Perry, Rasmussen, & Johnson. (2009). Xanthophyll (lutein, zeaxanthin) content in fruits, vegetables and corn and egg products. Journal of Food Composition and Analysis, 22(1), 9-15.

Persson, Svensson, & Adlercreutz. (2000). Enzymatic fatty acid exchange in digalactosyldiacylglycerol. Chem Phys Lipids, 104(1), 13-21. doi:10.1016/s0009-3084(99)00099-7 Philippon. (1984). Revue bibliographique. Méthodes de blanchiment-refroidissement des légumes avant congélation. Sciences des aliments, 4(4), 523-550.

Phisut. (2012). Spray drying technique of fruit juice powder: some factors influencing the properties of product.

Platel, & Srinivasan. (2016). Bioavailability of micronutrients from plant foods: an update. Critical Reviews in Food Science and Nutrition, 56(10), 1608-1619.

Pribil, Labs, & Leister. (2014). Structure and dynamics of thylakoids in land plants. Journal of experimental botany, 65(8), 1955-1972.

Priyadarshani. (2017). A review on factors influencing bioaccessibility and bioefficacy of carotenoids. Crit Rev Food Sci Nutr, 57(8), 1710-1717. doi:10.1080/10408398.2015.1023431 Qian, Decker, Xiao, & McClements. (2012). Nanoemulsion delivery systems: Influence of carrier oil

on β -carotene bioaccessibility. Food Chemistry, 135(3), 1440-1447. doi:https://doi.org/10.1016/j.foodchem.2012.06.047

Quarcoo, & Wireko-Manu. (2016). The effect of steam and hot water blanching on some quality attributes of cocoyam leaf. MOJ Food Process Technology, 2(5), 164-168.

Quek, Chok, & Swedlund. (2007). The physicochemical properties of spray-dried watermelon powders. Chemical Engineering and Processing: Process Intensification, 46(5), 386-392. doi:https://doi.org/10.1016/j.cep.2006.06.020

Raines. (2003). The Calvin cycle revisited. Photosynthesis Research, 75(1), 1-10.

Rakus, Gizak, Kasprzak, Zarzycki, Maciaszczyk-Dziubinska, & Dzugaj. (2013). The mechanism of calcium-induced inhibition of muscle fructose 1,6-bisphosphatase and destabilization of glyconeogenic complex. PloS one, 8(10), e76669. doi:10.1371/journal.pone.0076669

Rambuda, Nkomo, Soundy, Araya, Ngezimana, & Mudau. (2015). Influence of Postharvest Storage Temperature and Duration on Quality of Baby Spinach (Vol. 25).

Ramesh. (2020). Pasteurization and food preservation. In Handbook of food preservation (pp. 599-608): CRC Press.

Ramesh, Wolf, Tevini, & Bognar. (2002). Microwave blanching of vegetables. Journal of Food Science, 67(1), 390-398.

Rani, Rahman, Younis, & Akhtar. (2015). HPTLC: An Emerging Technique in Herbal Medicine. In (Vol. 4, pp. 25-37p): Research and Reviews: Journal of Herbal Sciences (RRJoHS).

Rao, Muraleedharan, & Humphreys. (2010). TEM specimen preparation techniques. Microscopy: science, technology, applications and education, 2, 1232.

Rao, Nolasco, Handa, Naldrett, Alvarez, & Majumder. (2020). Effect of pH and Heat Treatment on the Antioxidant Activity of Egg White Protein-Derived Peptides after Simulated In-Vitro Gastrointestinal Digestion. Antioxidants, 9(11), 1114. Retrieved from https://www.mdpi.com/2076-3921/9/11/1114

Rattes, & Oliveira. (2007). Spray drying conditions and encapsulating composition effects on formation and properties of sodium diclofenac microparticles. Powder Technology, 171(1), 7-14. Reboul. (2017). Vitamin E Bioavailability: Mechanisms of Intestinal Absorption in the Spotlight. Antioxidants (Basel), 6(4). doi:10.3390/antiox6040095

Reboul. (2019a). Mechanisms of Carotenoid Intestinal Absorption: Where Do We Stand? Nutrients, 11(4). doi:10.3390/nu11040838

Reboul. (2019b). Vitamin E intestinal absorption: Regulation of membrane transport across the enterocyte. IUBMB Life, 71(4), 416-423. doi:https://doi.org/10.1002/iub.1955

Reboul, Richelle, Perrot, Desmoulins-Malezet, Pirisi, & Borel. (2006). Bioaccessibility of Carotenoids and Vitamin E from Their Main Dietary Sources. Journal of Agricultural and Food Chemistry, 54(23), 8749-8755. doi:10.1021/jf061818s

Ribaya-Mercado, & Blumberg. (2004). Lutein and Zeaxanthin and Their Potential Roles in Disease Prevention. Journal of the American College of Nutrition, 23(sup6), 567S-587S. doi:10.1080/07315724.2004.10719427

Rich, Faulks, Wickham, & Fillery-Travis. (2003). Solubilization of carotenoids from carrot juice and spinach in lipid phases: II. Modeling the duodenal environment. Lipids, 38(9), 947-956. doi:10.1007/s11745-003-1148-z

Richter Reis. (2017). Effect of blanching on food physical, chemical, and sensory quality. New perspectives on food blanching, 7-48.

Rizos, Ntzani, Bika, Kostapanos, & Elisaf. (2012). Association Between Omega-3 Fatty Acid Supplementation and Risk of Major Cardiovascular Disease Events: A Systematic Review and MetaanalysisOmega-3 Fatty Acids and Cardiovascular Disease. JAMA, 308(10), 1024-1033. doi:10.1001/2012.jama.11374

Rizvi, Raza, Ahmed, Ahmad, Abbas, & Mahdi. (2014). The role of vitamin e in human health and some diseases. Sultan Qaboos Univ Med J, 14(2), e157-165.

Roberts, & Moreau. (2016). Functional properties of spinach (Spinacia oleracea L.) phytochemicals and bioactives. Food & Function, 7(8), 3337-3353. doi:10.1039/C6FO00051G

Rocha, Nitenberg, Girard-Egrot, Jouhet, Maréchal, Block, & Breton. (2018). Do Galactolipid Synthases Play a Key Role in the Biogenesis of Chloroplast Membranes of Higher Plants? Frontiers in Plant Science, 9. doi:10.3389/fpls.2018.00126

Rock, Lovalvo, Emenhiser, Ruffin, Flatt, & Schwartz. (1998). Bioavailability of β -Carotene Is Lower in Raw than in Processed Carrots and Spinach in Women. The Journal of Nutrition, 128(5), 913-916. doi:10.1093/jn/128.5.913

Rock, & Swendseid. (1992). Plasma beta-carotene response in humans after meals supplemented with dietary pectin. Am J Clin Nutr, 55(1), 96-99. doi:10.1093/ajcn/55.1.96

Rodriguez-Amaya. (1999). Changes in carotenoids during processing and storage of foods. Archivos latinoamericanos de nutrición, 49(3 Suppl 1), 38S-47S.

Rodriguez-Amaya. (2019). Natural food pigments and colorants. In Bioactive molecules in food (pp. 867-901): Springer.

Rodríguez-Roque, de Ancos, Sánchez-Vega, Sánchez-Moreno, Cano, Elez-Martínez, & Martín-Belloso. (2016). Food matrix and processing influence on carotenoid bioaccessibility and lipophilic antioxidant activity of fruit juice-based beverages. Food Funct, 7(1), 380-389. doi:10.1039/c5fo01060h

Romano, Giordano, Vitiello, Grottaglie, & Musso. (2012). Comparison of the Frying Performance of Olive Oil and Palm Superolein. Journal of Food Science, 77. doi:10.1111/j.1750-3841.2012.02663.x

Roy, & Kumar. (2014). A practical approach on SDS PAGE for separation of protein. International Journal of Science and Research, 3(8), 955-960.

Różyło, Piekut, Dziki, Smolewska, Gawłowski, Wójtowicz, & Gawlik-Dziki. (2022). Effects of Wet and Dry Micronization on the GC-MS Identification of the Phenolic Compounds and Antioxidant Properties of Freeze-Dried Spinach Leaves and Stems. Molecules, 27(23), 8174.

Rudra, Sarkar, & Shivhare. (2008). Thermal Degradation Kinetics of Chlorophyll in Pureed Coriander Leaves. Food and Bioprocess Technology, 1(1), 91-99. doi:10.1007/s11947-007-0016-z

Ryan, O'Connell, O'Sullivan, Aherne, & O'Brien. (2008). Micellarisation of carotenoids from raw and cooked vegetables. Plant Foods Hum Nutr, 63(3), 127-133. doi:10.1007/s11130-008-0081-0

Ryżak, & Bieganowski. (2011). Methodological aspects of determining soil particle-size distribution using the laser diffraction method. Journal of Plant Nutrition and Soil Science, 174(4), 624-633. Sablani. (2006). Drying of fruits and vegetables: retention of nutritional/functional quality. Drying

Technology, 24(2), 123-135.

Sagar, & Suresh Kumar. (2010). Recent advances in drying and dehydration of fruits and vegetables: a review. Journal of Food Science and Technology, 47, 15-26.

Sahaka. (2020). The digestion of galactolipids and its ubiquitous function in Nature for the uptake of the essential α -linolenic acid. Food & Function, 11(8), 6710-6744. doi:10.1039/D0FO01040E

Sahaka, Amara, Lecomte, Rodier, Lafont, Villeneuve, ... Carrière. (2021). Quantitative monitoring of galactolipid hydrolysis by pancreatic lipase-related protein 2 using thin layer chromatography and thymol-sulfuric acid derivatization. Journal of Chromatography B, 1173, 122674. doi:https://doi.org/10.1016/j.jchromb.2021.122674

Saini, Nile, & Park. (2015). Carotenoids from fruits and vegetables: Chemistry, analysis, occurrence, bioavailability and biological activities. Food Res Int, 76(Pt 3), 735-750. doi:10.1016/j.foodres.2015.07.047

Saini, Shetty, & Giridhar. (2014). GC-FID/MS Analysis of Fatty Acids in Indian Cultivars of Moringa oleifera: Potential Sources of PUFA. Journal of the American Oil Chemists' Society, 91(6), 1029-1034. doi:https://doi.org/10.1007/s11746-014-2439-9

Sakamoto, Miyagishima, & Jarvis. (2008). Chloroplast biogenesis: control of plastid development, protein import, division and inheritance. The Arabidopsis book/American Society of Plant Biologists, 6.

Sakata, Yamamoto, Ishikawa, Yagi, Etoh, & Ina. (1990). Chlorophyllone-a, a new pheophorbide-a related compound isolated from ruditapes philippinarum as an antioxidative compound. Tetrahedron Letters, 31(8), 1165-1168. doi:https://doi.org/10.1016/S0040-4039(00)88754-7

Salentinig. (2019). Supramolecular structures in lipid digestion and implications for functional food delivery. Current Opinion in Colloid & Interface Science, 39, 190-201. doi:https://doi.org/10.1016/j.cocis.2019.02.002

Salhi, Amara, Mansuelle, Puppo, Lebrun, Gontero, . . . Carrière. (2020). Characterization of all the lipolytic activities in pancreatin and comparison with porcine and human pancreatic juices. Biochimie, 169, 106-120. doi:https://doi.org/10.1016/j.biochi.2019.07.004

Salvia-Trujillo, Qian, Martín-Belloso, & McClements. (2013). Influence of particle size on lipid digestion and β -carotene bioaccessibility in emulsions and nanoemulsions. Food Chemistry, 141(2), 1472-1480.

Samborska, Poozesh, Barańska, Sobulska, Jedlińska, Arpagaus, . . . Jafari. (2022). Innovations in spray drying process for food and pharma industries. Journal of Food Engineering, 321, 110960. doi:https://doi.org/10.1016/j.jfoodeng.2022.110960

Sams. (2016). Relevant pH and lipase for in vitro models of gastric digestion. Food & Function, 7(1), 30-45.

Sams, Paume, Giallo, & Carrière. (2016). Relevant pH and lipase for in vitro models of gastric digestion. Food Funct, 7(1), 30-45. doi:10.1039/c5fo00930h

Savjani, Gajjar, & Savjani. (2012). Drug solubility: importance and enhancement techniques. ISRN Pharm, 2012, 195727. doi:10.5402/2012/195727

Schieber, & Carle. (2005). Occurrence of carotenoid cis-isomers in food: Technological, analytical, and nutritional implications. Trends in Food Science & Technology, 16(9), 416-422.

Schultz. (1990). Biosynthesis of α -tocopherol in chloroplasts of higher plants. Fett Wissenschaft

Technologie, 92(2), 86-91.

Schwartz, Cooperstone, Cichon, Joachim, & Giusti. (2017). Colorants. In Fennema's food chemistry (pp. 681-752): CRC press.

Schwartz, & Lorenzo. (1990). Chlorophylls in foods. Critical Reviews in Food Science and Nutrition, 29(1), 1-17. doi:10.1080/10408399009527511

Schwartz, & Lorenzo. (1991). Chlorophyll Stability During Continuous Aseptic Processing and Storage. Journal of Food Science, 56(4), 1059-1062. doi:https://doi.org/10.1111/j.1365-2621.1991.tb14641.x

Schwartz, & Von Elbe. (1983). Kinetics of Chlorophyll Degradation to Pyropheophytin in Vegetables. Journal of Food Science, 48(4), 1303-1306. doi:https://doi.org/10.1111/j.1365-2621.1983.tb09216.x

SCOF. (2006). European Food Safety Authority.

Seiwert, Witt, Janshoff, & Paulsen. (2017). The non-bilayer lipid MGDG stabilizes the major lightharvesting complex (LHCII) against unfolding. Scientific Reports, 7(1), 5158. doi:10.1038/s41598-017-05328-7

Sekar, & Ramasamy. (2015). Photosynthetic Energy Conversion: Recent Advances and Future Perspective. Electrochemical Society Interface, 24, 67-73. doi:10.1149/2.F06153if

Sensoy. (2021). A review on the food digestion in the digestive tract and the used in vitro models. Current research in food science, 4, 308-319.

Seppanen, Song, & Saari Csallany. (2010). The Antioxidant Functions of Tocopherol and Tocotrienol Homologues in Oils, Fats, and Food Systems. Journal of the American Oil Chemists' Society, 87(5), 469-481. doi:10.1007/s11746-009-1526-9

Serna-Cock, Vargas-Muñoz, & Aponte. (2015). Structural, physical, functional and nutraceutical changes of freeze-dried fruit. African Journal of Biotechnology, 14(6), 442-450.

Sevenich, Reineke, Hecht, Fröhling, Rauh, Schlüter, & Knorr. (2015). Impact of different water activities (a w) adjusted by solutes on high pressure high temperature inactivation of Bacillus amyloliquefaciens spores. Front Microbiol, 6, 689. doi:10.3389/fmicb.2015.00689

Severini, Baiano, De Pilli, Carbone, & Derossi. (2005). Combined treatments of blanching and dehydration: study on potato cubes. Journal of Food Engineering, 68(3), 289-296.

Severini, Giuliani, De Filippis, Derossi, & De Pilli. (2016). Influence of different blanching methods on colour, ascorbic acid and phenolics content of broccoli. Journal of Food Science and Technology, 53, 501-510.

Seybold, Fröhlich, Bitsch, Otto, & Böhm. (2004). Changes in contents of carotenoids and vitamin E during tomato processing. J Agric Food Chem, 52(23), 7005-7010. doi:10.1021/jf049169c

Sharma, Jana, & Chavan. (2012). Functionality of Milk Powders and Milk-Based Powders for End Use Applications—A Review. Comprehensive Reviews in Food Science and Food Safety, 11(5), 518-528. doi:https://doi.org/10.1111/j.1541-4337.2012.00199.x

Shilpa, Shwetha, Raju, & Lakshminarayana. (2020). Factors affecting bioaccessibility and bioefficacy of carotenoids. In Carotenoids: Properties, processing and applications (pp. 41-73): Elsevier.

Shishir, & Chen. (2017). Trends of spray drying: A critical review on drying of fruit and vegetable juices. Trends in Food Science & Technology, 65, 49-67.

doi:https://doi.org/10.1016/j.tifs.2017.05.006

Sias, Ferrato, Grandval, Lafont, Boullanger, Caro, . . . Carrière. (2004). Human Pancreatic Lipase-Related Protein 2 Is a Galactolipase †. Biochemistry, 43, 10138-10148. doi:10.1021/bi049818d Sicari, Loizzo, Sanches Silva, Romeo, Spampinato, Tundis, . . . Musarella. (2020). The Effect of Blanching on Phytochemical Content and Bioactivity of Hypochaeris and Hyoseris Species (Asteraceae), Vegetables Traditionally Used in Southern Italy. Foods, 10(1). doi:10.3390/foods10010032

Silva, & Gibbs. (2004). Target Selection in Designing Pasteurization Processes for Shelf-Stable High-Acid Fruit Products. Critical Reviews in Food Science and Nutrition, 44(5), 353-360. doi:10.1080/10408690490489251

Spinach production in 2021; Crops/Regions/World/Production Quantity/Year from pick lists. (2023). UN Food and Agriculture Organization.

Sridhar, Ponnuchamy, Kumar, & Kapoor. (2021). Food preservation techniques and nanotechnology for increased shelf life of fruits, vegetables, beverages and spices: a review. Environmental Chemistry Letters, 19(2), 1715-1735. doi:10.1007/s10311-020-01126-2

Staehelin. (2003). Chloroplast structure: from chlorophyll granules to supra-molecular architecture of thylakoid membranes. Photosynth Res, 76(1-3), 185-196. doi:10.1023/a:1024994525586

Staehelin, & Paolillo. (2020). A brief history of how microscopic studies led to the elucidation of the 3D architecture and macromolecular organization of higher plant thylakoids. Photosynthesis Research, 145(3), 237-258. doi:10.1007/s11120-020-00782-3

Stanley, Bourne, Stone, & Wismer. (1995). Low temperature blanching effects on chemistry, firmness and structure of canned green beans and carrots. Journal of Food Science, 60(2), 327-333.

Steet, & Tong. (1996). Degradation kinetics of green color and chlorophylls in peas by colorimetry and HPLC. Journal of Food Science, 61(5), 924-928.

Suhag, Nayik, Karabagias, & Nanda. (2021). Development and Characterization of a Nutritionally Rich Spray-Dried Honey Powder. Foods, 10(1). doi:10.3390/foods10010162

Sun, Wang, Sui, She, Zhai, Wang, & Deng. (2012). Effect of particle size on solubility, dissolution rate, and oral bioavailability: evaluation using coenzyme Q¹⁰ as naked nanocrystals. Int J Nanomedicine, 7, 5733-5744. doi:10.2147/ijn.S34365

Sun, Wang, Wang, & Guo. (2018). Effects of Processing on Structure and Thermal Properties of Powdered Preterm Infant Formula. J Food Sci, 83(6), 1685-1694. doi:10.1111/1750-3841.14162

Sutcharit, Wattanakul, Price, Di Bari, Gould, Yakubov, . . . Gray. (2023). Chloroplast/Thylakoid-rich Material: A Possible Alternative to the Chemically Synthesised Flow Enhancer Polyglycerol Polyricinoleate in Oil-Based Systems. Food Research International, 112472. doi:https://doi.org/10.1016/j.foodres.2023.112472

Suzuki, Honda, Mukasa, & Kim. (2005). Effects of lipase, lipoxygenase, peroxidase, and rutin on quality deteriorations in buckwheat flour. Journal of Agricultural and Food Chemistry, 53(21), 8400-8405.

Swapnil, Meena, Singh, Dhuldhaj, Harish, & Marwal. (2021). Vital roles of carotenoids in plants and humans to deteriorate stress with its structure, biosynthesis, metabolic engineering and functional aspects. Current Plant Biology, 26, 100203. doi:https://doi.org/10.1016/j.cpb.2021.100203

Syamila, Gedi, Briars, Ayed, & Gray. (2019). Effect of temperature, oxygen and light on the degradation of β -carotene, lutein and α -tocopherol in spray-dried spinach juice powder during storage. Food Chem, 284, 188-197. doi:10.1016/j.foodchem.2019.01.055

Szymańska, & Kruk. (2008). Tocopherol content and isomers' composition in selected plant species. Plant Physiology and Biochemistry, 46(1), 29-33.

Taïbi, Taïbi, Abderrahim, Ennajah, Belkhodja, & Mulet. (2016). Effect of salt stress on growth, chlorophyll content, lipid peroxidation and antioxidant defence systems in Phaseolus vulgaris L. South African Journal of Botany, 105, 306-312.

Tan, & Francis. (1962). Effect of Processing Temperature on Pigments and Color of Spinacha. Journal of Food Science, 27(3), 232-241. doi:https://doi.org/10.1111/j.1365-2621.1962.tb00086.x Tao, Ye, & Cao. (2021). Isomerization and degradation of all-trans- β -carotene during in-vitro digestion. Food Science and Human Wellness, 10(3), 370-374. doi:https://doi.org/10.1016/j.fshw.2021.02.028

Tavman, Otles, Glaue, & Gogus. (2019). 4 - Food preservation technologies. In C. M. Galanakis (Ed.), Saving Food (pp. 117-140): Academic Press.

Teng, & Chen. (1999). Formation of pyrochlorophylls and their derivatives in spinach leaves during heating. Food Chemistry, 65(3), 367-373. doi:https://doi.org/10.1016/S0308-8146(98)00237-4

Thakur, Raigond, Singh, Mishra, Singh, Lal, & Dutt. (2020). Recent updates on bioaccessibility of phytonutrients. Trends in Food Science & Technology, 97, 366-380. doi:https://doi.org/10.1016/j.tifs.2020.01.019

Thirumdas, Sarangapani, & Annapure. (2015). Cold Plasma: A novel Non-Thermal Technology for Food Processing. Food Biophysics, 10(1), 1-11. doi:10.1007/s11483-014-9382-z

Thomas, Holm, Rades, & Müllertz. (2012). Characterising lipid lipolysis and its implication in lipidbased formulation development. Aaps j, 14(4), 860-871. doi:10.1208/s12248-012-9398-6

Thuenemann. (2015). Dynamic Digestion Models: General Introduction. In K. Verhoeckx, P. Cotter, I. López-Expósito, C. Kleiveland, T. Lea, A. Mackie, T. Requena, D. Swiatecka, & H. Wichers (Eds.), The Impact of Food Bioactives on Health: in vitro and ex vivo models (pp. 33-36). Cham: Springer International Publishing.

Tijskens, Schijvens, & Biekman. (2001). Modelling the change in colour of broccoli and green beans during blanching. Innovative Food Science & Emerging Technologies, 2(4), 303-313. doi:https://doi.org/10.1016/S1466-8564(01)00045-5

Tola, & Ramaswamy. (2018). Novel processing methods: updates on acidified vegetables thermalprocessing.CurrentOpinioninFoodScience,23,64-69.doi:https://doi.org/10.1016/j.cofs.2018.06.003

Tontul, & Topuz. (2017). Spray-drying of fruit and vegetable juices: Effect of drying conditions on the product yield and physical properties. Trends in Food Science & Technology, 63, 91-102. doi:https://doi.org/10.1016/j.tifs.2017.03.009

Torcello-Gómez, Gedi, Ibbett, Nawaz Husain, Briars, & Gray. (2019). Chloroplast-rich material from the physical fractionation of pea vine (Pisum sativum) postharvest field residue (Haulm). Food Chemistry, 272, 18-25. doi:https://doi.org/10.1016/j.foodchem.2018.08.018

Traber. (2014). Vitamin E inadequacy in humans: causes and consequences. Advances in nutrition, 5(5), 503-514.

Tränkner, Tavakol, & Jákli. (2018). Functioning of potassium and magnesium in photosynthesis, photosynthate translocation and photoprotection. Physiologia Plantarum, 163(3), 414-431. doi:https://doi.org/10.1111/ppl.12747

Tyssandier, Lyan, & Borel. (2001). Main factors governing the transfer of carotenoids from emulsion lipid droplets to micelles. Biochimica et Biophysica Acta (BBA)-Molecular and Cell Biology of Lipids, 1533(3), 285-292.

Tze, Chong, Yusof, Chin, A. Talib, Taip, & Aziz. (2012a). Physicochemical and nutritional properties of spray-dried pitaya fruit powder as natural colorant. Food science and biotechnology, 21. doi:10.1007/s10068-012-0088-z

Tze, Han, Yusof, Ling, Talib, Taip, & Aziz. (2012b). Physicochemical and nutritional properties of spray-dried pitaya fruit powder as natural colorant. Food Science and Biotechnology, 21(3), 675-682.

Umair, Jabeen, Ke, Jabbar, Javed, Abid, . . . Adam Conte-Junior. (2022). Thermal treatment alternatives for enzymes inactivation in fruit juices: Recent breakthroughs and advancements. Ultrason Sonochem, 86, 105999. doi:10.1016/j.ultsonch.2022.105999

Urquieta-Herrero, Cornejo-Mazón, Gutiérrez-López, & Garcia-Pinilla. (2021). Effect of two pasteurization methods on the content of bioactive compounds and antioxidant capacity of nance (Byrsonima crassifolia) pulp and their kinetics of loss during refrigerated storage. Revista Mexicana de Ingeniería Química, 20(2), 663-678.

Van Bennekum, Werder, Thuahnai, Han, Duong, Williams, . . . Hauser. (2005). Class B scavenger receptor-mediated intestinal absorption of dietary beta-carotene and cholesterol. Biochemistry, 44(11), 4517-4525. doi:10.1021/bi0484320

Van Loey, Ooms, Weemaes, Van den Broeck, Ludikhuyze, Indrawati, . . . Hendrickx. (1998). Thermal and Pressure–Temperature Degradation of Chlorophyll in Broccoli (Brassica oleracea L. italica) Juice: A Kinetic Study. Journal of Agricultural and Food Chemistry, 46(12), 5289-5294. doi:10.1021/jf980505x

Van Loo-Bouwman, Naber, Minekus, van Breemen, Hulshof, & Schaafsma. (2014). Food matrix effects on bioaccessibility of β -carotene can be measured in an in vitro gastrointestinal model. J Agric Food Chem, 62(4), 950-955. doi:10.1021/jf403312v

Vardanega, Muzio, Silva, Prata, & Meireles. (2019). Obtaining functional powder tea from Brazilian ginseng roots: Effects of freeze and spray drying processes on chemical and nutritional quality, morphological and redispersion properties. Food Research International, 116, 932-941. doi:https://doi.org/10.1016/j.foodres.2018.09.030

Vázquez, García-Risco, Jaime, Reglero, & Fornari. (2013). Simultaneous extraction of rosemary and spinach leaves and its effect on the antioxidant activity of products. The Journal of Supercritical Fluids, 82, 138-145. doi:https://doi.org/10.1016/j.supflu.2013.07.004

Velasco, Lim, Pangborn, & Whitaker. (1989). Enzymes Responsible for Off-Flavor and Off-Aroma in Blanched and Frozen-Stored Vegetables 1. Biotechnology and Applied Biochemistry, 11(1), 118-127.

Vera Zambrano, Dutta, Mercer, MacLean, & Touchie. (2019). Assessment of moisture content measurement methods of dried food products in small-scale operations in developing countries: A review. Trends in Food Science & Technology, 88, 484-496. doi:https://doi.org/10.1016/j.tifs.2019.04.006

Verhoeckx, Cotter, López-Expósito, Kleiveland, Lea, Mackie, . . . Wichers. (2015). The impact of food bioactives on health: in vitro and ex vivo models.

Verpoorte, van der Heijden, Ten Hoopen, & Memelink. (1999). Metabolic engineering of plant secondary metabolite pathways for the production of fine chemicals. Biotechnology letters, 21(6), 467-479.

Vitali Čepo, Radić, Turčić, Anić, Komar, & Šalov. (2020). Food (Matrix) Effects on Bioaccessibility and Intestinal Permeability of Major Olive Antioxidants. Foods, 9(12). doi:10.3390/foods9121831

Vu, Roth, Tamura, Samarakoon, Shiva, Honey, . . . Welti. (2014). Head-group acylation of monogalactosyldiacylglycerol is a common stress response, and the acyl-galactose acyl composition varies with the plant species and applied stress. Physiol Plant, 150(4), 517-528. doi:10.1111/ppl.12132

Walkling-Ribeiro, Noci, Cronin, Lyng, & Morgan. (2009). Shelf life and sensory evaluation of orange juice after exposure to thermosonication and pulsed electric fields. Food and Bioproducts Processing, 87(2), 102-107. doi:https://doi.org/10.1016/j.fbp.2008.08.001

Wang, & Benning. (2012a). Chloroplast lipid synthesis and lipid trafficking through ER–plastid membrane contact sites. Biochemical Society Transactions, 40(2), 457-463. doi:10.1042/bst20110752

Wang, Chang, Chou, Chen, Deng, Harvey, . . . Bickford. (2005). Dietary supplementation with blueberries, spinach, or spirulina reduces ischemic brain damage. Experimental Neurology, 193(1), 75-84. doi:https://doi.org/10.1016/j.expneurol.2004.12.014

Wang, Li, Jiang, Qi, & Zhou. (2014). Relationship between secondary structure and surface hydrophobicity of soybean protein isolate subjected to heat treatment. Journal of chemistry, 2014. Wang, Liu, Mei, Nakajima, & Yin. (2012b). Preliminary study into the factors modulating β -carotene micelle formation in dispersions using an in vitro digestion model. Food hydrocolloids, 26(2), 427-433. doi:https://doi.org/10.1016/j.foodhyd.2010.11.018

Wang, Xing, Wang, & Guo. (2017). The analysis of the causes of protein precipitate formation in the blanched soymilk. Food Chemistry, 218, 341-347.

Wang, Xu, Jin, Zhang, Lai, Zhou, . . . Yang. (2016). A Putative Chloroplast-Localized Ca2+/H+ Antiporter CCHA1 Is Involved in Calcium and pH Homeostasis and Required for PSII Function in Arabidopsis. Molecular Plant, 9(8), 1183-1196. doi:https://doi.org/10.1016/j.molp.2016.05.015

Wang, Yang, Wan, & Li. (2019). The Significance of Calcium in Photosynthesis. International Journal of Molecular Sciences, 20(6), 1353. Retrieved from https://www.mdpi.com/1422-0067/20/6/1353 Ward, & Matejtschuk. (2021). The principles of freeze-drying and application of analytical technologies. Cryopreservation and freeze-drying protocols, 99-127.

Wason, Verma, & Subbiah. (2021). Validation of process technologies for enhancing the safety of low-moisture foods: A review. Comprehensive Reviews in Food Science and Food Safety, 20(5), 4950-4992. doi:https://doi.org/10.1111/1541-4337.12800

Wattanakul, Sahaka, Amara, Mansor, Gontero, Carrière, & Gray. (2019). In vitro digestion of galactolipids from chloroplast-rich fraction (CRF) of postharvest, pea vine field residue (haulm) and spinach leaves. Food & Function, 10(12), 7806-7817. doi:10.1039/C9FO01867K

Wattanakul, Syamila, Briars, Ayed, Price, Darwish, . . . Gray. (2020). Effect of steam sterilisation on

lipophilic nutrient stability in a chloroplast-rich fraction (CRF) recovered from postharvest, pea vine field residue (haulm). Food Chemistry, 334, 127589. doi:10.1016/j.foodchem.2020.127589

Wattanakul, Syamila, Darwish, Gedi, Sutcharit, Chi, . . . Gray. (2022). Bioaccessibility of essential lipophilic nutrients in a chloroplast-rich fraction (CRF) from agricultural green waste during simulated human gastrointestinal tract digestion. Food & Function, 13(9), 5365-5380. doi:10.1039/D2FO00604A

Weber, & Grune. (2012). The contribution of β -carotene to vitamin A supply of humans. Molecular Nutrition & Food Research, 56(2), 251-258. doi:https://doi.org/10.1002/mnfr.201100230

Whitaker. (1986). Fatty-acid composition of polar lipids in fruit and leaf chloroplasts of "16:3"- and "18:3"-plant species. Planta, 169(3), 313-319. doi:10.1007/bf00392125

Whitaker. (2018). Principles of enzymology for the food sciences: Routledge.

Wijendran, & Hayes. (2004). DIETARY n-6 AND n-3 FATTY ACID BALANCE AND CARDIOVASCULAR HEALTH. Annual Review of Nutrition, 24(1), 597-615. doi:10.1146/annurev.nutr.24.012003.132106

Wijk, & Kessler. (2017). Plastoglobuli: Plastid Microcompartments with Integrated Functions in Metabolism, Plastid Developmental Transitions, and Environmental Adaptation. Annual Review of Plant Biology, 68(1), 253-289. doi:10.1146/annurev-arplant-043015-111737

Wójtowicz, Grzyb, Szach, Mazur, & Gieczewska. (2021). Bean and Pea Plastoglobules Change in Response to Chilling Stress. Int J Mol Sci, 22(21). doi:10.3390/ijms222111895

Xiao. (2011). Pancreatic lipase-related protein-2 (PLRP2) can contribute to dietary fat digestion in human newborns. J Biol Chem, 286(30), 26353-26363. doi:10.1074/jbc.M111.249813

Yang, & McClements. (2013). Vitamin E bioaccessibility: Influence of carrier oil type on digestion and release of emulsified α -tocopherol acetate. Food Chemistry, 141(1), 473-481.

Zayadan, Kakimova, Bolatkhan, Sandybayeva, Kosalbayev, & Nurabayeva. (2021). Production of biohydrogen from cyanobacteria: current challenges and opportunities. International Journal of Biology and Chemistry, 14(1), 4-20.

Zeb. (2012). Oxidation and formation of oxidation products of β -carotene at boiling temperature. Chemistry and Physics of Lipids, 165(3), 277-281. doi:https://doi.org/10.1016/j.chemphyslip.2012.02.005

Zhai, Zhao, Yin, Chen, Wang, Li, . . . Ma. (2013). Two wheat glutathione peroxidase genes whose products are located in chloroplasts improve salt and H2O2 tolerances in Arabidopsis. PloS one, 8(10), e73989.

Zhang, & Hamauzu. (2004). Phenolics, ascorbic acid, carotenoids and antioxidant activity of broccoli and their changes during conventional and microwave cooking. Food Chemistry, 88(4), 503-509.

Zhang, Pang, Xuewu, Ji, & Jiang. (2005). Role of peroxidase in anthocyanin degradation in litchi fruit pericarp. Food Chemistry, 90(1-2), 47-52.

Zheng, & Lu. (2011). Effect of microwave pretreatment on the kinetics of ascorbic acid degradation and peroxidase inactivation in different parts of green asparagus (Asparagus officinalis L.) during water blanching. Food Chemistry, 128(4), 1087-1093.

Zhu, Ji, Eum, & Zude. (2009). Evaluation of the non-enzymatic browning in thermally processed apple juice by front-face fluorescence spectroscopy. Food Chemistry, 113(1), 272-279.

doi:https://doi.org/10.1016/j.foodchem.2008.07.009

Zhu, Zhang, Hinds, Sun, & Tiwari. (2021). Applications of ultrasound to enhance fluidized bed drying of Ascophyllum Nodosum: Drying kinetics and product quality assessment. Ultrasonics Sonochemistry, 70, 105298. doi:https://doi.org/10.1016/j.ultsonch.2020.105298

Žolnere, Arnold, Hull, & Everett. (2019). Cheese proteolysis and matrix disintegration during in vitro digestion. Food Structure, 21, 100114. doi:https://doi.org/10.1016/j.foostr.2019.100114